# Impact Assessment of the Partial Closure of the Ham Luong Estuary

Multidisciplinary Project MDP 343



## Impact Assessment of the Partial Closure of the Ham Luong Estuary

by

## MDP 343

Student Name	Student Number
Jasper van Asselt	4956516
Hannah Bervoets	4837746
Celine Bloem	4666607
Vincent Kindermann	4657934
Ruben Koopman	4893247
Linde Zwaan	4713346

Supervisors:	Dr. ir. Cong Mai Van,
	ir. H. Alkisaei,
	Dr. S. Truong Hong
Faculty:	Civil Engineering and Geosciences, Delft

Cover: Bird's Eye View of Green Farmland Landscape by Tom Fisk









## Preface

This report is the result of a Multidisciplinary Project on the closure of the Ham Luong estuary in the Mekong Delta, Vietnam. The project came into being by the long-standing collaboration between Thuyloi University and Delft University of Technology. The Multidisciplinary Project is part of the curriculum of the Master Civil Engineering of the Delft University of Technology.

By means of different master track specialisations within the field of Civil Engineering (Structural Engineering, Hydraulic and Offshore Structures, River Engineering, and Coastal Engineering), the project aims to provide more insights into the ongoing discussion on the closure of the Ham Luong estuary. This document can be used by any party that is interested in the current situation of the Ham Luong estuary, the effect of the closure of the estuary on the hydro- and morphodynamic characteristics, the impact of the closure on the surroundings, and an optimal flood protection design. In reading this report, prerequisite knowledge in the fields of hydraulic modelling and hydraulic structures is recommended. When interpreting and/or reproducing the results presented in this report, the large number of assumptions, simplifications, and uncertainties which are inherent to this project and the scope should be taken into account accordingly.

We would like to thank Dr. Truong Hong Son, our supervisor at Thuyloi University in Hanoi, for his supervision in this project and his great advice. He helped us set up the project by providing us a lot of useful suggestions and valuable information. Furthermore, he also brought us into contact with different experts on the topic and he joined us during our field trip. Without him, this project would not have been possible. We would like to thank the Southern Institute for Water Resource Planning (SIWRP), the Institute of Coastal and Offshore Engineering (ICOE) and Royal HaskoningDHV for their professional insights which have been a great addition to our work. Additionally, we would like to thank Lindsey Schwidder from the Innovation & Impact Centre at TU Delft, who brought us in contact with our supervisors and was the first one to provide us with information about a possible research project in Vietnam. Last but not least, we want to thank our supervisors from TU Delft: Dr. Ir. Cong Mai Van and Ir. Hoessein Alkisaei. They provided us useful advice and guidance throughout the project.

> Jasper van Asselt, Hannah Bervoets, Celine Bloem, Vincent Kindermann, Ruben Koopman, Linde Zwaan

Hanoi Tuesday October 26th, 2023

## Abstract

The Mekong Delta in Vietnam is facing several challenges as a result of climate change. Among others, the effects include an increase in river discharge during the wet season, leading to river floods, and a decrease in river discharge during the dry season. The decrease in discharge results in a shortage of fresh water required for irrigation and drinking water. Besides that, the combination of sea-level rise, land subsidence, and decreased river discharge during the dry season results in saltwater intrusion. This threatens freshwater supply even more. Furthermore, there is an increasing risk of floods from the sea due to low land elevation and the rising sea level in combination with the occurrence of storm surges.

The scope of this research is the area around the Ham Luong estuary, which is a branch of the Mekong River. The partial closure of this river branch is considered by the Vietnamese government as a measure to reduce the effect of the above-mentioned effects of climate change. However, not enough research has been conducted yet on the impact of a partial closure on the Ham Luong estuary. This has lead to the following research question:

"What is the impact of various closure scenarios on the hydraulic characteristics and social activities in the Ham Luong estuary, considering a 75-year forecast?"

The region of the Ham Luong estuary is characterised by its intensive agri- and aquaculture. More than 60% of the inhabitants is directly active within the agri- or aquaculture. As these activities are strongly dependent on the salinity of the estuary, they are highly affected by the effects of climate change. The region is densely populated with more than 125,000 inhabitants living near the Ham Luong estuary. It is clear that the effects of climate change are threatening the region in hydraulic aspects, as well as socio-economic aspects. A partial closure could reduce these effects, but will influence the region in several ways. In order to estimate the impact, a combination of hydraulic and socio-economic aspects is assessed based on a criteria set. This criteria set contains the criteria of freshwater supply, agricultural and aquaculture adaptation, biodiversity, stable riverbanks, and navigability. These criteria will be tested on a total of four alternative interventions in the Ham Luong estuary. Three alternatives with a storm surge barrier and one alternative without a storm surge barrier. All alternatives include heightening of the existing dyke system, as this seems to be inevitable when aiming for long-term development in the region. The extend of dyke heightening is subject to the choice of alternative.

As a part of the impact analysis, a Delft3D model was built to analyse the hydrodynamic and morphodynamic processes in the Ham Luong estuary. The model was restricted to the chosen spatial scope, which only covers the Ham Luong estuary, without any upstream bifurcations. The model gave insights in processes like salt intrusion, sedimentation rates, and water levels. However, due to model simplifications and assumptions, the outcomes of the model where not useful for quantitative assessments. Still, the results are used to compare the impact of the different alternatives to each other. As expected, the alternatives that include a storm surge barrier will provide more possibilities to retain fresh water than the alternative without a barrier. From the results, it followed that the limited spatial scope excludes the redistribution of upstream discharge. It is recommended to look at a larger scale of the Mekong Delta when assessing hydro- and morphodynamic processes.

Forming a flood protection system, the structural design of such a storm surge barrier, together with a quick estimation of a dyke system. The dyke system is different for each alternative, depending on the presence and the location of a barrier. The barrier design includes a thorough analysis on feasibility of gate types, technical requirements, load combinations, design of dimensions, and the operation. The load combinations take hydrostatic, hydrodynamic, wind, and soil loads into account. The design of the dimensions is done for the gates, sill, lifting structure, pier, foundation, and the bed protection.

By assessing the above-mentioned criteria, a preferred solution is identified. This preference is based on a Multi-Criteria Analysis, which includes weighted scores for all alternatives. The outcome of the Multi-Criteria Analysis appears to be very sensitive to the rating and weights of the criteria, which makes it difficult to identify one of the alternatives as the preferred solution based on only the score on the different criteria. For this reason more research is needed. However, when including a cost estimation of the four alternatives, it can be stated that the alternative of no storm surge barrier and only the corresponding extensive dyke heightening could be considered as most cost-beneficial alternative and therefore as the preferred solution.

It is expected that with or without closure of the Ham Luong estuary the system will change. The availability of fresh water will be improved by the presence of a closure, although more research is needed to specify this further. The increasing salt intrusion, as a result of Relative Sea-Level Rise (RSLR) will lead to agricultural and aquaculture adaptation in all alternatives. Either due to the construction of the barrier, or due to the gradual RSLR. A closure also has effect on the biodiversity, stability of the river banks, and navigability in the river. When implementing a closure these effects should be further investigated to assess the effect quantitatively.

Pı	reface	iii
$\mathbf{A}$	bstra	iv
Li	st of	Figures viii
Li	st of	Tables x
Li	st of	Abbreviations xi
1	<b>Intro</b> 1.1 1.2 1.3	Deduction  1    Challenges  1    Research objective  2    Report structure  2
2	Met 2.1 2.2 2.3 2.4 2.5	hodology3Analysis of the system3Approach scenario selection3Modelling process3Approach design of flood protection system3MCA approach3
3	Syst 3.1 3.2 3.3	em description  4    Local Climate  4    Hydraulic characteristics  5    3.2.1  Estuary characteristics  5    3.2.2  Upstream river characteristics  5    3.2.3  Coastal characteristics  5    3.2.4  Salt intrusion  6    3.2.5  Solution  6    3.2.4  Salt intrusion  6    3.3  Land use  6    3.3.1  Land coverage  6    3.3.2  Flood protection measures  6    3.3.4  Ecology  8    3.3.5  Drinking water  9    Stakeholder Analysis  10    3.4.1  Agricultural and aquaculture sector  10    3.4.2  Citizens  10    3.4.3  Vietnamese government  11    3.4.4  Local authorities  11    3.4.5  Environmental organisations  11    3.4.4  Local authorities  11    3.4.4  Local authorities  11    3.4.5  Environmental organisations  11
4	Scen 4.1 4.2 4.3 4.4	nario selection14Criteria14Weather conditions14Alternatives15IPCC Climate change scenario16
5	Mod 5.1 5.2 5.3 5.4	lel  17    Model simplifications  17    Conceptual Model  17    Delft3D  17    Model settings  18    Model settings  19    5.4.1  Grid and bathymetry  19

		5.4.2	Simulation time	19
		5.4.3	Boundary conditions	19
		544	Initial conditions	20
		545	Thin dama	20
		5.4.5 E 4.6	Warra	20
		0.4.0	waves	20
	0.0	verinca		20
		5.5.1	Water level	20
		5.5.2	Discharge	21
		5.5.3	Salinity	22
	5.6	Sensiti	vity analysis	23
6	Res	ults		<b>25</b>
	6.1	Model	results	25
		6.1.1	Water levels	25
		6.1.2	Wave height	27
		6.1.3	Salinity	27
		6.1.4	Sediment	28
	6.2	Sedime	entation tool	30
7	Des	ign of t	flood protection system	<b>31</b>
	7.1	Dvkes		31
	72	Refere	nce projects	33
	1.2	7 2 1	Cai Lon - Cai Be sluice gates	33
		7.2.1	Ba Lai Bivor sluice	22
	79	Teeribi		00 94
	1.5	reasion	Inty study	34
	(.4 7 5	Techni		39
	7.5	Load c	ombinations	39
	7.6	Loads		41
		7.6.1	Hydrostatic loads	41
		7.6.2	Hydrodynamic loads	41
		7.6.3	Wind loads	42
		7.6.4	Soil loads	42
		7.6.5	Load schemes	43
	7.7	Struct	ural design	45
		7.7.1	Gates	45
		7.7.2	Sill	47
		773	Lifting structure	47
		7.7.4	Diore	18
		775	Foundation	40
		1.1.5		49
	-	(.(.)	Bed protection	49
	7.8	Operat	Jon	50
0	т	4 15	1	۳1
8	Imp	act Ev	aluation	51
	8.1	Criteri	a assessment	51
		8.1.1	Freshwater supply	51
		8.1.2	Agricultural and aquaculture adaptation	51
		8.1.3	Biodiversity	53
		8.1.4	Stable riverbanks	54
		8.1.5	Navigability	55
		8.1.6	Overview criteria assessment	55
	8.2	Criteri	a performance of the alternatives	56
	8.3	Cost a	- nalvsis	57
	8.4	Prefer	red solution	58
		_ 101011		50
9	Con	clusior	1	59
-				
10	Disc	cussion		60
	10.1	Limita	tion of the scope	60
	10.2	Uncert	ainties regarding collection of data	60

10.3 Model limitations	. 61 . 61 . 62
11 Recommendations	63
11.1 Scope recommendations	. 63 . 63 . 63
References	64
Appendices	69
Appendix A Field trip	69
Appendix B Data collection    B.1 Bathymetry    B.2 Upstream river discharge    B.2 Tide	71 . 71 . 71 . 71
B.4  Relative sea-level rise    B.5  Storm surge    B.6  Sediment characteristics    B.7  Wind conditions	. 74 . 74 . 74 . 75
B.7  Wind conditions    B.8  Wave climate    B.9  Salinity levels for agriculture	. 75 . 76 . 77
Appendix C Simplified cross-section	78
Appendix D Sediment Tool	80
Appendix E Calculations flood protection system    E.1 Dyke calculations    E.2 Hydrostatic load calculations    E.3 Hydrodynamic load calculations    E.4 Boreholes    E.5 Relationship between opening size and peak tidal flow during mean tide    E.6 Combined hydrodynamic and -static load on the gate    E.7 Load combination factors    E.8 Load schematisation on the gate    E.9 Euler buckling frame beams    E.10 Load schematisation and moments girders    E.11 Dimensions and internal bending moments of the storm surge barrier    E.12 The elevation of the water and barrier during operation    E.13 Moments and forces resulting from external loads    E.14 Top view storm surge barrier    E.15 Bed protection calculations	<b>81</b> . 81 . 82 . 91 . 92 . 92 . 92 . 94 . 95 . 96 . 97 . 98 . 100 . 100 . 101 . 102
Appendix F Meetings    F.1 Meeting SIWRP    F.2 Meeting Royal HaskoningDHV    F.3 Meeting ICOE	<b>104</b> . 104 . 107 . 109
Appendix G Survey Ben Tre	111

## List of Figures

T • T	Overview Mekong Delta	1
3.1	Overview of the main activities and processes in Ben Tre Province	4
3.2	A small sluice in a side channel	7
3.3	Ben Tre dyke protection and sluices	8
3.4	Mode of transport in Ben Tre from 2010 until 2014	8
3.5	Drinking water supply areas and pumping stations	9
3.6	Different ports in Ben Tre Province	12
3.7	Stakeholder Power Interest Matrix	13
4.1	Selection of locations of the closure, retrieved from OpenStreetMap	15
5.1	Conceptual Model	17
5.2	Delft3D processes within one time-step	19
5.3	Comparison of water level from datasets to output of Delft3D	21
5.4	Comparison discharge from dataset to output of Delft3D	22
5.5	Verification salinity	23
6.1	Water levels at observation point 7	25
6.2	Water levels at Observation Point 2	26
6.3	Significant wave height in the Ham Luong estuary during storm conditions in 2100	27
6.4	Maximum salt intrusion in 2100 during low discharge season in the Ham Luong River	28
6.5	Salinity levels on the sea side at the barier	28
6.6	Cumulative sedimentation/erosion without barrier	29
6.7	Cumulative sedimentation/erosion with barrier	29
6.8	Mud concentration in case of closed barriers.	30
7.1	Cross-section of the sea dyke design, including dimensions of the height and width in m	31
7.2	Dyke system	32
7.3	Cai Lon River barrier	33
7.4	Ba Lai River barrier	34
7.5	River barrier types	37
7.6	CPT at Vung Tau	43
7.7	Complete load schemes for all critical LCs	44
7.8	The design of the gate of the Hartelkering (Voorendt, 2014)	45
7.9	Top view of steel frame structure of the gate, showing the two arcs and nine horizontal girders.	46
7.10	Vertical girder geometry, deflection and stresses	47
7.11	Piled embankment with batter piles, indicated as 'inclined piles' (Sakleshpur & Madhav, 2013).	49
8.1	Visualisation of agricultural adaptation	52
8.2		02
	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay	55
A.1	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay    Field trip photos	55 70
A.1 B.1	Field trip photos     Bathymetry Ham Luong estuary, recorded in 2022	55 70 71
A.1 B.1 B.2	Field flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay    Field trip photos	55 70 71 72
A.1 B.1 B.2 B.3	Field flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay    Field trip photos    Bathymetry Ham Luong estuary, recorded in 2022    Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River    Discharge time series in the Ham Luong estuary	55 70 71 72 73
A.1 B.1 B.2 B.3 B.4	Field flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay    Field trip photos    Bathymetry Ham Luong estuary, recorded in 2022    Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River    Discharge time series in the Ham Luong estuary    Tidal signals	55 70 71 72 73 74
A.1 B.1 B.2 B.3 B.4 B.5	Field flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay     Field trip photos     Bathymetry Ham Luong estuary, recorded in 2022     Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River     Discharge time series in the Ham Luong estuary     Tidal signals     Wind conditions near Ham Luong River mouth	55 70 71 72 73 74 76
A.1 B.1 B.2 B.3 B.4 B.5 C.1	Field flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay     Field trip photos     Bathymetry Ham Luong estuary, recorded in 2022     Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River     Discharge time series in the Ham Luong estuary     Tidal signals     Wind conditions near Ham Luong River mouth     Exponential functions, fitted on the bathymetry data from the model	55 70 71 72 73 74 76 78
A.1 B.1 B.2 B.3 B.4 B.5 C.1 E.1	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay     Field trip photos     Bathymetry Ham Luong estuary, recorded in 2022     Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River     Discharge time series in the Ham Luong estuary     Tidal signals     Wind conditions near Ham Luong River mouth     Exponential functions, fitted on the bathymetry data from the model     Hydrostatic loads for different load combinations	55 70 71 72 73 74 76 78 83
A.1 B.1 B.2 B.3 B.4 B.5 C.1 E.1 E.2	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay     Field trip photos     Bathymetry Ham Luong estuary, recorded in 2022     Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River     Discharge time series in the Ham Luong estuary     Tidal signals     Wind conditions near Ham Luong River mouth     Exponential functions, fitted on the bathymetry data from the model     Hydrostatic loads for different load combinations     Wave forces on a closed structure for different load combinations	$52 \\ 55 \\ 70 \\ 71 \\ 72 \\ 73 \\ 74 \\ 76 \\ 78 \\ 83 \\ 86$
A.1 B.1 B.2 B.3 B.4 B.5 C.1 E.1 E.2 E.3	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay     Field trip photos     Bathymetry Ham Luong estuary, recorded in 2022     Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River     Discharge time series in the Ham Luong estuary     Tidal signals     Wind conditions near Ham Luong River mouth     Exponential functions, fitted on the bathymetry data from the model     Hydrostatic loads for different load combinations     Wave forces on a closed structure for different load combinations     Inertia and drag coefficients, based on Keulegan Carpenter Number	55 70 71 72 73 74 76 78 83 86 89
$\begin{array}{c} {\rm A.1} \\ {\rm B.1} \\ {\rm B.2} \\ {\rm B.3} \\ {\rm B.4} \\ {\rm B.5} \\ {\rm C.1} \\ {\rm E.1} \\ {\rm E.2} \\ {\rm E.3} \\ {\rm E.4} \end{array}$	Field flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay     Field trip photos     Bathymetry Ham Luong estuary, recorded in 2022     Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River     Discharge time series in the Ham Luong estuary     Tidal signals     Wind conditions near Ham Luong River mouth     Exponential functions, fitted on the bathymetry data from the model     Hydrostatic loads for different load combinations     Wave forces on a closed structure for different load combinations     Morison forces on the piers for different load combinations	55 70 71 72 73 74 76 78 83 86 89 91
$\begin{array}{c} {\rm A.1} \\ {\rm B.1} \\ {\rm B.2} \\ {\rm B.3} \\ {\rm B.4} \\ {\rm B.5} \\ {\rm C.1} \\ {\rm E.1} \\ {\rm E.2} \\ {\rm E.3} \\ {\rm E.4} \\ {\rm E.5} \end{array}$	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay     Field trip photos     Bathymetry Ham Luong estuary, recorded in 2022     Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River     Discharge time series in the Ham Luong estuary     Tidal signals     Wind conditions near Ham Luong River mouth     Exponential functions, fitted on the bathymetry data from the model     Hydrostatic loads for different load combinations     Wave forces on a closed structure for different load combinations     Morison forces on the piers for different load combinations     Soil layer distribution near Vung Tau	55 70 71 72 73 74 76 78 83 86 89 91 91
$\begin{array}{c} {\rm A.1} \\ {\rm B.1} \\ {\rm B.2} \\ {\rm B.3} \\ {\rm B.4} \\ {\rm B.5} \\ {\rm C.1} \\ {\rm E.1} \\ {\rm E.2} \\ {\rm E.3} \\ {\rm E.4} \\ {\rm E.5} \\ {\rm E.6} \end{array}$	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay     Field trip photos     Bathymetry Ham Luong estuary, recorded in 2022     Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River     Discharge time series in the Ham Luong estuary     Tidal signals     Wind conditions near Ham Luong River mouth     Exponential functions, fitted on the bathymetry data from the model     Hydrostatic loads for different load combinations     Wave forces on a closed structure for different load combinations     Morison forces on the piers for different load combinations     Soil layer distribution near Vung Tau     Location and soil layers of boreholes	55 70 71 72 73 74 76 78 83 86 89 91 91 92
$\begin{array}{c} {\rm A.1} \\ {\rm B.1} \\ {\rm B.2} \\ {\rm B.3} \\ {\rm B.4} \\ {\rm B.5} \\ {\rm C.1} \\ {\rm E.1} \\ {\rm E.2} \\ {\rm E.3} \\ {\rm E.4} \\ {\rm E.5} \\ {\rm E.6} \\ {\rm E.7} \end{array}$	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay    Field trip photos    Bathymetry Ham Luong estuary, recorded in 2022    Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River    Discharge time series in the Ham Luong estuary    Tidal signals    Wind conditions near Ham Luong River mouth    Exponential functions, fitted on the bathymetry data from the model    Hydrostatic loads for different load combinations    Wave forces on a closed structure for different load combinations    Morison forces on the piers for different load combinations    Soil layer distribution near Vung Tau    Location and soil layers of boreholes    Relationship between opening size and peak tidal flow during mean tide	55 70 71 72 73 74 76 78 83 86 89 91 91 92 92
$\begin{array}{c} {\rm A.1} \\ {\rm B.1} \\ {\rm B.2} \\ {\rm B.3} \\ {\rm B.4} \\ {\rm B.5} \\ {\rm C.1} \\ {\rm E.1} \\ {\rm E.2} \\ {\rm E.3} \\ {\rm E.4} \\ {\rm E.5} \\ {\rm E.6} \\ {\rm E.7} \\ {\rm E.8} \end{array}$	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay    Field trip photos    Bathymetry Ham Luong estuary, recorded in 2022    Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River    Discharge time series in the Ham Luong estuary    Tidal signals    Wind conditions near Ham Luong River mouth    Exponential functions, fitted on the bathymetry data from the model    Hydrostatic loads for different load combinations    Wave forces on a closed structure for different load combinations    Morison forces on the piers for different load combinations    Soil layer distribution near Vung Tau    Location and soil layers of boreholes    Relationship between opening size and peak tidal flow during mean tide    Hydrostatic and -dynamic Load LC1	55 70 71 72 73 74 76 78 83 86 89 91 91 92 92 93
$\begin{array}{c} {\rm A.1} \\ {\rm B.1} \\ {\rm B.2} \\ {\rm B.3} \\ {\rm B.4} \\ {\rm B.5} \\ {\rm C.1} \\ {\rm E.1} \\ {\rm E.2} \\ {\rm E.3} \\ {\rm E.4} \\ {\rm E.5} \\ {\rm E.6} \\ {\rm E.7} \\ {\rm E.8} \\ {\rm E.9} \end{array}$	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay     Field trip photos     Bathymetry Ham Luong estuary, recorded in 2022     Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River     Discharge time series in the Ham Luong estuary     Tidal signals     Wind conditions near Ham Luong River mouth     Exponential functions, fitted on the bathymetry data from the model     Hydrostatic loads for different load combinations     Wave forces on a closed structure for different load combinations     Morison forces on the piers for different load combinations     Soil layer distribution near Vung Tau     Location and soil layers of boreholes     Relationship between opening size and peak tidal flow during mean tide     Hydrostatic and -dynamic Load LC1	52 55 70 71 72 73 74 76 78 83 86 89 91 91 92 92 93 93
$\begin{array}{c} {\rm A.1} \\ {\rm B.1} \\ {\rm B.2} \\ {\rm B.3} \\ {\rm B.4} \\ {\rm B.5} \\ {\rm C.1} \\ {\rm E.1} \\ {\rm E.2} \\ {\rm E.3} \\ {\rm E.4} \\ {\rm E.5} \\ {\rm E.6} \\ {\rm E.7} \\ {\rm E.8} \\ {\rm E.9} \\ {\rm E.10} \end{array}$	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay    Field trip photos    Bathymetry Ham Luong estuary, recorded in 2022    Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River    Discharge time series in the Ham Luong estuary    Tidal signals    Wind conditions near Ham Luong River mouth    Exponential functions, fitted on the bathymetry data from the model    Hydrostatic loads for different load combinations    Wave forces on a closed structure for different load combinations    Morison forces on the piers for different load combinations    Soil layer distribution near Vung Tau    Location and soil layers of boreholes    Relationship between opening size and peak tidal flow during mean tide    Hydrostatic and -dynamic Load LC2    Hydrostatic and -dynamic Load LC5	52 55 70 71 72 73 74 76 78 83 86 89 91 91 92 92 93 93 94
$\begin{array}{c} {\rm A.1} \\ {\rm B.1} \\ {\rm B.2} \\ {\rm B.3} \\ {\rm B.4} \\ {\rm B.5} \\ {\rm C.1} \\ {\rm E.1} \\ {\rm E.2} \\ {\rm E.3} \\ {\rm E.4} \\ {\rm E.5} \\ {\rm E.6} \\ {\rm E.7} \\ {\rm E.8} \\ {\rm E.9} \\ {\rm E.10} \\ {\rm E.11} \end{array}$	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay    Field trip photos    Bathymetry Ham Luong estuary, recorded in 2022    Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River    Discharge time series in the Ham Luong estuary    Tidal signals    Wind conditions near Ham Luong River mouth    Exponential functions, fitted on the bathymetry data from the model    Hydrostatic loads for different load combinations    Wave forces on a closed structure for different load combinations    Morison forces on the piers for different load combinations    Soil layer distribution near Vung Tau    Location and soil layers of boreholes    Relationship between opening size and peak tidal flow during mean tide    Hydrostatic and -dynamic Load LC1    Hydrostatic and -dynamic Load LC5    Hydrostatic load on the gate	52 55 70 71 72 73 74 76 83 86 89 91 92 93 93 94 95
$\begin{array}{c} {\rm A.1} \\ {\rm B.1} \\ {\rm B.2} \\ {\rm B.3} \\ {\rm B.3} \\ {\rm B.5} \\ {\rm C.1} \\ {\rm E.1} \\ {\rm E.2} \\ {\rm E.3} \\ {\rm E.4} \\ {\rm E.5} \\ {\rm E.6} \\ {\rm E.7} \\ {\rm E.8} \\ {\rm E.9} \\ {\rm E.10} \\ {\rm E.11} \\ {\rm E.12} \end{array}$	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay    Field trip photos    Bathymetry Ham Luong estuary, recorded in 2022    Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River    Discharge time series in the Ham Luong estuary    Tidal signals    Wind conditions near Ham Luong River mouth    Exponential functions, fitted on the bathymetry data from the model    Hydrostatic loads for different load combinations    Wave forces on a closed structure for different load combinations    Morison forces on the piers for different load combinations    Soil layer distribution near Vung Tau    Location and soil layers of boreholes    Relationship between opening size and peak tidal flow during mean tide    Hydrostatic and -dynamic Load LC1    Hydrostatic load on the gate	555 70 71 72 73 74 76 78 83 86 89 91 92 92 93 94 95 95
$\begin{array}{c} {\rm A.1} \\ {\rm B.1} \\ {\rm B.2} \\ {\rm B.3} \\ {\rm B.3} \\ {\rm B.4} \\ {\rm B.5} \\ {\rm C.1} \\ {\rm E.1} \\ {\rm E.2} \\ {\rm E.3} \\ {\rm E.4} \\ {\rm E.5} \\ {\rm E.6} \\ {\rm E.7} \\ {\rm E.8} \\ {\rm E.9} \\ {\rm E.10} \\ {\rm E.111} \\ {\rm E.122} \\ {\rm E.13} \end{array}$	Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay    Field trip photos    Bathymetry Ham Luong estuary, recorded in 2022    Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River    Discharge time series in the Ham Luong estuary    Tidal signals    Wind conditions near Ham Luong River mouth    Exponential functions, fitted on the bathymetry data from the model    Hydrostatic loads for different load combinations    Wave forces on a closed structure for different load combinations    Morison forces on the piers for different load combinations    Soil layer distribution near Vung Tau    Location and soil layers of boreholes    Relationship between opening size and peak tidal flow during mean tide    Hydrostatic and -dynamic Load LC1    Hydrostatic load on the gate    Hydrostatic load on the gate	52 55 70 71 72 73 74 76 83 86 89 91 92 93 93 94 95 95 95

E.15	Loads and moment vertical girder	7
E.16	Front view of the storm surge barrier	8
E.17	Cross section with loads of the storm surge barrier for LC1	8
E.18	Cross section with loads of the storm surge barrier for LC2	9
E.19	The heights of the water and barrier during operation and rest	0
E.20	Top view of the storm surge barrier	1
E.21	$\alpha$ value for scour protection	2
E.22	$K_v$ coefficient behind sill	3
F.1	Presentation at SIWRP	5
F.2	Group photo with Bao in front of the SIWRP building	6
F.3	Interaction and discussion during the meeting at Royal HaskoningDHV	8
F.4	Presentation from ICOE concerning the Mekong Delta	9
F.5	Group photo with the employees of ICOE	0

## List of Tables

3.1	Economic sector distribution in Ben Tre Province	0
3.2	Overview of the stakeholders with their interests 1	3
4.1	Critical situations	.5
5.1	Overview of the 24 simulations run with Delft3D	.8
6.1	Total sedimentation layer after 5 months	30
7.1	Criteria scores for water control structures	38
7.2	Weight for each criterion	38
7.3	Score with weight per gate types, where $W = Weight$ and $S = Score. \dots \dots$	39
7.4	List of technical requirements	\$9
7.5	Load combinations and conditions	1
7.6	CPT results Vung Tau	13
7.7	Soil distribution based on CPT data 4	13
7.8	Properties of arc and girder beams 4	6
7.9	Girder buckling check	6
8.1	Impact evaluation agricultural adaptation 5	53
8.2	Alternatives evaluation	55
8.3	Weight determination of the criteria	6
8.4	Weighted scores for alternatives, where $W = Weight$ and $S = Score. \dots \dots$	6
8.5	Coastal defence costs for a km stretch 5	57
8.6	Initial costs of the different alternatives	6
8.7	Maintenance costs of the different alternatives per year	68
8.8	Total costs of the different alternatives for 2100 forecast	<b>i</b> 8
B.1	Climate change predictions on the discharge of the Mekong River	$^{\prime}2$
B.2	Most relevant tidal constituents	'4
B.3	Suspended sediment concentrations	$^{\prime}5$
B.4	Wave climate data measured at Bach Ho station from 1986-2006	76
C.1	Results and verification cross-section analysis	'9
D.1	Input for sedimentation tool	30
E.1	Maximum and minimum water levels in 2100	32
E.2	Drag and inertia coefficients	39
E.3	Internal forces acting on the storm surge barrier	)9
E.4	Bending moments and horizontal forces per LCs, resulting from the load schematisation 10	)()

## List of Abbreviations

ADB	Asian Development Bank	
$\mathbf{AR}$	Assessment Report	
$\mathbf{BwN}$	Building with Nature	
CPT	Cone Penetration Test	
CRWES	Center of Rural Water and Environment Sanitation	
DONRE	Department of Natural Resources and Environment	
GRDP	Gross Regional Domestic Product	
HAT	Highest Astronomical Tide	
HCMC	Ho Chi Minh City	
ICOE	Institute of Coastal and Offshore Engineering	
IPCC	Intergovernmental Panel on Climate Change	
IUCN	International Union for Conservation of Nature	
JONSWAP	Joint North Sea Wave Project	
KC	Keulegan–Carpenter	
LAT	Lowest Astronomical Tide	
$\mathbf{LC}$	Load Combination	
LMD	Lower Mekong Delta	
MARD	Ministry of Agricultural and Rural Development	
MCA	Multi-Criteria Analysis	
MONRE	Ministry of Natural Resources and Environment	
MRC	Mekong River Commission	
MSL	Mean Sea Level	
NE	North-East	
RCP	Representative Concentration Pathway	
RSLR	Relative Sea-Level Rise	
SIWRP	Southern Institute for Water Resource Planning	
SIWRR	Southern Institute of Water Resource Research	
SLR	Sea-Level Rise	
$\mathbf{SW}$	South-West	
SWAN	Simulating WAves Nearshore	
UC	Unity Check	
USAID	U.S. Agency for International Development	
VND	Vietnamese Dong	
WSC	Water Supply One-member Limited Liability Company	
WWF	World Wildlife Fund	

## 1 Introduction

The Mekong Delta is home to over 18 million people, adding up to 19% of the total population in Vietnam and it is also the area where the Mekong River meets the South China Sea. The delta consists of two major rivers: the Hau River (also known as Bassac) and the Tien River (also known as Mekong), and is made up of eight major estuary branches. These can be seen in Figure 1.1. The Mekong River is a crucial source for irrigation, freshwater supply, and sediment supply in the delta. Intensive rice production, aquaculture, and fruit production in the Mekong Delta contribute largely to food security in Vietnam (Woillez & Espagne, 2022).



Figure 1.1: Overview Mekong Delta (The Asia-Pacific Journal, 2010), (Kuenzer et al., 2013)

#### 1.1 Challenges

The devastating effects of climate change have started to become more and more evident over the last few decades, and unfortunately the Mekong Delta is no exception. In fact, the Mekong Delta is considered to be one of the most vulnerable deltas worldwide subject to the effects of climate change. Especially changes in river runoff, Sea-Level Rise (SLR) and land subsidence are expected to affect the delta in multiple ways:

- 1. The increase in river discharge during the wet season results in regular flooding of the rivers. Due to urbanisation, these river floods threaten the safety of inhabitants (Netherlands Advisory Team of the Mekong Delta Plan, 2013).
- 2. The decrease in river discharge during the dry season results in a shortage of fresh water required for irrigation and drinking water.
- 3. The combination of SLR, subsidence of the ground and decreased river discharge during the dry season results in saltwater intrusion, which influences freshwater supply and consequently increases groundwater salinity.
- 4. The rising sea level, in combination with the occurrence of storm surges as a result of typhoon storms will cause high inundation risks for the low-laying Mekong Delta. In the case of a SLR of 100 cm, 22% of the region of Ben Tre would be inundated (Tran Thuc et al., 2016).

As a response to these changes, the Dutch Advisory Team for the Mekong Delta Plan has drafted a master plan for the Mekong Delta; The Mekong Delta Plan. This plan is to be used as a guideline by any party involved in the sustainable development of the Mekong Delta (Netherlands Advisory Team of the Mekong Delta Plan, 2013).

#### 1.2 Research objective

The Mekong Delta Plan proposes a set of measures which could counteract the unwanted effects of climate change threatening the Mekong Delta. Amongst these measures is the closure of the Ham Luong estuary, one of the six branches of the Tien River. Additionally, the Ministry of Agricultural and Rural Development (MARD) of Vietnam also suggests to close off the Ham Luong estuary by means of a barrier to prevent salt-water intrusion (Ministry of Agriculture and Rural Development & Southern Institute for Water Resource Planning, 2021). The permanent or temporal closure of the estuary will prompt the system to adapt to the new hydrodynamics, which will greatly impact many facets of the estuary. These impacts have not yet been investigated thoroughly, and are therefore largely unknown. Moreover, there are many options for the location and the design of a closure. This has led to the formulation of the main research question of this report.

"What is the impact of various closure alternatives on the hydraulic characteristics and social activities in the Ham Luong estuary, considering a 75-year forecast?"

The main goal of this research question is to analyse the impact of proposed alternatives, which can support local authorities in their decision-making. The goal of this report is to build a basis for transparent and straightforward decision-making of the preferred solution, based on prioritisation of the assessed impact on different aspects of the system. The spatial scope of this report is restricted to the area starting in the North at the bifurcation of the Ham Luong River from the Mekong River to the coastal zone in the South Chinese Sea. To address the main research question, it has been subdivided into multiple sub-questions:

- 1. What are the key characteristics of the Ham Luong estuary considering economic and social activities and hydro-morphodynamic properties?
- 2. What is the impact of the partial closure of the Ham Luong estuary on the hydro- and morphodynamics of the system?
- 3. What would be the optimal design for a flood protection system?
- 4. Considering the effect of the hydro-morphodynamic changes on the social and economic activities, what decision in terms of closure of the estuary can be regarded as the best fitting solution for the system?

#### 1.3 Report structure

The report is structured as follows. In Section 2 the methodologies which were used in this report are presented. Next, a concise description of the system is provided in Section 3. This is followed by the selection of the scenarios that will be evaluated in Section 4. Following that, the conceptual model and the Delft3D model are introduced in Section 5. The results of the model in terms of water level, salinity and morphology are presented in Section 6. Additionally, this section refers to the results of a tool to assess additional morphodynamic changes. Subsequently, Section 7 dives into the design of the flood protection system which consists of the most optimal design of a river barrier and the design of the adjacent dyke system for each of the alternatives. In Section 8 the impacts of the different alternatives are compared with a Multi-Criteria Analysis (MCA). Additionally, a cost analysis is performed to assess the cost efficiency of the different alternatives. Finally, Section 9, Section 10 and Section 11 contain the conclusion, discussion and recommendations for this project.

## 2 Methodology

This section contains clarification on the approaches that were applied to eventually answer the research questions posed in Section 1. To begin with, an extensive analysis of the system was conducted. This was followed by the translation of the system of interest into a model. This model was in turn used to evaluate the effect of different closure scenarios on the hydraulic and morphodynamic characteristics of the estuary. Next, the the design of the flood protection system was elaborated. Finally, the changes in hydromorphodynamics and the design of the flood protection system were used as input for the MCA, which generated a best fitting solution.

#### 2.1 Analysis of the system

In order to get a thorough understanding of the system of interest and to define the importance of different elements, a system analysis was conducted. This "System description" contains comprehensive area studies and a stakeholder analysis. By means of literature research, online research, visual research during the field trip, and a stakeholder survey, the information was collected for the chosen area of interest and other factors associated to that area. Here, a multidisciplinary approach was necessary to tackle the complexity of the system. The gathered information was used to map the different stakeholders, usage of land, and other important processes within the system.

#### 2.2 Approach scenario selection

The knowledge gained through the system analysis was used to formulate criteria by which the impact on the system can be assessed. Having selected these criteria, the different modelling scenarios were identified, starting with the seasonal and weather conditions that are most crucial to the system in 2100. Next, the possible locations of the closure were selected. Finally, a decision was made on which Intergovernmental Panel on Climate Change (IPCC) climate change scenario was used throughout the report.

#### 2.3 Modelling process

The effect of the closure on the hydro- and morphodynamics was studied using the modelling suite Delft3D. First, a conceptual model was created in which the system of interest was translated into key concepts and relationships. Next, numerical data was collected through various sources, of which Thuyloi University, SIWRP and ICOE were the main providers. These datasets were modified to useful data with Excel and Python.

From the conceptual model and the data collection the boundary conditions, initial conditions and bathymetry were derived. Then, the model was calibrated and verified using provided measurement data. This was followed by a sensitivity analysis, in which the effect of changing certain parameters was evaluated. Finally, the model was run for different weather and seasonal combinations and different closure alternatives, which were defined in the section on Scenario Selection.

#### 2.4 Approach design of flood protection system

In this section, the design of the flood protection system is presented. First, a preliminary design was made for the adjacent river dyke system for each of the alternatives. The water level data from Delft3D and the data collection were used for this. Furthermore, a literature study on reference projects and barrier types for a storm surge barrier was performed. After that an MCA was conducted to find the best fitting gate type for the storm surge barrier. The different proposed barrier types were rated with weights on various evaluation criteria. For the resulting barrier type a preliminary design calculation was made. This was a modular design. This design consisted of gates, a sill, lifting structure, piers, foundation and bed protection. Furthermore, a short description of the operation of the gates was made for future commissioning.

#### 2.5 MCA approach

To decide on the best fitting solution for the system, an MCA was used to compare the different alternatives. Each of the criteria, which were identified in the scenario selection, were given a weight of importance. Then, the alternative options and their relative rating for each criterion were displayed in an MCA-matrix. By multiplying the rating with the weight per criteria, the total ratings per alternative were calculated. After a comparison, the best option was defined.

## 3 System description

As mentioned in the Section 1, the spatial scope of this project is restricted to the area surrounding the Ham Luong estuary. The area which is affected by the river branch is outlined in Figure 3.1. It is located in the southern part of Vietnam, in the province of Ben Tre, where the Ham Luong estuary flows into the South China Sea as a branch of the Tien River. The largest city near the river is the city of Ben Tre with nearly 125,000 inhabitants and the overall area is sparsely populated. The coastal area connected to this estuary is also considered in this project. In this section the various characteristics of the area will be discussed. The aspects to be considered are divided into four main topics: local climate, hydraulic characteristics, land use, and stakeholders. Figure 3.1 shows a brief overview of the main activities in Ben Tre, which will be further explained in the remainder of this section.



Figure 3.1: Overview of the main activities and processes in Ben Tre Province

#### 3.1 Local Climate

Ben Tre Province has a tropical monsoon climate with a wet and a dry season. The average annual temperature in the Mekong Delta over the past two decades was 27°C (WorldData.info, n.d.). The wet season in Ben Tre extends from May through November, while the dry season lasts from December to April. During the wet season typically 80% to 85% of the annual precipitation rate is recorded. Over the past two decades, the average annual rainfall has amounted up to 1,733 mm (VESDEC, 2020). During the wet season, the prevailing monsoon is from the South-West (SW), whereas in the dry season, the North-East (NE) monsoon dominates. The NE monsoon winds are dominant over the SW winds in terms of magnitude and frequency. Besides monsoon winds, typhoons occur regularly in the South China Sea and have an impact on the Vietnamese mainland. Typically, the highest wind speeds occur during typhoons. In the period of 1952 to 2020, a total of 51 typhoons were observed along the Vietnamese coast (Thuyloi University, 2020).

Climate change projections indicate that by 2100, the average annual temperature and annual rainfall will have increased significantly compared to the reference period of 1986-2005 (Tran Thuc et al., 2016). Additionally, the Ministry of Natural Resources and Environment (MONRE) reports suggest that 5-day extreme rainfall events will intensify by 60% by 2100 in contrast to the 1986-2014 base period. Temperature is expected to increase

the same amount as the global average, with a range between  $1.0^{\circ}$  and  $3.4^{\circ}$  (World Bank Group & Asian Development Bank, 2021). The forecasts from MONRE do not indicate significant changes in the frequency and intensity of typhoons (Tran Thuc et al., 2016).

#### 3.2 Hydraulic characteristics

Hydraulic characteristics refer to physical properties and behaviour of water flow. These characteristics can differ depending on the location in the estuary. For this reason, a distinction is made between three areas; the estuary, the upstream part of the river, and the coastal area. Additionally, salt intrusion, which covers all three areas, is discussed.

#### 3.2.1 Estuary characteristics

The Ham Luong estuary is approximately 70 km long. This is measured from the bifurcation at the most north-western part of the province of Ben Tre until the coast. The estuary is very wide; at some points the estuary width can be up to 2.7 km, measured in Google Maps. The depth of the bed of the Ham Luong estuary is considered shallow, but the exact depths may vary along the estuary. The Mekong Delta is an extremely flat and low delta area. The slope towards the sea has a gradient of 5 mm/km (Tromp, 2013). In the Ham Luong River eight small islands are present. The largest one is approximately 9 km in length and 1 km in width. The shapes of the islands change slightly over time due to erosion and sediment deposition in the river. These river islands are used mainly for aquacultural activities such as fishing ponds, but also contain resorts and living areas.

Within the estuary arm, several channels are connected to the Ham Luong River. Two main side channels can be identified. These channels are connecting the Ham Luong River to the Tien River in the North (Ben Tre Channel) and the Chien River in the South (Mo Cay Channel). Not much data is available about the discharge in these connecting channels. The connecting channels are water bodies with high sedimentation rates, as the flow velocities are much lower than the flow velocities in the main branches (Tromp, 2013). This implies that the discharge is significantly smaller than in the main branch.

The river naturally flushes sediment towards the sea, but also loses sediment in sand mining processes. Certain areas have been appointed for legal sand mining for limited volumes (Ministry of Agriculture and Rural Development, 2016). Throughout the whole Mekong Delta, there are illegal sand mining processes, which can have a significant impact on the riverbank erosion. Due to the incision of the riverbed by sand mining, an increase in tidal amplification can be observed (Loc et al., 2021).

The soil in the Mekong Delta is extremely acidic at many locations, caused by the deposition of pyritic sediments from the upstream river. When this sediment, which contains iron disulfide, is exposed to oxygen it acidifies the soil. The depth at which the acid soil is found, is often within 1 m of the surface. This is primarily a problem in the upper Mekong River Delta, but also in the lower delta where it is difficult to drain water. In Ben Tre, it is estimated that around 1.3% of the total land area is contaminated with acid soil, which is relatively small compared to other areas (Ministry of Agriculture and Rural Development, 2016).

#### 3.2.2 Upstream river characteristics

To describe the upstream river characteristics, the whole catchment area of the Mekong River needs to be taken into account. Precipitation trends in the catchment area exert a significant influence on the flow regime. During the wet season, there is a high discharge peak, while in the dry season, the lowest discharge is observed. The high and low discharge seasons in the Ham Luong estuary demonstrate a time lag of a few months with respect to the wet and dry season (Nguyen Trug Viet et al., 2006). The highest upstream discharge can be observed in August and September, while the smallest discharge occurs in April and May (Kummu et al., 2008).

Upstream of the Ham Luong estuary, in Cambodia, Laos and China, multiple hydropower dams have the potential to withhold water and sediment that would originally direct flow towards the lower Mekong Basin (MRC, n.d.). Due to a lack of administration and available data, and the large size of the Mekong River system, it is very difficult to point out the effects of these dams (Adamson et al., 2009). The same can be said about upstream sand mining, which is difficult to monitor and prove. Currently, the effects of these interventions in the upstream river on the discharge and sediment flow cannot accurately be defined.

#### 3.2.3 Coastal characteristics

The downstream boundary of the system of interest is formed by the coastal area at the mouth of the Ham Luong estuary. According to the Process Framework developed by Galloway (1975), the delta can be classified as a tide- and wave-dominated delta. This entails that the delta is subject to the influence of both tides and waves, which in turn influence how the delta develops in terms of morphology.

The wave climate near the Ham Luong estuary is strongly influenced by the behaviour of the winds that prevail there. As stated in Section 3.1, the most important winds that can be distinguished occur during the NE monsoon and the SW monsoon. Waves created by these monsoon winds are similar to swell waves (Hoan Van Huan & Nguyen Huu Nhan, 2006).

The water level at the coast consists of three contributors: the tidal range, storm surge and RSLR, including absolute SLR and land subsidence. At the coastline of the Mekong Delta, the tidal regime is characterised as "mixed, more semi-diurnal". Storm surges are considered a dangerous phenomenon along the Vietnamese coastline. Storm surge is the piling up of the water surface as a result of friction between high wind speeds, during typhoons, and the water surface (Truong Hong Son, 2012).

#### 3.2.4 Salt intrusion

As mentioned before, a large consequence of climate change is the saltwater intrusion in the Ham Luong River. This saltwater intrusion has an extensive impact on the area. Increased salinity levels influence the type of agriand aquaculture that can be practised in the areas near the river and put pressure on the availability of fresh water for domestic use (Toan, 2014). The saltwater intrusion is caused by a combination of reduced river flow (due to droughts induced by climate change, and the construction of hydropower dams upstream) and absolute SLR. Additionally, the aforementioned increase in tidal amplification due to sand mining also gives rise to an increase in salt water in the Ham Luong estuary.

In previous research, the intrusion of salt was investigated in the dry years of 1998, 2010, 2016, and 2020. Figure 3.1, at the beginning of this Section 3, shows the areas affected by salt intrusion in these years. The figure shows that the salt intrusion has reached further inland into the Mekong Delta in recent years. It can be seen that the 4 g/L salinity isoline of 2020 touches the bifurcation of the Tien River and the Ham Luong River, 70 km inland. This shows that the whole area surrounding the Ham Luong River is affected by salt intrusion.

#### 3.3 Land use

The province of Ben Tre has a wide variety of land use types. In assessing the impact of the closure of the Ham Luong estuary, it is crucial to be aware of these different types of land use and the various elements in the system. Therefore, this section focuses on the different aspects of land utilisation, including land coverage, current flood protection measures, shipping, ecology, and drinking water.

#### 3.3.1 Land coverage

An overview of the land use in Ben Tre Province can be seen in Figure 3.1. The most common kinds of food produced in Ben Tre are fruits, coconut, rice and shrimps. Of these products, coconut and shrimps are the most cost-beneficial, while rice is the least cost-beneficial. The fruit plants are located more inland due to their low resistance to salt water. Coconut plants are more resistant to salt water than other fruit plants and are thus planted more seawards. The same goes for shrimp farms which use salt water. A rice field that is harvested one time a year is called a 1-rice paddy and a rice field that is harvested two or three times a year is 2/3-rice paddy. In the delta, 2/3-rice paddy is harder to harvest because therefore it should also grow in the dry season and rice is not very resistant to salt. An option is to alternate between rice and shrimp production using flood-based agri- and aquaculture. Rice can grow in the fields during the wet season, and during the dry season, when the water is saltier, shrimps can grow at the same location. Currently, most shrimps are produced in coastal areas with a high salinity level. Close to the coastline, mangrove forests thrive.

#### 3.3.2 Flood protection measures

To protect the delta region from floods, different types of flood protection measures have been implemented in the area at the coast including mangroves. Mangroves are trees that grow in tidal waters and can trap sediment, preventing erosion, while they protect the low dykes from direct sea impact (Groenewold et al., 2018). The mangroves grow in the area naturally. Due to the decline of mangrove forests, the protections in front of the dykes are weakened. In the province of Ben Tre there are low sea dykes present. These are simple earth dykes which are mostly covered in grass (Cong San et al., 2017). To prevent the coastline from receding, several small-scale erosion measures have been taken along the coast. These handmade structures use materials such as wood and sandbags, which are also often used to repair sea dyke breaches.

Besides coastal protection, there are also protection measures present along the river banks of the Ham Luong River. An extensive protection system is present to prevent salinisation in the area (Käkönen, 2008). The dykes around the Ham Luong estuary are about 3.0 m in height and 5.0 m in width, with an inner slope of 1:2 and an outer slope of 1:3 (Institute of Coastal and Offshore Engineering, 2020). The dyke height is with respect to the Vietnam Datum 2000 reference system. This is almost the same as with respect to Mean Sea Level (MSL) but it can differ for various locations in Vietnam. For this project it is assumed that it is approximately the same as to MSL. Along the river banks, mangrove trees also grow. However, these are not large forests such as along the coast. The mangrove trees only grow a couple tens of metres inland. In Figure 3.1 the positions of the dykes and mangrove forest are shown.

There several are other river barriers built in the area. An example in Ben Tre Province is the Ba Lai River barrier. This is a relatively small barrier consisting of a dam and a gated section. Furthermore, in the Mekong Delta there is also a large barrier called the Cai Lon barrier. It is located in the west of the Mekong Delta in the province of Kien Giang and is a storm surge barrier with gates over the entire length of the river. Both barriers are built to prevent salt intrusion in the area. More information on them can be found in Section 7.2.

Along the Ham Luong River there are also numerous small sluice gates, such as shown in Figure 3.2. These can block off channels to prevent flooding and salt intrusion. There are plans to build more sluices and their locations are visible in Figure 3.3. The location of the dykes are also visible in this figure.



Figure 3.2: A small sluice in a side channel, located 35 km upstream of the Ham Luong River mouth. The photo was made from the Kinh Bridge during the field trip.



Figure 3.3: Ben Tre dyke protection and sluices (Korbee et al., 2019)

#### 3.3.3 Shipping

While the primary shipping route from Phnom Phen to Ho Chi Minh City is not through the Ham Luong River (Shibasaki et al., 2021), there are several small boats making use of the river. The Mekong Delta has more transportation over water than the rest of Vietnam due to investments in channels during the French colonial rule (Ministry of Agriculture and Rural Development, 2016). In Ben Tre, more goods are transported by inland waterways than by road, but inland waterway transportation has been decreasing as a result of investments in roads and bridges. For public transport, more passengers are using the road instead of waterways. The numbers per transport mode are shown in Figure 3.4. Furthermore, it is important to note that tourism vessels are increasing in size due to the increase in popularity. These vessels also travel on the Ham Luong River (VietnamPlus, n.d.).



Figure 3.4: Mode of transport in Ben Tre from 2010 until 2014 (Ministry of Agriculture and Rural Development, 2016)

#### 3.3.4 Ecology

Ecology describes relationships between the environment and living organisms, usually of a particular area. This is a very broad subject, hence only the main considered ecosystems will be described here. According to the Ministry of Agriculture and Rural Development (2016) the most important ecosystems are fish, birds

and mangroves. In general in the Mekong Delta, the mammal fauna is not diverse, especially when compared to its fish fauna (McElwee & M Horowitz, 1999). The large diversity of fish is caused by the combination of fresh, brackish and salt water. Some rare species are even on the International Union for Conservation of Nature (IUCN) Red List, published in 2011. These fish migrate from the estuary into fresh water in higher areas (Ministry of Agriculture and Rural Development, 2016). Of the fish species with a known migration pattern, 165 out of 189 migratory fish species are long-distance migrants. Furthermore, it is expected that 90% of the fish species get migration cues when there is a variation in water level or discharge (Brian Eric, 2006).

There are a lot of birds in the Mekong Delta. A protected bird sanctuary in Ben Tre, called the Vam Ho Bird Sanctuary, is located next to the Ba Lai River. Another protected area is Than Phu Nature Reserve, which encloses the mangrove forests near the coast in the Than Phu district in Ben Tre.

Mangroves grow in saline and brackish water where the sea meets the coast in (sub)tropical environment. The specific conditions created by tidal fluctuations, with their interplay of salt and fresh water, reduces the competition among other plant species. This is a significant benefit for mangrove forests (University of Florida, 2023). There are sixteen mangrove species in Ben Tre; two of which grow along the Ham Luong estuary. These particular species are not protected by law or regulations.

The Department of Natural Resources and Environment (DONRE) regularly tests water quality by taking samples at different locations in Ben Tre Province. More than half of the water samples was heavily contaminated and classified as level 5 out of 5, indicating measures should be taken. Only 30% of the water samples were classified as level 1 and 2, which indicates water that can be used in daily activities (Ministry of Agriculture and Rural Development, 2016). This means that the living conditions of animals in the water are threatened by poor water conditions.

#### 3.3.5 Drinking water

There are two organisations that provide the area with fresh water. The Water Supply One-member Limited Liability Company (WSC) provides the city of Ben Tre and its surrounding area with fresh water. They operate four water supply plants, which are Son Dong, Huu Dinh, Cho Lach, and Luong Quoi. The supply capacity of the pumps is 52,800 m<sup>3</sup>/day in total (Ministry of Agriculture and Rural Development, 2016). The regions that are used for water supply and the locations of the plants of the WSC and the Center of Rural Water and Environment Sanitation (CRWES) can be seen in Figure 3.5. Besides these water supply plants, several rural water supply plants exist and they account for a total water supply of  $1,583 \text{ m}^3/\text{day}$ .



Figure 3.5: Drinking water supply areas and pumping stations (Ministry of Agriculture and Rural Development, 2016)

The saltwater intrusion in the river has led to difficulties in freshwater supply in Ben Tre. It affects the channels in the province as well as water pumps. At the Huu Dinh water supply plant the pumping has stopped because the borehole was too saline. Furthermore, salt water causes corrosion to pipes that contain fresh water, damaging the water supply system (Ministry of Agriculture and Rural Development, 2016). From the field trip it was also found that many people extract their fresh water from the river for domestic use, which is not an option if the water in the river is too saline. More information about the field trip can be found in Appendix A.

#### 3.4 Stakeholder Analysis

Within this section, attention is paid to the most important stakeholders involved in this project. It provides insights into each stakeholder, including their concerns, needs, and the extent of their influence. It concludes with an overview table with each stakeholder and their interests, in addition to a Power Interest Matrix.

#### 3.4.1 Agricultural and aquaculture sector

The agricultural and aquaculture sectors surrounding the Ham Luong estuary involves a variety of farmers. Figure 3.1 shows how the land is used in Ben Tre Province. Vietnam plays a significant role in global rice production, contributing approximately 8% of the world's total rice output, which is equal to 42 million tonnes. Within Vietnam, the Mekong Delta produces nearly 25 million tonnes and is therefore regarded as the heart of rice production in the country (Food and Agriculture Organization of the United Nations, 2018). One hectare of rice has an average yearly net income in 2015 of 16 million Vietnamese Dong (VND). Vietnam also produces 37% of the total world production of shrimp, with the production in the delta located near the coast. One hectare of shrimp has a yearly average net income in 2015 of 45 million VND. In addition to this is, it important to mention that shrimp cultivation comes with significantly higher costs than other farming practices. Approximately 30% of the total land area is used for fruit cultivation. Of the fruit production in the Ben Tre Province, the output is mostly distributed via middlemen, who distribute it to supermarkets in various provinces in Vietnam. Therefore, the exact quantities are uncertain. Ben Tre Province is seen as the most important area for coconut production and the share of coconut production has been increasing recently. One hectare with coconuts has a yearly average net income in 2015 of 70 million VND. Coconuts are also processed into higher value products, like coconut oil. In 2016, 526,000 tonnes of coconut were produced, of which 90% was exported (Ministry of Agriculture and Rural Development, 2016). Farmers change from producing rice to producing coconut due to their high resistance to salinity, smaller labour intensity, and high market price. Changes from rice production to shrimp farming are also observed.

In Ben Tre Province, 60% of the labour force is active in the agricultural or aquaculture sector. Their primary requirement is water supply. Depending on the product to be cultivated, brackish water or fresh water is preferred (Ministry of Agriculture and Rural Development, 2016). Fresh water availability in the future is their biggest concern. If the salinity levels increase, farmers would be forced to shift to different farming methods or other products. Their interest is to have ideal conditions to grow their products, to sell as many products as possible. Combining this with the high production rate, the agricultural sector is a very important stakeholder. While they do not have much influence on decision making, they are vital for the economy of Ben Tre.

#### 3.4.2 Citizens

The province of Ben Tre is mainly inhabited by the Kinh people, although there are other minority groups including Chinese, Muong Tay, Khmer and Cham people. They all communicate in Vietnamese and there are little differences observed between the different groups (Ministry of Agriculture and Rural Development, 2016). The citizens work in different sectors, of which the agricultural and aquacultural economy is the largest. The service sector is also growing, taking up approximately 31% of the income of the province. The industry and construction sector are the smallest economy with only approximately 21% of the Gross Regional Domestic Product (GRDP). Currently with 15% of the workforce, the industry sector has grown vastly in recent years. Industries such as agricultural and aquatic product processing, paper production, wind energy, garments and mechanical engineering are the most developed industries (Ministry of Agriculture and Rural Development, 2016). These numbers are summarised in Table 3.1.

Table 3.1: Economic sector distribution in Ben Tre Pro-	vince
---	-------

Economic sector	Percentage of GRDP	Percentage of workforce
Agri- and aquaculture	44%	60%
Services	31%	22%
Industry and construction	21%	15%

The citizens are interested in the development and safety of their city. While their perspectives are more frequently being considered during stakeholder meetings (Ta Quynh Hoa et al., 2019), citizens still tend to think that these gatherings primarily serve as informative sessions to present the new plans, rather than an opportunity for them to exert influence. The reason for this is the lack of discussion with the stakeholders, which results in a low participation level. The local authorities sometimes attribute this to the low education level of the citizens (Andrew-Wells Dang et al., 2015). Therefore the influence of the citizens is relatively low compared to the other stakeholders, even though their interest is very high.

#### 3.4.3 Vietnamese government

The Vietnamese government eventually decides which projects are carried out and is responsible for the safety and prosperity of Vietnamese citizens. The Mekong Delta is a region of economic importance due to its agricultural and aquaculture activity. Keeping this region safe and productive is the main goal of the Vietnamese government, while their concern is economic decline. There are a lot of different ministries that make decisions in the area. The Vietnamese government is a stakeholder with high power and high interest.

#### 3.4.4 Local authorities

Local authorities in Vietnam are, in principle, local representatives of the state. Their responsibility is to carry out the plan of the state on a regional scale, because Ben Tre Province is under direct jurisdiction of the Vietnamese government. In the province, the organisation has the same structure as in the Vietnamese government, with a People's Council (local legislature), a People's Committee (local executive), and the same departments as the state (Finance, Transport, Health, and more). The departments correspond to their matching department at the national level (Albrecht et al., 2010).

The province and the Vietnamese government have corresponding goals, which are more specific for the province. For example, the Vietnamese government and the province both want to increase the productivity regarding rice cultivation. The province has knowledge about its region and knows that harvesting rice 2 or 3 times a year will be too intensive in Ben Tre due to the expected rise in salinity levels in the dry season, so it will adjust the goal to fit the local possibilities (Ministry of Agriculture and Rural Development, 2016). The province is also responsible for local public services, including the infrastructure, public transport and water supply. This gives the local authorities high power and also a high interest.

#### 3.4.5 Environmental organisations

A big environmental organisation that is active in Ben Tre Province is World Wildlife Fund (WWF). Currently WWF is investing in sustainable shrimp-rice production. This project hopefully results in a reduction of pollution in the Ham Luong River (WWF, n.d.-a). WWF has the goal to sustain natural life for the benefit of people and wildlife. In the Mekong Delta this results in protecting the natural wildlife and morphology, addressing water risks, climate-smart food production and promoting nature based solutions (WWF, n.d.-b). Environmental organisations have less power than the development organisations, but their interest could be considered a little higher since the impact of any intervention in the river can be significant to them.

#### 3.4.6 Development organisations

A few examples of development organisations are: The World Bank, U.S. Agency for International Development (USAID) and the Asian Development Bank (ADB) (ADB, 2020). The goal of these organisations is to improve the world by addressing various challenges and opportunities, such as poverty, climate change, health, education, governance, and energy. These organisations are also active in the Mekong Delta (USAID, n.d.). They cannot make any legal decisions in Vietnam, but they could stop funding projects if they disagree with certain projects or events. They should be kept informed and satisfied to retain financial contributions.

#### 3.4.7 Maritime sector

Ben Tre Province has three seaports: Binh Thang Port, An Thuy Port and An Nhon Port. They form the connection between the Tien River and the South China Sea. In addition to that, there is one important river port in the Tien River, which is called the Giao Long River Port, with a capacity of 200,000 tonnes per year (Vietnam Briefing, 2020). All these ports are located outside the Ham Luong estuary, visible in Figure 3.6, due to the limited depth of 2.2 metres at certain locations. Some small harbours located in the Ham Luong River are primarily used for small industries. For example, the harbour at the An Hiep Industrial Zone can take ships of 2,000 - 3,000 tonnes (Vietnam Inland Waterways, n.d.), while the Ben Tre cruise port is used for tourists (CruiseMapper, n.d.). There have been plans for more industrial zones, but they have not been realised yet. Furthermore, it is important to note that 40% of the watercrafts are unregistered, which makes it difficult to get a clear picture of the maritime vessels in the Ham Luong River. During the field trip, it became evident that fishing vessels were using the river, as stated in Appendix A. The main concern for the maritime sector is reaching their destination via water. If there would be an obstacle in the Ham Luong River, it would have limited impact on these smaller vessels, as these small ships have the option to navigate through inland waterways. In conclusion, their interest and influence are relatively small.



Figure 3.6: Different ports in Ben Tre Province, showing three sea ports and one river port, all outside the Ham Luong River.

#### 3.4.8 Tourism

In 2018, the Mekong Delta welcomed approximately 1.8 million tourists out of a total of 7.8 million tourists in Vietnam. These visitors typically spend an average of two days exploring the Mekong Delta and spend relatively modest budgets for their trips. While there is a government initiative to boost both the number of tourists and their expenditures in the region, the current tourism figures indicate that tourism might not be the most significant stakeholder at present (Nguyen Hoang Phuong, 2019). Furthermore, it is important to note that small tourist boats could travel via side channels to another estuary. Therefore their power and interest are relatively low.

#### 3.4.9 Overview stakeholders

In Table 3.2 each stakeholder is shown with a summarised description of its interest. In Figure 3.7 each stakeholder is placed on a Power Interest Matrix, based on the information provided above. The most important stakeholders are placed under 'key stakeholders'. It can be seen that the local authorities have more power than the Vietnamese government, on the area of Ben Tre. Due to the size of the Mekong River, many local authorities are actively involved in the decision making. Therefore, it is important that the local authorities actively work together to resolve the challenges in the Mekong Delta. Furthermore, it is important to also involve the agricultural sector in the decision making process.

Stakeholder	Interest
Agricultural and aquaculture sector	Functional (natural) conditions to produce products
Citizens	Safe and future-proof living area
Local authorities	Safe and productive province in the future for its citizens
Vietnamese government	Safe and productive area in the future for Vietnam
Environmental organisations	Protecting ecology in a sustainable way
Development organisations	Improving the overall prosperity of Ben Tre Province
Maritime sector	Accessibility to their destination
Tourism	Accessibility to their destination

Table 3.2: Overview of the stakeholders with their interests



Figure 3.7: Stakeholder Power Interest Matrix

## 4 Scenario selection

This section elaborates on the selection of certain criteria and scenarios that are considered in this project. These are are based on the knowledge gained from the system description in Section 3. Firstly, the criteria that play a role in assessing the best fitting solution are mentioned. These criteria will eventually be used to compare the different closure alternatives. Next, a selection is made on the weather conditions that are considered in the simulations, taking the computational time into account. Moreover, three potential closure locations are determined. Which form the basis of the four different alternative closure scenarios that are considered in this project. Lastly, the relevant IPCC climate scenario is chosen.

#### 4.1 Criteria

In order to evaluate the different alternatives, a criteria set is defined. These criteria are based on the different functions of the estuary that were identified in the system description in Section 3. For each criterion, an explanation is given to justify why it has been selected. The costs of the four alternatives are not considered as criteria, but will be used for the cost-benefit analysis, where the outcome of the four alternatives will be linked to the corresponding costs. In this criteria selection, it has been decided not to include the criteria of flood safety. Instead, this criteria is implemented as a structural design requirement of the flood defense systems.

#### Freshwater supply

The Ham Luong River functions as an important source of fresh water for domestic use. The extent to which salt water reaches into the river significantly impacts the freshwater supply. A closure of the Ham Luong River could change the salt intrusion lengths in the river.

#### Agricultural adaptation

From the system description and the stakeholder analysis, it was found that the agricultural sector plays a large role in the system of interest. Adding to this, the agricultural activities, including aquaculture, are highly dependent on the salinity of water. A possible closure could affect the locations that are suited for specific agricultural activities. It could cause a change in area division for the aqua- versus agriculture. Adaptation of current farmers is in this case inevitable, and should be considered carefully when opting for an alternative.

#### Biodiversity

Changes in the Ham Luong estuary will have impact on the species that define the biodiversity of the estuary. Not only climate change, but also a possible intervention in the estuary can have implications for their living environment. These effects and implications should be taken into account when assessing the various alternatives.

#### Stable riverbanks

Intervening in a morphodynamic system will eventually result in adaptation to a new morphodynamic equilibrium. This can lead to erosion or sedimentation of the riverbanks. The stability of the banks directly near the barriers will not be assessed, as the stability of these banks can be considered as technical requirements of the barrier. This criterion focuses on estuary-wide morphodynamic changes.

#### Navigability

The navigability is important for the transport on water on the Ham Luong River. Any intervention in the river can have significant consequences for its navigability, affecting factors such as the river's flow velocity and depth. The magnitude and impact of these consequences depend on the alternative.

#### 4.2 Weather conditions

As mentioned in Section 3.1, the system of interest is characterised by a high and low discharge season. Different weather conditions can be attributed to these seasons. These weather conditions influence the hydraulic inputs of the system of interest in multiple ways. During the the high discharge season, for instance, large amounts of rainfall are expected, resulting in a larger upstream discharge into the system. These hydraulic boundary conditions can be divided into two categories; upstream discharge and downstream water level. The downstream water level is a combination of RSLR, the tide, waves, and storm surge.

Combining these different hydraulic boundary conditions ultimately leads to a large variety of model combinations that can be simulated. Unfortunately, due to a lack of (computational) time it is not possible to simulate every possible combination. This is also not necessary as the goal of this project is to focus only on the most critical situations for the system. For this reason the most likely combinations of upstream and downstream boundary conditions that threaten the aforementioned assessment criteria the most, are selected and are defined as the "critical situations". In selecting the critical situations, both the present weather conditions and those in 2100 were considered. The different boundary conditions each have a different timescale in which they occur. For instance, storm surge occurs over the course of a few days, whereas larger discharges can be observed over multiple months during the wet season (Thuyloi University, 2020). For this reason a suitable timescales of the critical situations were also selected.

The combinations of hydraulic boundary conditions have led to the three most critical situations, which are listed in Table 4.1. A situation that was omitted is the combination of low discharge and a storm. Based on historical data of typhoons along the coast of Vietnam, it can be assumed that strong typhoons do not occur during dry season (Thuyloi University, 2020). For this reason, this situation is not considered.

Scenario	Upstream	Downstream	Timescale
A: Low discharge season	Low discharge $(2100)$	RSLR (2100) + Tide + Waves	Months
B: High discharge season	High discharge (2100)	RSLR (2100) + Tide + Waves	Months
C: High discharge season	High discharge (2100)	RSLR (2100) + Tide + Waves	Days
+ storm		+ Storm surge	

Table 4.1: Critical situations

#### 4.3 Alternatives

Throughout this report, four alternatives are considered. These four alternatives consist of three alternatives with a barrier and one alternative without a barrier. In all alternatives dyke heightening is included to prevent flooding. Within the three alternatives that include a barrier, the location of the barrier is different. In the preliminary selection of these locations, the aspects that have been taken into account are bathymetry, exposure to hydraulic loads, prevention of salt-water intrusion, and land use. Based on these four aspects, the following three locations have been determined as potential locations. The reasoning of these locations is explained in the section below.



Figure 4.1: Selection of locations of the closure, retrieved from OpenStreetMap

#### Alternative 1

The location of Alternative 1 is the most seaward location of the selection. This makes it the location that is most exposed to hydraulic loads from the sea. The seaward location can significantly attribute to preventing salt intrusion. This also brings its downsides with respect to the possibility of aquacultural activities, which need salt water, in upstream areas. In the present situation, the aquacultural activities are only located along the first 10 km of the estuary and the location of the closure would not hinder these activities. However, a change in land-use is anticipated along the estuary as aquacultural activities expand farther inland due to the rising salinity levels.

#### Alternative 2

The location of Alternative 2 is based on the presence of an island, 14 km inland of the estuary mouth. This makes the individual spans of the barriers smaller. A reference project of a similar closure in the Mekong Delta is mentioned in Section 7.2. On Location 2 the depth of one of the two river branches is significant smaller than on the other locations. Location 2 is further inland than location 1, making it less exposed to hydraulic loads from the sea, but also reducing its prevention of salt intrusion. This location has some more space for aquacultural activities, but has a big impact on the land-use on the island.

#### Alternative 3

The location of Alternative 3 is the most landward location. The width of the river has reduced significantly compared to location 1. This could lead to high benefits in the construction costs. This location corresponds with proposed locations by Tromp (2013) and the Southern Institute of Water Resource Research (SIWRR). This location is the most sheltered location and is expected to experience the least hydraulic loads from sea. It gives room for the longest stretch of coastline for aquacultural activities, but is also expected to allow the highest salinity levels in surrounding areas.

The four alternatives will be further analysed in the upcoming sections. The effect on hydro- and morphodynamic processes in the system will be researched and modelled. In addition, the impact on social and economic activities will be evaluated.

#### 4.4 IPCC Climate change scenario

As the devastating effects of climate change have become more and more evident in the last few decades, it is of huge importance to take climate change scenarios into account. In 2016, the Ministry of Natural Resources and Environment of Vietnam formulated the Vietnamese interpretation of the fifth assessment of the IPCC (AR5). In this assessment, the different climate change projections of IPCC are compared to climate models, predicting different aspects of climate change in Vietnam. The outcomes serve as a guideline to ministries, sectors and provinces which enable the development and implementation of effective plans to respond to climate change and SLR. The assessment's most important recommendation is as follows: "the Representative Concentration Pathway (RCP)8.5 scenarios should be applied to the permanent projects and long-term plans". Therefore, throughout this report, the RCP8.5 scenario will be the leading climate change projection. The corresponding values for relevant effects of climate change, like SLR or increased precipitation, are obtained from the assessment (Tran Thuc et al., 2016).

## 5 Model

This section presents the model which was used to assess the hydro- and morphodynamic response of the system to the partial closure of the Ham Luong estuary. To begin with, the simplifications, which have led to the formation of the conceptual model, are stated. Next, the conceptual model containing the different simulations, which were run in Delft3D, is presented. This is followed by a short intermezzo on Delft3D and its governing equations. Subsequently, the settings and input of Delft3D which were used in the simulations are shared. This is followed by the calibration and verification of the model, after which the sensitivity analysis is performed.

#### 5.1 Model simplifications

As mentioned in Section 4.2, concessions have been made regarding the simulation of different weather conditions as it is not possible to simulate every potential weather condition. Similarly, there are other model simplifications which are required. To begin with, Section 3.2.1 refers to sand mining activities in the estuary and Section 3.2.2 touches upon the presence of hydropower dams in the upstream part of the Mekong River. Both processes influence sediment transport rates in the estuary. However, due to the little amount of information available, it is difficult to estimate these magnitudes. Thus, these processes have been neglected throughout this project. Moreover, as mentioned in Section 3.2.1, there are numerous small branches present within the Ham Luong estuary. It is difficult to include all these small branches in Delft3D, as a high level of detail is required to include these branches in the model. It was also decided not to include water quality in the model. Although Section 3.2.1 and Section 3.3.4 address the implications of contaminated water and soil, a trade-off was made between the importance and the difficulty of modelling and it was chosen to be excluded in the model. Lastly, it has been decided to model in 2D, as a 3D model takes up too much computational time when applied to such a large region. As a consequence, the representation of salinity in this model is influenced.

#### 5.2 Conceptual Model

In order to be able to accurately model the Ham Luong estuary, the area is translated into a conceptual model, which is a simplified version of the Ham Luong estuary and is shown in Figure 5.1. The figure provides an overview of the input necessary to run the model.



Figure 5.1: Conceptual Model; dark blue area indicates the part of the estuary in which the hydro- and morphodynamics are modelled, dark green area indicates the land that is part of the system of interest, orange contours mark the input boundaries of the hydraulic model.

Section 4.2 and Section 4.3 refer to the different weather conditions and the locations of the closures to be considered in this report. The combination of these weather conditions with the different locations leads to the formulation of the different model simulations. A barrier can be either open or closed, depending on operational instructions. An overview of the different model simulations can be found in Table 5.1. The runs which have 2025 in their names are simulations that are equivalent to the present time, so no effects of climate change were included in these simulations. The remainder of the simulations are set in 2100.

Run	Weather condition	Barrier	Barrier type	Timescale	Processes
		location			
A_2025	A: Low discharge season	-	No barrier	Months	Salinity, sediment
A_2100	A: Low discharge season	-	No barrier	Months	Salinity, sediment
A_L1_closed	A: Low discharge season	L1	Closed	Months	Salinity, sediment
A_L1_open	A: Low discharge season	L1	Open	Months	Salinity, sediment
A_L2_closed	A: Low discharge season	L2	Closed	Months	Salinity, sediment
A_L2_open	A: Low discharge season	L2	Open	Months	Salinity, sediment
A_L3_closed	A: Low discharge season	L3	Closed	Months	Salinity, sediment
A_L3_open	A: Low discharge season	L3	Open	Months	Salinity, sediment
$B_{-}2025$	B: High discharge season	-	No barrier	Months	Salinity, sediment
B_2100	B: High discharge season	-	No barrier	Months	Salinity, sediment
$B_L1_closed$	B: High discharge season	L1	Closed	Months	Salinity, sediment
$B_L1_open$	B: High discharge season	L1	Open	Months	Salinity, sediment
B_L2_closed	B: High discharge season	L2	Closed	Months	Salinity, sediment
B_L2_open	B: High discharge season	L2	Open	Months	Salinity, sediment
B_L3_closed	B: High discharge season	L3	Closed	Months	Salinity, sediment
B_L3_open	B: High discharge season	L3	Open	Months	Salinity, sediment
$C_{-}2025$	C: High discharge + storm	-	No barrier	Days	Waves, wind
C_2100	C: High discharge + storm	-	No barrier	Days	Waves, wind
$C_L1_closed$	C: High discharge + storm	L1	Closed	Days	Waves, wind
C_L1_open	C: High discharge + storm	L1	Open	Days	Waves, wind
$C_L2_closed$	C: High discharge + storm	L2	Closed	Days	Waves, wind
C_L2_open	C: High discharge + storm	L2	Open	Days	Waves, wind
C_L3_closed	C: High discharge + storm	L3	Closed	Days	Waves, wind
C_L3_open	C: High discharge + storm	L3	Open	Days	Waves, wind

Table 5.1: Overview of the 24 simulations run with Delft3D

#### 5.3 Delft3D

Delft3D is used to evaluate the conceptual model which is introduced in the previous section. Delft3D is a process-based numerical model that simulates hydrodynamics, sediment transport, water quality, morphodynamic processes and the interactions between these processes in coastal and estuarine environments. Delft3D makes use of the shallow water equations, a simplified form of the Navier-Stokes equations, to simulate water flow. In solving the governing hydraulic, sediment and other equations a time-step and spatial-steps need to be defined. These spatial-steps can be defined in the format of a grid.

Figure 5.2 depicts the schematic that Delft3D adheres to. Delft3D requires the boundary conditions, initial conditions, and bathymetry (bottom depth) as input. Based on these inputs the corresponding hydrodynamics are calculated in terms of waves and flow. These are in turn used to calculate the sediment transport. This influences the bathymetry, which is to be updated and then used again as input and all the aforementioned processes are repeated.



Figure 5.2: Delft3D processes within one time-step (Van Der Vegt et al., 2014)

#### 5.4 Model settings

This subsection refers to the settings which were used as input for the Delft3D model. The grid, simulation time, boundary conditions, initial conditions, thin dams, and wave processes are considered here. For the remainder of the settings it can be assumed that default settings were used.

#### 5.4.1 Grid and bathymetry

From Thuyloi University a dataset on the batymetry of the Ham Luong estuary was received, more information on this can be found in Appendix B.1. From this bathymetry a grid was made. In order to account for the influence of the tide on the upstream boundary of the model, it was decided to extend the grid 130 km beyond the upstream boundary. The calculations with simplified cross-sections are shown in Appendix C. Due to this extension, the grid has become quite large. This influenced the computational time significantly, and it was therefore decided to create a coarser grid. The smallest grid size is approximately 60 metres. As a consequence, the ratio of the grid cells in M-direction (the M-smoothness values) do not align with their recommended values at every grid cell in the model. Due to a lack of time, it was decided to accept that the processes at these locations are possibly underestimated and take this into consideration when interpreting the model results. Also on the seaside, an extended grid was used. This was done to compute waves.

#### 5.4.2 Simulation time

As mentioned before, three different types of weather scenarios are considered in this project; low discharge season, high discharge season, and high discharge in combination with a storm. Each have different simulation times. The low discharge season occurs roughly from January to June, so these simulations were performed over a five month period starting at the first of January, to the first of June. For the high discharge season, also five months were modelled, now starting at the first of June, and ending at the first of November. The storm during the high discharge season was modelled for four days, from the 18th of August to the 22nd of August, as these dates align with the highest tidal water levels. A time step of one minute was used, to comply with the Courant number.

#### 5.4.3 Boundary conditions

As can be seen in Figure 5.1, the model has four boundaries: one upstream and three downstream near the coast (North, East, and South). The upstream boundary is characterised by the discharge. During low discharge season a constant discharge of 256 m<sup>3</sup>/s is used, while a simplified discharge time series is used during the high discharge season. This if further elaborated on in Appendix B.2. The downstream boundaries are characterised by the tide and the waves. Section 5.4.6 provides more information on waves. Using T-tide, the tidal constituents that are governing near the Ham Luong estuary were found (more on this approach can be found in Appendix B.3) and used as input for the East boundary. In order to account for RSLR (Appendix B.4) and storm surge (Appendix B.5) an extra tidal constituent (a\_0) was introduced. For the North and South boundary a Neumann time series boundary condition was used. Moreover, salinity levels were also defined at the boundaries. For the upstream discharge, a value of 0 g/L, salinity is considered (Gugliotta et al., 2017). The salinity at the open coast is based on the default value within Delft3D for marine waters, and is defined as 31 g/L. Finally, the upstream boundary also includes sediment transport rates, more on this can be found in Appendix B.6.

#### 5.4.4 Initial conditions

Initial conditions regarding water level, salinity, and sediment are needed. The initial conditions for water level, salinity and sediment concentrations are based on an initial value map. The processes of salt intrusion and sediment transport in an estuary are slow processes. When starting with one uniform initial value, the spin-up time that is needed to obtain realistic gradual gradients in salinity would be large. In order to avoid this long spin-up time in the different simulations, two separate simulations were run in advance. These two runs created an initial value maps for the low discharge season, as well as one initial value map for the high discharge season. These initial value maps contain water level, salinity and sediment concentration per grid cell and do therefore reduce the spin-up time significantly.

#### 5.4.5 Thin dams

In order to be able to model the effect of the closure of the Ham Luong estuary, thin dams were used which represent a barrier. The barrier was modelled with open gates, and closed. The simulations in which the gates were closed, the barrier was represented by thin dams which span the entire width of the river. The 'open gate' simulations are represented by thin dams which span about 30% of the river width.

#### 5.4.6 Waves

Due to the long computation time when including the wave module in Delft3D a decision was made about modelling with waves. For weather scenario A en B the wave impact is modelled for only a duration of 4 days, without processes like salinity and sediment, as there is no need to simulate wave action for the whole simulation period of 5 months. Scenario C has an duration of 4 days, which makes it possible to model waves for this scenario. The input values for the significant wave height, wave period and wind speed are visible in Appendix B.8 and Appendix B.7. For scenario C a return period of 100 year is chosen with the corresponding wave height.

#### 5.5 Verification

The model has been verified using multiple datasets which were obtained from several Vietnamese institutes which were visited during the field trip. In doing so, the following different processes were considered; water level, discharge, and salinity. As too little data on sediment concentrations was available to verify the model, sedimentation rates could not be verified. This contributes largely to uncertainties in the interpretation of the sediment behaviour, which is elaborated further in the Sensitivity analysis (Section 5.6), and the Discussion (Section 10).

#### 5.5.1 Water level

Three datasets are available with measurements on the water level at different locations in the Ham Luong River. The first two measurement sets were provided by the ICOE and contain the hourly water level at An Thuan and My Hoa over the course of 21 years (2000 to 2021). Bearing in mind that the dataset in An Thuan was also used to generate the tidal constituents in T-Tide, the water levels were compared to the output water levels of Delft3D and are shown in subgraphs A and C in Figure 5.3. For simplicity, only five months in the dry season are visualised. The other dataset which was available was provided by the SIWRP, and contains the hourly water level at Tren Song for 15 days in April 2023. Again this dataset was compared to the water level of the Delft3D simulation. This is shown in subgraph E of Figure 5.3. The simulations in Delft3D were performed over the same time periods, and an upstream discharge was used which is in accordance with the 15% fraction of the discharge in Kratie measured at that time (Mekong River Commission, 2023a).

At first glance, it looks as if the Delft3D water level is out of phase with the datasets of the institutes. When shifting the water level of Delft3D 3 days and 22 hours forward, the datasets seem to align better visually. Especially the peaks of the water levels are more in accordance. This phase shift can be explained by multiple factors; the spin-up time of the Delft3D model plays a role, as is the fact that not all tidal constituents (only the eleven largest) are taken into account in the Delft3D model. Additionally, the choice of initial conditions of the water level may also influence the shift. When bearing in mind the approximately four day shift of the Delft3D water level, the output of the Delft3D model does quite well resemble the water level of the datasets in terms of magnitude and tidal phase. An attempt was made to identify the cause of the tidal shift. By analysing the water level output from the Delft3D model, using the same tidal analysis tool (T-tide) as done in Appendix B.3, the tidal constituents of the modelled water level can be compared to the initial constituents of the input. By doing so, indeed significant difference in phase can be seen, while the amplitudes show a good fit. However, the specific cause of the phase shift was not identified. In view of time, no elaborate further quantitative methods were used to asses the model's performance.



**Figure 5.3:** Comparision of water level from datasets to output of Delft3D. A: Water level in An Thuan, dataset provided by ICOE. B: Water level in An Thuan, same datasets as in subfigure A except the Delft3D data is shifted 3 days 22 hours forward. C: Water level in My Hoa, dataset provided by ICOE. D: Water level in My Hoa, same datasets as in figure C except the Delft3D data is shifted 3 days 22 hours forward. E: Water level in Tren Song, dataset provided by SIWRP. F: Water level in Treng Song, same datasets as in figure E except the Delft3D data is shifted 3 days 22 hours forward. G: Locations of the measurement stations in the Ham Luong estuary

#### 5.5.2 Discharge

Besides water level, the dataset at Tren Song provided by the SIWRP also contains the discharge at that location for the same dates. Again these were compared to the outputs of Delft3D. Delft3D however, does not provide the discharge as an output. Only the water level and depth-averaged velocity are included in the output. In order to compute the discharge some assumptions were made. The depth-averaged velocity at the measurement station was assumed to be constant over the width of the river. Additionally, a time-dependent wet cross-section was calculated to resemble the tidal discharge. To do so, the wet cross section per timestep was computed by adding or subtracting the width of the water level at MSL multiplied with the water level to the cross section of the river below MSL. This wet cross section was in turn multiplied with the depth-averaged velocity. These discharges were compared, and are displayed in Figure 5.4. It can be observed from the figure that, when shifting the Delft3D outputs approximately four days, the discharge measured at the measuring station at Tren Song is in accordance with the discharge computed with Delft3D.



Figure 5.4: Comparison discharge from dataset to output of Delft3D. A: Discharge in Tren Song, dataset provided by SIWRP. B: Discharge in Tren Song, same datasets as in figure A except the Delft3D data is shifted 3 days 22 hours forward.

#### 5.5.3 Salinity

Not much data is available to verify the modelling of salt intrusion. The most suited data was provided by ICOE. This data includes salinity values near the river mouth at An Thuan in 2015, 2016 and 2019. The data does not contain yearly measurements, but only series of 5-month measurements with regularly missing values. In order to verify the process of salinity in the Delft3D model, the same period of the year of the available data was modelled in Delft3D. The measurements and modelled values are shown in Figure 5.5. It can be seen that the model output is not matching the fluctuation of the measurements very well. Besides the daily fluctuations in salinity that might be the result of the daily varying discharge, a gradual decrease in salinity values can be seen in the measurements. The decrease in salinity around May and June is the result of the increasing discharge of the river towards the end of the low discharge season, which leads to lower salinity. As the verification runs did assume a constant discharge, this trend is not visible in the Delft3D output.

Furthermore, the modelled salinity shows significant lower minimum salinity values than the measurements. There can be several reasons for this difference. One of the possible causes of this mismatch is the fact that a 2D-model is used with a homogeneous density. In the model, the density of fresh water is assumed for both river discharge and seawater. In reality, the density difference between salt water and fresh water is causing a density-driven flow of seawater into the estuary, resulting in a salt wedge. This density-driven flow is not represented in the model, which could partially explain the lower salinity values in the model compared to measurements.

It becomes clear that the variable river discharge is strongly affecting the salinity in the river mouth. For the simulations, a simplified time series during high discharge season, or a fixed value during low discharge season was used. Reducing the discharge value increases the salt intrusion distance, but not as much as measurements have shown. The modelled salt intrusion is mainly driven by the tide and does not show significant variations, as for example a strongly fluctuating discharge would do.

Some more reasons for the underestimation of the salt intrusion might be found in other details of the model. For the viscosity and temperature, uniform default values have been used in the model. The viscosity values are connected to the grid concerning stability, but also to morphodynamic and hydrodynamic processes. An increase in salinity results in an increase of viscosity, which also depends on temperature (Sánchez-Juny et al., 2019). In reality these values are varying over along the estuary, which impacts connected salinity values. Additionally, the grid itself could also be a cause of the poor representation of salinity values. Especially at the interface between river mouth and sea, the used grid has too large grid cells. As this part of the system affects salt intrusion strongly, any inaccuracy related to the grid can cause discrepancies.

These discrepancies indicate that the model results are not quite as realistic as initially desired. For this reason, it should be carefully noted that outputs concerning the length of the salt intrusion, and the concentrations at these locations that follow from the model cannot be assumed to be true. Keeping in mind that salinity is underestimated in the model, it is however still possible to compare the salinity outputs of the different simulations amongst each other. As all simulations are subject to the same model settings, and thus display the same inaccuracies.



Figure 5.5: Verification salinity. Comparison between measured salinity values of three years and modelled salinity values.

#### 5.6 Sensitivity analysis

To get a better understanding of the model and to quantify uncertainty, a sensitivity analysis is performed. By changing an input parameter, the range of outcomes can say something about the effect of that parameter on the results. This provides insight on the uncertainty range of the output considering assumptions and uncertainties in the input. For this model, the sensitivity analysis is performed for Manning's roughness coefficient and the median sediment diameter,  $D_{50}$ , of the sand. Due to lack of computational time, the simulation time is only one month, which can be enough to compare the initial differences, but will not give detailed information about further outcomes.

Manning's coefficient represents the roughness of the river bed, and is used to estimate the flow rate in a channel (Battjes & Labeur, 2017). In the model, the chosen value of 0.02 is taken to be uniform over the whole estuary, while there can be spatial variation due to vegetation or other obstructions. To analyse the sensitivity of the model to this coefficient, simulations were performed with a uniform value twice as high (0.04) and a value twice as low (0.01) as the assumed value, during one month with low discharge, without any intervention.

From these simulations it can be concluded that the change in Manning's coefficient has a significant effect on salinity values. Differences in salinity are in the order of a few ppt at the estuary mouth, where the lower Manning's coefficient results in higher salinity values. About 10 km upstream, the salinity values for a higher Manning's coefficient reach zero over time, while the salinity values for a Manning's coefficient of 0.01 reach values 5 times higher than for the default value of 0.02. The reduction of salinity was to be expected with a higher roughness, because the effect of tide has a smaller reach in a less smooth estuary. For a Manning's coefficient of 0.01, the tidal influence and the related salt intrusion reaches much further inward than for the default value. The effect on depth averaged velocity is in the order of 0.1 m/s over the whole estuary. The lower the Manning's coefficient, the higher the depth averaged velocity. This can be related to morphodynamic processes such as sediment transport and erosion or sedimentation, which must be taken into account. So, a higher Manning's coefficient results in lower salinity levels and lower depth averaged velocities in the estuary and should be chosen carefully.

For the median sediment diameter  $(D_{50})$  the default value of 200 µm is used throughout the simulations. To check the sensitivity of the model on the choice of this median sediment diameter, a value of 150 µm and a value of 300 µm were compared to the original value. The same default sediment transport formulas were used for each simulation. These simulations were done for one month during high discharge, which includes mud transport as well as sand.

The effect on the sediment sand amount is as expected: a lower  $D_{50}$  of the sand results in a higher amount of sand to be transported. This affects the cumulative erosion/sediment, which has different results over the length of the estuary, in the order of 0.0001 m to 0.01 m. In the river mouth, for example, a lower  $D_{50}$  results in more erosion, while the opposite occurs near the most downstream located island. These local changes result in small changes of the bed level, but there are no significant differences visible for water depth and water level. Same can be said about the salinity values, which shows that the choice of the median sediment diameter has little influence on the outcome of the model. These comparisons are done for a simulation that only covers one month, while morphodynamic changes can take much longer and reaching an equilibrium can take multiple decades, which is not taken into account in this sensitivity analysis. In the case of lack of data and information, choosing the default value, based on other research and data, seems to be the best option.

Small changes in parameters can often result in significant changes of the results. Those parameters have to be carefully considered and compared to other researches and experiments. In this case, it is more important to research and analyse the value of Manning's coefficient than the precise  $D_{50}$  of the sand. Knowledge about the uncertainties of the input is imperative for the interpretation of the results and its ranges of uncertainties.

### 6 Results

This section shows the findings of the Delft3D model presented in the previous section. Additionally, the results of an assisting sedimentation tool are shown. These results are to be used in the assessment of the different alternatives.

#### 6.1 Model results

First the results on the water levels in Delft3D are presented, which is followed by the evaluation on the wave height and salinity. Finally, the sediment behaviour in the model is commented on. Before going into details about the model results, it must be noted that due to the scope of this project the processes upstream of the barrier in case of closure could not be analysed. This is due to the fact that no upstream bifurcations are included in this model, which has the consequence that the water piles up behind the barrier. This results in unrealistic water levels, and salinity and sediment processes. For this reason, the upstream parts of the simulations in which the barrier is closed were not interpreted. More comments on the scope can be found in Section 10. This applies to the results upstream of the barriers of the following runs mentioned in Table 5.1; "A.L1\_closed", "A.L2\_closed", "A.L3\_closed", "B.L1\_closed", "B.L2\_closed", "C.L1\_closed", "C.L1\_closed".

#### 6.1.1 Water levels

The water levels were investigated at multiple locations. This section only focuses on the water levels at two locations, as a clear distinction can be made between the water levels upstream, further away from the locations of the barriers and downstream near the mouth of the river. The two locations which were considered are shown in Figure 6.1 and Figure 6.2.



**Figure 6.1:** Water levels at observation point 7. A: The daily maximum water level at the observation point in the dry season. B: The daily maximum water level at the observation point in the wet season. C: The water levels during a storm at the observation point. D: Location of Observation Point 7.
The water levels upstream were analysed at the location of Observation Point 7 (OP7). The results are shown in Figure 6.1. Only the water levels of two runs are shown; the water level with no intervention in 2025 and 2100. The upstream water levels of the alternative barriers with open gates produced similar results to the water level at 2100 without interventions. The water levels in 2100 are significantly higher than the water level in 2025, which can be attributed to a larger discharge and RSLR. During the high and low discharge season, the water levels frequently rise above the elevation height of the dykes. Furthermore, it should be noted that a storm in the wet season will produce water levels above the dyke elevation, for both the 2025 and 2100 scenarios.

Figure 6.2 displays the downstream water level at the location of Observation Point 2 (OP2). Again no large differences can be observed between the water levels for runs in which the gates of the barriers are opened compared to the run with no intervention in 2100, and for this same reason these water levels were excluded from the graph. There is however a small difference in water level which can be observed when looking at the closed scenarios. Especially in the wet season, the water levels are lower when the barriers are closed. This can be attributed to the fact that the large discharge corresponding to the wet season is blocked by the barrier, thus reducing the water levels downstream. No water level difference can be observed during a storm in the wet season between the different closure cases. This can be attributed to the fact that the storm surge and waves are dominant over the discharge differences in affecting the water level.



**Figure 6.2:** Water levels at Observation Point 2. A: The daily maximum water level at the observation point in the dry season. B: The daily maximum water level at the observation point in the wet season. C: The water levels during a storm at the observation point. D: The daily maximum water level at the observation point in the dry season, zoomed in on a period end of April. D: The daily maximum water level at the observation point in the wet season, zoomed in on a period end of September. F: Location of Observation Point 7.

#### 6.1.2 Wave height

In Figure 6.3 the significant wave height is visible during storm conditions in 2100 in the high discharge season. The maximum significant wave height in the Ham Luong river during storm conditions is 3 metre. It can be seen that the waves that originate from the sea do not propagate far into the estuary. The reason for this is that the storm is approaching from the SW direction. The barriers at Alternative 1, 2 and 3 do not experience the influence of waves that originate from the sea. The Ham Luong estuary is a very wide estuary, which results in wind waves originating from inside the estuary. These waves are up to 1.5 metre, visible in Figure 6.3. For the normal weather conditions, the maximum significant wave height during the low discharge season is 1.1 metre and during the high discharge season is 0.5 metre. These waves originate from the sea and are only present near the river mouth. The difference in significant wave height between the high and low discharge season results from the different wind climates during the year. The significant wave height from the waves that are formed in the river during the both seasons are smaller 0.5 metre.



Figure 6.3: Significant wave height in the Ham Luong estuary during storm conditions in 2100

#### 6.1.3 Salinity

The maximum salt intrusion in the Ham Luong estuary is visible in Figure 6.4. This result is obtained from run A\_2100, as during low discharge season the salt intrusion is at is maximum. As stated in Section 5.5.3, the salinity levels in this model are underestimated, which means that it is expected that the salt intrusion will reach further into the estuary than is visible in Figure 6.4. When a barrier is closed for 2 months in 2100, the downstream side of the barrier behaves like a basin. This basin is slowly filled with salt water as visible in Figure 6.5. The 'basin' of the barrier at location 1 is closer to the sea and thus more saline than with a barrier at location 2 or 3 after two months of closure. However these two months should be interpreted carefully, as salinity is underestimated in the model, it is likely that this filling of the basin will occur faster than is shown in Figure 6.5. When looking at the salinity levels of A\_2100, in Figure 6.5, it looks like the salinity at the location of the barrier in Alternative 3 is already 0 ppt. However, this is a result of the underestimation of salinity in the model.



Figure 6.4: Maximum salt intrusion in 2100 during low discharge season in the Ham Luong River, with the 4pp and 0 ppt boundaries in white



Figure 6.5: Salinity levels on the sea side at the barier along the river after 2 months closure in the low discharge season in 2100

#### 6.1.4 Sediment

Any intervention in a system like the Ham Luong estuary will lead to changes in morphology until a new equilibrium is reached. In the upcoming analysis on sedimentation data, the year of 2025 is seen as the base year. The sedimentation will be described in terms of cumulative erosion/sedimentation rates in metres. Not every simulation from Table 5.1 will be discussed in these results. In terms of cumulative sedimentation, the simulations that include high river discharge lead to the most notable results. Therefore, only simulations with high river discharge are included in these results.

Before interpreting the results regarding sedimentation and erosion, it should be noted that the modelling of the riverbanks is not expected to match reality properly. The used bathymetry data, received from Thuyloi University, does not include water depths less than 2 metre, which results in unrealistic modelling of the riverbanks. However, it is assumed that erosion/sedimentation patterns that occur at the most outer grid cells can be used to interpret the rates at the riverbanks.

Three main drivers of sedimentation/erosion are expected in these simulations. RSLR and increased river discharge will change the morphodynamic equilibrium from 2025 into a new equilibrium in 2100. In Figure 6.6,

the differences in sedimentation/erosion between the base year in 2025 and the simulation of 2100, with no barrier, is shown. This shows the effect of SLR and increased river discharge on the bed level. Only in the more upstream reach of the river, significant differences in bed level change can be seen. These differences are in the magnitude of maximum 1 metre for both erosion and sedimentation. In these upstream reaches, exactly the same differences can be seen, when looking at the three simulations with barriers in 2100.



Figure 6.6: Difference in cumulative sedimentation/erosion between base year 2025 and 2100 without a barrier.

The third driver that is expected to induce changes in erosion and sedimentation rates, when comparing 2100 with 2025, is the implementation of a barrier. In Figure 6.7, the differences in sedimentation between the base year and the three runs with barriers on different locations in 2100 are shown. The results are as expected: near the barriers, high sedimentation and erosion rates can be seen. Erosion is mainly taking place just upstream of the barrier, as a result of the contraction of flow through the barriers. Sedimentation rates occur just downstream of the barriers when the flow rates are decreasing again. However, in Figure 6.7 only the local effects of the barrier are shown. When looking at sedimentation/erosion effects more upstream or downstream of the barrier locations, no differences between the four alternatives are observed.



Figure 6.7: Differences in cumulative sedimentation/erosion between base year 2025 and three situations with barriers in 2100.

### 6.2 Sedimentation tool

As mentioned in Section 5.5, insufficient data is available to verify the sedimentation processes that are modelled in Delft3D. Therefore, an additional tool is used. This tool is based on the sedimentation formulas of Van Rijn (Van Rijn, n.d.). This tool is used to estimate the sedimentation rates and concentration in the case of closed barriers. As mentioned in Section 6, it is not possible to make well-based estimations about processes that are related to the upstream discharge while barriers are closed. Therefore, in this section the focus is only on sedimentation rates on the seaside of the barrier, while it is closed.

The used tool is a quick assessment on sedimentation rates and concentration of fine sediments in a basin that is influenced by a tidal regime via one entrance. In the case of the Ham Luong estuary, the part of the estuary between the closed barrier and the river mouth is considered as the basin. In Appendix D the input of the tool for the three different barrier locations is given in Table D.1. In Figure 6.8 the development of the concentration of mud during four tidal cycles is shown. A clear difference between the different locations can be seen.



Figure 6.8: Mud concentration in case of closed barriers.

Besides the concentration, the tool also returns the total sedimentation layer. This sedimentation rate is initially calculated per tidal cycle. The extreme situation during dry season, where the barrier is closed for 5 months in order to prevent salt intrusion and store fresh water, is considered as a critical situation. This period would lead to a total sedimentation layer as stated in Table 6.1, depending on the location of the barrier.

Table 6.1: Total	sedimentation	layer	after $5$	months
------------------	---------------	-------	-----------	--------

Location	L1	L2	L3
Total sedimentation layer (m)	0.160	0.133	0.117

# 7 Design of flood protection system

In this section, the design of the flood protection system will be described. The flood protection system consists of dykes and a storm surge barrier. A storm surge barrier is a structure with a partial movable barrier that can be closed temporarily (Mooyaart et al., 2014). The function of the dykes is to withstand the water levels in the estuary to ensure flood safety, which is considered a strict requirement for all four alternatives. The storm surge barrier has the main function of closure in case of low discharge from the Ham Luong River or during storm surge, which can occur in combination with high discharge. Its goal is to prevent saltwater intrusion and flooding from the sea, while retaining fresh water in the estuary. The storm surge barriers' secondary function is to open during high discharges to let the river water flow into the ocean.

The structure of this section is as follows. First, in Section 7.1 the design of the dyke systems for the four different alternatives is shown. This design is on a conceptual level and focuses on the differences between the four alternatives. In the sections Section 7.2 to Section 7.8, the design of the storm surge barrier is described. This design is on a more detailed level than the dyke design but is also a preliminary design.

## 7.1 Dykes

In this subsection, a conceptual design of the dyke system is given. These dykes comply to the requirement of flood safety. Here, the design of the required dyke heights is shown, after which an overview of the dyke system is given.

The needed dyke system depends on the different alternatives. The length and needed height of the dyke system differs per alternative. In this design, a distinction between two dyke types is made: sea dykes and river dykes. The sea dykes are on the side of the structure that faces the sea and the river dykes facing the river upstream. This means that every alternative has a different dyke system design. Note that this design of the dyke system stops at the mouth of the river, while in reality the dykes continue at the coastline, but this is outside the scope of this project. First, the height of the sea dykes will be calculated, after which the height of the river dykes is given.

The sea dykes are the dykes that are exposed to storm surge levels and wave action. The current sea dykes have a height of 3.0 m as stated in Section 3.3.2. Due to the expected RSLR and other extreme weather conditions the dykes need to be elevated to protect the hinterland. To determine a future-proof crest height of the year 2100, the general approach for wave overtopping discharges given in the "Manual on wave overtopping of sea defences and related structures" has been used (Van der Meer et al., 2018). For safety reasons, the angle of the incoming wave is chosen to be 60°. Furthermore, a vertical wall has been placed on top of the dyke, with a height of 1.0 m. The choice for a wall on top of the dyke gives the opportunity for smaller width of the dyke, which creates less hinder for the people living around the dyke. The material of the dyke embankment revetment is grass as noted in Section 3.3.2. Due to the absence of a berm at the moment and the presence of houses close to the water, no berm will be constructed because this would also increase the width of the dyke. The dyke calculations, which can be found in Appendix E.1, result in a final crest-free board of 6.8 m, excluding the vertical wall with a height of 1.0 m. A drawing of the cross-section of the dyke can be seen in Figure 7.1. A width of the dyke crest of 4.5 m is chosen for the possibility of building a road on the dyke.



Figure 7.1: Cross-section of the sea dyke design, including dimensions of the height and width in m.

It has to be noted that this dyke height is designed on wave action, storm surge, and RSLR. This is relevant for the first stretch of kilometres land inward. As stated in Section 6.1.2 at a certain distance from the coast, the wave action at the dyke has reduced. This will lead to a reduction in the needed crest height.

The river dykes are the dykes that are exposed to high water levels in the river, which can be the result of tide or

high discharge. However, the river dykes are not exposed to storm surges, as the river dykes are located in the hinterland of the storm surge barrier. Furthermore, no significant wave action on the river dykes is expected, as stated in Section 6.1.2. The height of the river dykes is based on the maximum expected water levels in the Ham Luong estuary during high discharge season, which includes RSLR. According to the Delft3D model, the water levels will rise to more than 3.5 m, see Section 6.1.1, and thus a needed dyke height of 4.0 m can be estimated. These dyke heights are not specifically designed for the case of a closed barrier during high discharge season, as the upstream effects of this scenario are not included in the Delft3D model. It is a model approximation, therefore a large safety margin of 0.5 m is chosen for the dyke height. The positions of the river dykes and sea dykes per alternative can be seen below in Figure 7.2. The existing dyke system in 2025 in Figure 7.2A is an estimation, based on information that was provided by ICOE. The dyke height for Alternative 4 (Figure 7.2E), would only be required for the first kilometres, after which a lower dyke would suffice.



Figure 7.2: Dyke system: The existing dyke system for the current situation (A) and the proposed dyke systems for the four alternatives (B to E). The height of the sea dykes includes the 1.0 m height of the wall on top of the dyke

## 7.2 Reference projects

There are two projects in the Mekong Delta that are used as reference for the Ham Luong River barrier. The Ba Lai River sluice in Ben Tre Province and the Cai Lon - Cai Be sluice gates located in the province of Kien Gian, near the Gulf of Thailand. This subsection will elaborate on these structures, starting with the Cai Lon - Cai Be sluice gates.

### 7.2.1 Cai Lon - Cai Be sluice gates

In the province of Kien Giang, in the western part of the Mekong Delta, a large project for water control is carried out. The Cai Lon - Cai Be sluice gates are used to prevent salt intrusion in the agricultural areas in the provinces of Kien Giang, Hau Giang, Ca Mau, and Bac Lieu. Additional flood protection measures were taken, consisting of three barriers: The Cai Lon, Cai Be, and Xeo Ro sluices. There is a dyke connecting the sluices with the national highway on top of it. Furthermore, sluice gates along the An Bien and sea embankments are part of this 3.3 trillion VND ( $\in$ 136.8 million) project (Tuoi Tre News, 2022).

This barrier is the largest in the area and therefore a very interesting reference project, close to the Ham Luong River. The Cai Lon sluice in particular is very large, spanning 470 m and consisting of 11 gates, each 40 m in width. The gates are vertical lift gates and because of the large span, there is a large steel truss supporting the steel plates that form the barrier. The Cai Lon River barrier can be seen in Figure 7.3. It has two ship locks with a width of 15 m and a length of 100 m. Furthermore, the structure is also designed to have a bridge on top of it (GIZ, 2019).



Figure 7.3: Cai Lon River barrier (Tuoi Tre News, 2022)

While the purpose of constructing the sluice was to control the inflow of salt water, it is important to note that salt water was already used for agriculture prior to the construction of the sluice, due to the cost-effectiveness. Consequently, the river gates have to be open during high tide to allow salt intrusion, leading to high energy costs because the gates have to be open all the time. Furthermore, fresh water still needs to be pumped into the area from the Hau River, which is associated with relatively high costs. This information was obtained from a meeting with experts from Royal HaskoningDHV, see Appendix F.2.

#### 7.2.2 Ba Lai River sluice

In the Ben Tre Province, just outside the chosen area of interest, there is a gated barrier in the Ba Lai River. It was built in 2002 and forms an almost permanently closed barrier. The barrier is placed at a position in the river where there is an island. Only on one side of the island, the barrier has gates, while on the other side, a closure dam has been constructed, which can be seen in Figure 7.4a. The gated section of the barrier can be seen on the right side of the island and the closure dam is on the left of the island. The entire dam has a total length of 544 m and has 10 gates, each 10 m in length (VNCOLD, 2014). There are two gate types in this structure. A barge gate, which is a gate that pivots around a vertical axis to close, and a vertical lift gate operated by a crane (Voorendt, 2022). In Figure 7.4c and Figure 7.4d the gate system and lifting crane for the gates are visualised. The gated section is only 100 m in length. For the Ham Luong River, the dam will have larger dimensions because the width of the Ba Lai River is much smaller than the Ham Luong River. However, it is a good project to refer to concerning the effects it has had over the past two decades on the area.



(a) Ba Lai River barrier top vieuw Source: Google Maps



(c) Ba lai Gate system



(b) Gated section of the Ba Lai barrier



(d) Ba Lai crane

Figure 7.4: Ba Lai River barrier

The Ba Lai River barrier has reduced salt intrusion in the area, which was the main concern at that time. However, it had some negative effects on people living nearby. Farmers working with salt water had to change their business to comply with the freshwater system. Fishermen faced a similar challenge because of the depletion of natural saltwater resources. Furthermore, the dam reduced the river flow circulation leading to the accumulation of different kinds of pollution. The largest factor leading to accumulated pollution is aquacultural discharge. Also, landslides at river banks have become an issue in the area since the building of the structure (Xuan Quang et al., 2017). These are possible side effects that must be taken into account for a barrier in the Ham Luong River. Currently, due to these effects on the area the Ba Lai River barrier is not considered to be a successful project.

### 7.3 Feasibility study

In this subsection, a feasibility study on the different types of gated structures is performed. There are different assessment criteria on which the structure feasibility can be assessed and there is only a limited number of storm surge barriers built worldwide. All of these structures show a great variety in hydraulic gate types, because of differences in function and environmental conditions (Mooyaart et al., 2014). This shows that there is not a clear guideline on how to choose a certain storm surge barrier type. Thus, an MCA is performed. The initial costs are an important factor, however, these cannot be quantified as the driving forces as costs are not well known. Therefore, they are left out of the analysis (Mooyaart et al., 2014). The structural requirements of the structure are shown in the list below (Molenaar & Voorendt, 2022):

- Constructability
- Stability
- Strength
- Maintainability
- Adaptability

To see how well the different alternative design options score on the structural requirements, evaluation criteria are formed. The evaluation criteria are derived from the system description. With the evaluation criteria, the

different alternative options for design can be compared and the best option can be chosen. The list of design criteria is as follows (Molenaar & Voorendt, 2022):

- Ease of construction; this is about the simplicity of design and how easy it is to build such a design in the Ham Luong River. The depth of the river is for instance important for the type of structure that can be built. The construction time is also included in this criterion.
- Ease of adaptability; this is the ease at which the structure can be adapted to new conditions. For instance, the ease to improve the height of the barrier in the case of a higher RSLR than expected.
- Sustainability; this is about the influence of the structure on the ecosystems. The amount of CO<sub>2</sub> emission is one indicator, but the amount of materials used and how much of the local ecosystem must make way for the structure are also sustainability indicators.
- Ease of maintenance; this is a combination of future predicted maintenance costs, simplicity, and the strength of the structure.
- Durability; this says something about the strength and the robustness of the structure. The structure must be able to withstand the large forces that will be exerted on it by wind, water, and soil.
- Aesthetics; this is a difficult criterion because it considers the beauty of the structure, which is a personal consideration. It is assessed by how a structure fits in its surroundings and how impressive the structure is to people in general.
- Navigation possibility; this is the possibility of smaller and bigger boats to pass the barrier, in case of an open barrier. For some structure types, it is easy for all boat types to pass the barrier and for some structure types, it is not. For some structures, a sluice has to be included in the design to let boats pass.

For the technical design of the barrier, there are various options to consider. Several types of water control structures can be used when building a barrier (partially) with gates in it. A couple of these structures can be seen in Figure 7.5 and are listed below and elaborated with the criteria (Voorendt, 2022):

- Vertical lift gates; these are gates that can be translated vertically into place from a higher position. The operating mechanism can consist of hydraulic jacks or a system of cables, pulleys, and counterweights. When closed, the gate rests on a sill beam. The vertical lift gates score very high on ease of construction because the design is easy and the execution is easier than for most other structures. Its design is very similar to that of the Cai Lon sluice mentioned in Section 7.2.1. Furthermore, it is a structure that can be maintained very well because the doors can be lifted out of the water. It can be adjusted in future works. Aesthetically it is not a very beautiful structure that fits in the environment. It is a robust structure so it scores good on durability. It is not very sustainable because it can block a lot of sediment and significantly change the river morphology. Also, the navigation of ships will be hindered. Lastly, the vertical lift gates move vertically which requires a large amount of power, although the use of counterweights could reduce this significantly (Molenaar W.F., 2011).
- Drop gates; these type of gates are very similar to vertical lift gates. They are designed to be translated into position from the bottom of the river bed. Because the gates are often submerged it makes ease of maintenance much more difficult. The construction costs are also relatively high and it costs a large amount of power to lift the gates. On the other hand, the drop gates are better for navigation possibilities because ships can easily pass over the gates when closed.
- Flat gates; these lay flat on the ground and can be rotated to a vertical position when water on one side of the structure has to be retained. Ships face no hindrance to passing this gate when the gate is lying flat on the ground. However, this type of barrier gate performs not very well at ease of maintenance since the gates are submerged. This makes the construction also a bit more difficult than for the vertical lift gates. They cannot be lifted out of the water easily. Furthermore, these gates are more difficult to adapt in the future.
- Visor dams; these are curved barriers that are rotated into place vertically. The curvature of the barrier is in the horizontal plane. Visor dams are very impressive large structures that score high on their aesthetics. They also have a lower impact on the river morphology and thus score high on sustainability. Ships can easily pass this structure. Visor dams are not easy to design, construct, and maintain because of their complexity. Since they are large steel structures they need more maintenance and are not the most durable structures.

- Radial gates; these are gates with a radial arm that vertically rotates the gate into place. The curvature of the gate is in the same plane as the rotation. Radial gates perform well on ease of maintenance because the gates can be lifted out of the water. However, they are less easy to construct than for instance vertical lift gates because they are more complex. Also, the navigation possibility is low. A sluice would have to be built to let ships pass. Their impact on sustainability is also significant due to the change in river morphology.
- Sector gates; these are similar to radial gates because they also have a radial arm. However, these gates are rotated into places horizontally such as the Maeslantkering in The Netherlands. They score very high on sustainability because they have little influence on the river when the barrier is open. Their design and construction are complex, so they score low on these criteria. Furthermore, it is not easy to adapt this structure in the future. Aesthetically it is very good because of its futuristic looks. Lastly, the horizontal movement requires less power than vertical movement, the structure is therefore also sustainable.
- Inflatable dams; these are barriers that are placed on the bottom of a river and can inflate when water must be retained (Voorendt, 2022). The inflatable dam scores high on sustainability and aesthetics. It blends in with the environment very well when the barrier is open. Also, it lets ships pass very easily. However, the design on this large scale is very difficult and it is a structure that is not easy to maintain because it lays completely in the water. Furthermore, it is almost impossible to adapt it in the future. Although it can be a very large structure, it is not a very stable one because it is not a hard structure. Also, there has been little research on what the effect is of dynamic loading on the inflatable dam (Kuijper H.K.T., 1998).

Some of the options above are left out of consideration because they are, in general, not suitable for the function the barrier has to fulfil in this case. Mitre gates swing out from two side walls to form a barrier with an angle pointing towards the upper water level and are disregarded because they are usually relatively small and are thus not considered to be storm surge barriers. They are primarily used in locks for boats. Sliding gates cannot be closed in flowing water and are therefore not suitable (Kuijper H.K.T., 1998). Shutter weirs are crest walls built in a river with on-top shutters to regulate water flow. They are usually not used as flood protection but as water level control mechanisms and therefore not suitable for this design.





(c) Flat gate Source: (NPR, 2009), photo: Graziano Arici



(e) Radial gates Source: (USACE, n.d.)



(b) Principal of a drop gate Source: (Kuijper et al., 1998)



(d) Visor dam Source: (iStock, 2013), photo: yellowpaul



(f) Sector gates Source: (Watersnoodmuseum, n.d.-b)



(g) Inflatable dam Source: (ZJA, 2023)

Figure 7.5: River barrier types

To assess the different kinds of water control structures on the aforementioned criteria, an MCA is performed. The rating is done on a scale of - - to +++, with - - being a low performance on the specific criterion and +++ a high performance. A value of 0 is also possible if there is no significant upside or downside for the alternative on a criterion. There are therefore 7 possible values. In this MCA no 0 value was given. The ratings are based on the description of the barriers and their performance on the different assessment criteria. They are not compared to a basic situation but relative to each other. The results can be seen in Table 7.1.

Gate type	Vertical lift gates	Drop gates	Flat gates	Visor dams	Radial gates	Sector gates	Inflatable dams
Ease of construction	+++	++	++		++		
Durability	+++	+++	++	++	++	+	
Ease of maintenance	+++	-	++		+++	+++	
Sustainability				-		+++	++
Ease of adaptability	+++	++					
Aesthetics			+++	++	-	++	++
Navigation possibility	-	++	+++	++		+++	+++

Table 7.1: Criteria scores for water control structures

Each of the criteria is also given a certain weight depending on their importance, as visible in Table 7.2. If a criterion is more important than another one, the most important criterion is rewarded a 1 and the less important criterion is rewarded a 0 value. When criteria are equally important they both receive a value of 1. Durability and ease of construction are the main drivers that contribute to the start of the construction so these are considered the most important criteria. They both receive a value of 1 relative to each other. Ease of maintenance and sustainability are quite important for the environment and the stakeholders so these criteria are slightly less important. Furthermore, the ease of adaptability is a little bit less important than the aforementioned criteria. The structure is built for a lifetime of 100 years so the adaptability is partially included in the design. The least important criterion is aesthetics as this is not of big importance for the stakeholders. Also, navigation possibility is regarded as least important criterion which corresponds to Section 3. The results can be seen in Table 7.2.

Table 7.2: Weight for each criterion

Gate type	а	b	с	d	e	f	g	Summed total	Weight
Ease of construction (a)	х	1	1	1	1	1	1	6	25
Durability (b)	1	х	1	1	1	1	1	6	25
Ease of maintenance (c)	0	0	x	1	1	1	1	4	17
Sustainability (d)	0	0	1	x	1	1	1	4	17
Ease of adaptability (e)	0	0	0	0	x	1	1	2	8
Aesthetics (f)	0	0	0	0	0	x	1	1	4
Navigation possibility (g)	0	0	0	0	0	1	х	1	4

Finally, combining the weights and the scores of each structure type results in the final score for each structure. The scores of the criteria are converted to a value of 1-7 depending on how positive or negative the score in Table 7.1 is. The results can be seen in Table 7.3. According to this MCA, the vertical lift gates are the best option for this project. It is closely followed by the flat gates. Other types of gates are considered to be too complex to construct or have other unwanted features.

Criterion	V	ertical lift gates	Drop Flat gates gates		Flat gates	Visor dams		Radial gates		Sector gates		Inflatable dams		
	S	$W \cdot S$	S	$W \cdot S$	S	$W \cdot S$	S	$W \cdot S$	S	$W \cdot S$	S	$W \cdot S$	S	$W \cdot S$
Ease of construction	7	175	6	150	6	150	1	25	6	150	2	50	2	50
Durability	7	175	7	175	6	150	6	150	6	150	5	125	2	50
Ease of maintenance	7	117	3	50	6	100	1	17	7	117	7	117	1	17
Sustainability	2	33	2	33	2	33	3	50	1	17	7	117	7	117
Ease of adaptability	7	58	6	50	2	17	1	8	2	17	2	17	1	8
Aesthetics	2	8	2	8	7	29	6	25	3	13	6	25	7	29
Navigation possibility	3	13	6	25	7	29	6	25	1	4	7	29	7	29
Total		579		492		508		300		467		479		300

Table 7.3: Score with weight per gate types, where W = Weight and S = Score.

## 7.4 Technical Requirements

In this subsection the requirements for design of the storm surge barrier are listed. These requirements follow from Section 3 and Appendix B and are needed to continue with the design of the structure. In Table 7.4 the different requirements with their corresponding values and dimensions are listed.

 Table 7.4:
 List of technical requirements

Requirements	Value	Dimensions
Design lifetime	100	years
Design storm	75	years
Maximum volume of water discharge in 2100	7800	$m^3/s$
Minimum volume of water discharge in 2100	250	$m^3/s$
Flow from both directions	Yes	-
Type of gate	Vertical lift gate	-
Mean sea level	0	m+MSL
Elevation of bottom of discharge sluice	-10	m+MSL
Water level spring high tide (HAT)	1.38	m+MSL
Water level spring low tide (LAT)	-3.01	m+MSL
Storm surge	2.18	m
Expected relative sea level rise	1.56	m
Maximum water level east seaside of structure	5.12	m+MSL
Minimum water level east seaside of structure	-1.45	m+MSL
Maximum water level upstream side of structure	3.00	m+MSL
Minimum water level upstream side of structure	-1.60	m+MSL
Maximum head over structure	6.54	m
Maximum wind speed	40	m/s
Significant wave height at location 1	2.2	m
Peak wave period during tropical storm	9.7	s
Fresh water density	1000	$\rm kg/m^3$
Salt water density	1020	$\rm kg/m^3$

## 7.5 Load combinations

The storm surge barrier has to be designed for various weather conditions. Each situation results in different loads on the structure which all have to be withstood. Each Load Combination (LC) can be seen in Table 7.5. This table describes on what part of the structure the loads act and also gives the water levels during extreme weather conditions for both low discharge and high discharge season separately. It is important to note that the "lowest river water level" differs for low and high discharge season. To both will be referred in this section as "low river", which means the lowest river level in the given LC. The same applies to the "highest river water level", and the "lowest and highest sea water level". Furthermore, the storm is only present during high discharge season. The differences between the minimum and maximum water levels of both the river and sea are described in Appendix E.2.

The different forces on the structure are wind loads, hydrodynamic forces, and hydrostatic forces. It is important to note that when the gate is closed, it is only partially exposed to wind forces, so only the top of the gate is influenced by it. Moreover, it is assumed that wave loads, which are included in hydrodynamic loads, solely originate from the sea and not from the river discharge. Self-weight is excluded since it does not vary for the different LCs. Below, every LC is explained.

In LC1, during high discharge season when the sea is much higher than the river, the gates of the barrier have to be closed to prevent salt intrusion and flooding by the sea. The most critical situation would occur for a low river discharge, combined with a high sea level. Because of the high discharge season, a storm will result in the highest sea level.

LC2 describes a failure of the opening mechanism of the gates during high discharge season, when the river is high and the sea is low. In this situation, the gates should be open to let the river discharge flow out. However, if the gates stay closed, the hydrostatic load will become significantly more on the river side of the barrier.

The third critical load combination, LC3, describes another failure of the lifting mechanism during high discharge season. In this scenario, the gates fail to close during storm conditions. In this scenario, the critical conditions involve a high sea and a low river discharge. Due to the high discharge season, the sea will be highest during a storm. With the gates open, there is no head difference between both sides of the structure. However, there is a high flow velocity, resulting in drag and inertia loads, which are categorised as hydrodynamic loads. Additionally, as the gates are open, they are completely exposed to wind loads.

The fourth critical load combination, LC4, describes a low sea in combination with a high river during wet season. Therefore, the gates are open leading the same loads as in LC3, but the flow velocities are different and in the opposite direction.

LC5 is similar to LC1, but during low discharge season. So this load combination also has closed gates, a low river discharge and a high sea level. There is no storm surge, since this load combination is during the low discharge season.

For LC6, again a low discharge season is analysed. This is similar to LC2 and describes a low sea and a seasonal high river discharge. The gates are again closed.

LC7 is similar to LC3 and describes an open gate during a high sea level and a seasonally low river discharge during the low discharge season.

In the last load combination, LC8, also an opened situation is analysed. The water level of the river is high and the sea water is low tide. This is similar to LC4, except now during low discharge season.

In Table 7.5 the LCs and their details are listed. Some of the LCs will turn out to be not critical for this structure but for complete overview all of them are described in this section.

LC	Season	Gate	Wind loads	Hydrodynamic loads	Hydrostatic loads	Water	levels
LC1	High discharge	Closed	Piers Lifting structure Gates (partial)	Piers, Gates	Piers, Gates	Sea: River:	RSLR High tide Storm surge Seasonal low
LC2	High discharge	Closed	Piers Lifting structure Gates (partial)	Piers, Gates	Piers, Gates	Sea: River:	RSLR Low tide Seasonal high
LC3	High discharge	Open	Piers Lifting structure Gates	Piers	Equal on both sides	Sea: River:	RSLR High tide Storm surge Seasonal low
LC4	High discharge	Open	Piers Lifting structure Gates	Piers	Equal on both sides	Sea: River:	RSLR Low tide Seasonal high
LC5	Low discharge	Closed	Piers Lifting structure Gates (partial)	Piers, Gates	Piers, Gates	Sea: River:	RSLR High tide Seasonal low
LC6	Low discharge	Closed	Piers Lifting structure Gates (completely)	Piers Gates	Piers, Gates	Sea: River:	RSLR Low tide Seasonal high
LC7	Low discharge	Open	Piers Lifting structure Gates	Piers	Equal on both sides	Sea: River:	RSLR High tide Seasonal low
LC8	Low discharge	Open	Piers Lifting structure Gates	Piers	Equal on both sides	Sea: River:	RSLR Low tide Seasonal high

Table 7.5: Load combinations and conditions

## 7.6 Loads

In this subsection, all the various loads that act on the structure are calculated. First, the hydrostatic loads and then the hydrodynamic loads for the different load combinations, which are discussed in Section 7.5. Then, the maximum wind load for a storm is calculated and the soil pressure is determined with Cone Penetration Test (CPT) data. Lastly, and overview of the complete load schemes, including hydrostatic, hydrodynamic, and wind loads for all LCs is given in Figure 7.7.

### 7.6.1 Hydrostatic loads

The hydrostatic load on the structure is caused by the static water pressure from both sides of the gates. When the gates are open, the water pressure solely acts on the piers, while when the gates are closed the water pressure also acts on the gates. When the gates are closed, the difference in water head on either sides of the structure results in a large static load from the side where the water level is higher.

For the hydrostatic loads, the situations with a closed gate are important. These are: LC1, LC2, LC5 and LC6. When the gates are open there is an equal water level on both sides of the structure, which results in a resultant hydrostatic force equal to 0 kN. LC3, LC4, LC7 and LC8 are therefore disregarded for hydrostatic loads. Furthermore, LC6 can be disregarded for the calculation of the hydrostatic loads since the water levels are not significant compared to LC2. A calculation for these different LCs is made in Appendix E.2.

### 7.6.2 Hydrodynamic loads

Besides hydrostatic loads, hydrodynamic loads can also exert forces on the structure. When the gates are closed, the wave loads from the sea act on the gates and piers. When the gates are opened, drag and inertia forces, caused by water flow, will act on the piers. The flow direction can change depending on the water height of the river and the sea. Lastly, it is important to note that for the calculation of the hydrodynamic loads, the maximum value of the quantity for location 1, 2 or 3 is considered. This is a conservative assumption. First, the scenario with closed gates, which results in wave loads. Note again, that the waves are only coming from the seaside. Therefore, the LCs where the gate is "closed" and where the "sea is high" will be critical, during the low and the high discharge season, which is the case for LC1 and LC5. For the calculation of the hydrodynamic loads, first the wave length has to be calculated. This is done with the linear wave theory because it gives a good approximation for wave propagation in coastal areas. For calculations of the wave length, refer to Appendix E.3. Wave forces on vertical structures consist mainly of two components. These are the incident wave, propagating

towards the structure, and the reflected wave propagating towards the incoming wave (Holthuijsen, 2007). Linear waves are non-breaking waves (Voorendt, 2022). The force on a vertical wall for non-breaking waves can be determined using the pressure distribution in vertical direction, see Appendix E.3.

The scenarios with open gates will also be analysed for hydrodynamic loads. Water flow past the piers, results in drag and inertia forces. The critical situations will occur when the flow speeds are high, so basically when there is a high water level difference. The flow speed from the sea to the river will be the highest when the sea is at its highest during either high or low discharge season. Therefore LC3 and LC7 are analysed. For the flow speed from the river to the sea, the situation has to be analysed where the river is higher than the sea, which is during LC4 and LC8, of which LC4 during high discharge season is critical. Therefore LC8 will be left out of the further calculation.

In the situation that the gates are open, a constant flow of river discharge will pass the piers. The piers can be regarded as slender structures as their length will be much larger than their width. Therefore, the Morison equation for flow past slender structures is be used to determine the forces acting on the columns. For the calculation of the drag and inertia forces with the Morison equation for LC3, LC4, and LC7, see Appendix E.3. Wheeler stretching is added to account for the wave impact on the structure. Wheeler stretching extends the height over which the water impacts the structure to the top of the wave height.

#### 7.6.3 Wind loads

The wind load on the structure follows a logarithmic curve. For preliminary design, the lower wind speed at still water level is neglected. The total wind speed is modelled as a constant pressure over the height. The wind dynamic pressure can be calculated with the following formula (Voorendt, 2022):

$$q = \frac{1}{2} \cdot \rho \cdot u^2 \cdot C_d \tag{1}$$

In which:

 $q = \text{Wind dynamic pressure } [\text{kN/m}^2]$ 

 $\rho = \text{Air density (1.250) [kg/m^3]}$ 

 $C_d = \text{Drag coefficient for wind } (0.7) [-]$ 

u =Wind velocity [m/s]

The maximum wind speed during a tropical storm is 40 m/s, while the average wind speed is 19 m/s. The values for the other variables are given in the description. This results in a total wind dynamic pressure of 0.7 kN/m<sup>2</sup> for storm conditions and 0.02 kN/m<sup>2</sup> for the average conditions. These forces are relatively small and therefore only the load combinations with open gates are assumed to include wind loading, since the area of the gate when lifted is very large. Furthermore, the wind dynamic pressure is assumed to be 0.7 kN/m<sup>2</sup> for all load combinations, which is an conservative assumption.

The wind puts pressure on the part of the structure that is above the water level during a storm. Integrating the pressure over this area leads to the total force on the particular part of the structure, which is a simplification of the real situation. The wind setup in this case is equal to the storm surge that is caused by a typhoon. This storm surge level is equal to 2.2 m as found in Appendix B.5.

#### 7.6.4 Soil loads

To determine the soil pressure on the structure, the different soil layers have to be determined. This can be achieved with a CPT. No CPT data near the Ham Luong River is available and therefore CPT data at Tien Giang province in the Go Cong Dong district near the river mouth of the Mekong is used. The cone resistance can be seen in Figure 7.6 and the resulting pressure can be seen in Table 7.6. The different ground layers have been determined with the CPT and borehole data of borehole LK01 which can be found in Appendix E.4. The CPT data is taken at a different site (near Vung Tau), but still close to the region. Therefore, it is still usable for the design. The site near Vung Tau is connected to a different river branch of the Mekong River which could result in a different CPT compared to the Ham Luong River. Especially for the layers close to the surface, since these layers are influenced by a combination of river discharge and longshore sediment transport.



Table 7.6: CPT results Vung Tau

(Truong Hong, 2012)								
Depth (m)	CPT	$qc (kN/m^2)$						
-11.6	4.5	4500						
-13	33	3300						
-16	21	2100						
-19	21	2100						
-22	25.5	2550						
-24	27	2700						
-27	27	2700						
-29	33	3300						
-34	37.5	3750						
-37	40.5	4050						
-40	61.5	6150						

Figure 7.6: CPT at Vung Tau (Truong Hong, 2012)

The soil consists of 5 layers. The top layer is a layer of fine grained sand. The second layer is a sand layer with a medium particle size. Layer three is a stiff clay layer. Layer four is a sand layer which is characterised by its medium tight state, an layer five is a medium to course sand layer (Truong Hong Son, 2012).

Depth [m]	Soil type
-11.6 to -13.6	Fine sand
-13.6 to -22.6	Medium sand
-22.6 to -25.6	Stiff clay
-25.6 to -40.6	Medium tight sand
-40 and lower	Medium to course sand

#### 7.6.5 Load schemes

Figure 7.7 gives a visual overview of all the LCs. LC6 and LC8 are left out because they are definitely not critical for any situation as explained in Section 7.6.1 and Section 7.6.2. In the figure, the closed gates are visible in grey and if the gate is open, the opening is drawn as a white rectangle. The hydrodynamic, -static, and wind loading are translated to a point load at the centre of gravity of the original distributed load. Additionally, the water levels are visible and also whether a storm is present, which is schematized by a wave.





## 7.7 Structural design

In this subsection a preliminary design of the storm surge barrier in the Ham Luong estuary is made. The preliminary design includes the design of the gates, sill, lifting structure, piers, foundation, and bed protection.

## 7.7.1 Gates

The gates will be placed between two piers with equal distances. The dimensions of the gate are very important. A larger gate size leads to less costs in construction and also a larger flow area for the release of river discharge during the high discharge season. However, the larger the gates the more difficult it is to lift them out of the water because of their weight. The height of the gates depends on the water height it has to withstand. The maximum water column consists of the depth, RSLR, Highest Astronomical Tide (HAT), spring tide, and maximum wave height. The height of the sill at the bottom is 1.5 m which reduces the height of the gate. The total gate height will therefore be 14.7 m.

There is a linear relationship between the opening size of the gates and the flow during mean tide for different storm surge barriers (Mooyaart & Jonkman, 2017). For this design, the mean discharge in the high discharge season is taken instead of the flow during mean tide. The mean discharge in the high discharge season is 4000 m<sup>3</sup>/s as visible in Figure B.3. Using the graph in Appendix E.5, showing the relationship between opening size and peak tidal flow during mean tide, the total opening area must be approximately  $A_{opening} = 10,000 \text{ m}^2$ . With a mean depth of 10 m, the total width of all openings must be 1000 m. With piers of 4 m in width each, this results in 23 gates of 40 m. A width of 40 m is chosen for the gates because it approximates the width of a similar storm surge barrier in the area called the Cai Lon barrier, see Section 7.2.1.

Since the gates will be very large, a 3D truss will be made behind each skin plate for support. A similar truss has been constructed for the Hartelkering (Figure 7.5a) and the Cai Lon barrier. The truss consists of three large lens-shaped frames that support the vertical girders and the gate plate facing the river mouth. The steel frame of the Hartelkering with two lens-shaped frames is visible in Figure 7.8.



Figure 7.8: The design of the gate of the Hartelkering (Voorendt, 2014)

The gates rest on a sill, which is a concrete barrier wall that rests on the river bed and is supported by a pile foundation. The gates are lifted into or out of position with a cable system. Therefore tracks are built into the piers on both sides of the gate with wheels in them to easily let the gates slide.

The three lens-shaped structures are formed such that they will withstand the total momentum caused by the combined hydrostatic- and dynamic loads. These are visualised in Appendix E.6. The total combined force is assumed to be evenly distributed over the number of lens-shaped frames. In case of an open gate during maximum wind speed, the load on the gate should also be significant. A quick calculation with the wind load from Section 7.6.3, gives the distributed force of the wind on the gates. It turns out that the maximum wind pressure is about the same as the maximum water pressure. Due to the fact that the gates will probably be closed during maximum wind speed, and that an extreme value is taken for the maximum wind speed, the calculation of the frame will be continued with the maximum water level.

The loads on the gate are the largest for LC1. The forces act over the total width of the structure resulting in a line load on the gate frame. The maximum momentum is easily calculated with a forget-me-not for a simply

supported beam and with load combination factors for static and dynamic loads as explained in Appendix E.7. These loads are again schematised in Section 7.6.5. To transfer the forces from the plate to the arcs, girders and beams have to be placed behind the gate that form a frame connected to the arcs. The chosen properties of the arc beams and the girders connecting these are shown in Table 7.8. Furthermore, an E-modulus of 210,000 N/mm<sup>2</sup> is used and a steel quality S235 is considered in the calculations.

Beam type	Outer diameter (m)	Thickness (mm)	Moment of inertia $(mm^4)$	Area $(mm^2)$
Arc Beam	1.4	35	$3.40 \times 10^{10}$	$3.04 \times 10^5$
Girder Beam	0.8	25	$4.57 \times 10^9$	$124 \times 10^5$

Table 7.8: Properties of arc and girder beams

The structure can be seen in Figure 7.9, visualised with MatrixFrame. The internal arm between the two arcs in the middle has a length of 6.8 m. There are 9 girders connecting the two arcs for stability that reduce in length when moving to the sides of the frame.



Figure 7.9: Top view of steel frame structure of the gate, showing the two arcs and nine horizontal girders.

To calculate the stresses in the frame, the loads acting on top of the structure, the seaside, are given as input in MatrixFrame. The hydrostatic load is regarded as a permanent load and the hydrodynamic load is regarded as a variable load. The schematisation can be seen in Appendix E.8, in addition to the results for the stresses and the deflection of the structure calculated in MatrixFrame. The structure is strong enough to withstand the forces, since the total stress in the members is lower than the yield stress of S235. The Unity Check (UC) is  $\frac{180}{235} = 0.77$ . It can be seen that the total deflection in the middle is about 34 mm. This is a deflection of less than  $\frac{1}{1000}$ L.

The horizontal beams connecting the two arcs are checked individually on their resistance to buckling. For this calculation the formula for Euler buckling is used, see Appendix E.9. Because the cross-section is the same in both directions, only one direction has to be checked for this calculation. All girders have different loads but since the structure is symmetric only the first five girders have to be analysed. Girder 1 is the smallest located at the end of the truss and girder 5 is the middle one, which is the largest. The results for the buckling load of the girders are visualised in Table 7.9. The normal force caused by the loads is checked with the critical load for buckling, resulting in a unity check. All girders meet the requirements very easily with UCs << 1.

Girder	$L_b$ (mm)	$P_{cr}$ (kN)	N (kN)	UC
1	3000	$1.05 \times 10^6$	749	0.0007
2	4600	$4.48 \times 10^5$	785	0.0018
3	5600	$3.02 \times 10^5$	780	0.0026
4	6400	$2.31 \times 10^5$	759	0.0033
5	6800	$2.05 \times 10^5$	726	0.0035

Table 7.9:	Girder	buckling	check
------------	--------	----------	-------

The vertical girders that connect the frames to the gate plate have been checked in a similar way. The specific geometry can be seen in Figure 7.10a. The supports are the locations where the frame is connected to the vertical girders. The distance between girder centres is 4.0 m, as visible in Figure 7.9. The loads acting on the plate are distributed to the girders, following the distribution of hydrostatic and hydrodynamic loading acting on that plate, which can be seen in detail in Appendix E.6. This loading is multiplied with the centre to centre distance between the girders to calculate the distributed loads that can be seen in Appendix E.10. Both loads are multiplied with their corresponding safety factors to obtain the results. The resulting stresses and displacements are visualised in Figure 7.10b and Figure 7.10c. It can be seen that the stresses do not exert the maximum stress for S235. The UC is  $\frac{200}{235} = 0.85$ . The deflections are at maximum only 9 mm for a length of a single beam of 7.5 m. This is less than  $\frac{1}{800}$ L.



Figure 7.10: Vertical girder geometry, deflection and stresses

The most common skin-plate thickness for large hydraulic structures is 12 or 14 mm (Daniel & Paulus, 2019). Therefore, an initial thickness of 14 mm is chosen. This is a thin steel plate and, in order to support this plate, stiffeners that will redirect the forces to the vertical girders must be added to the construction.

#### 7.7.2 Sill

The width of the sill should not be less than the width of the gate, which is 6.8 m. Consequently, a width of 7 m is chosen for the sill. The height is based on a simple design rule which results in a sill height of  $\frac{1}{25}L = \frac{1}{25}(40) = 1.6$  m (Tromp, 2013). For simplicity, a sill height of 1.5 m is chosen.

#### 7.7.3 Lifting structure

On top of the pier, there is a lifting structure designed to lift the gate, allowing the maximum river water level to flow beneath it. This maximum river water level is 10 m, with an additional 3 m during high discharge, as indicated in Appendix E.2. This results in a lifting structure height of 13 m. The total height from the riverbed to the top of the lifting structure is therefore 16.2 m (pier) plus 13 m (lifting structure) = 29.2 m. See Appendix E.11 for the visualisation. The lifting barrier is made from reinforced concrete and it is assumed

that it is square-shaped. The width and length will be the same as the width of the pier, which will be given in Section 7.7.4.

The height of the lifting structure is designed for the maximum river water level. During a failure of closure of the gate during a storm (LC3), the water level and waves would reach the bottom of the gate which would cause an extra force. This will not be significant compared to the other loads. This is better explained in Appendix E.12.

#### 7.7.4 Piers

The pier is made out of reinforced concrete and distributes the forces from the gate to the foundation and soil. It has to resist the loads resulting from the sea and river and is an important part of the design of a storm surge barrier. The width of the pier is assumed to be 4 m, which is approximately the same width as the piers of the Cai Lon - Cai Be sluice. The pier should have a height equal to the gate height, which is 14.7 m as stated in Section 7.7.1, plus the sill height of 1.5 m. This results in a pier height of 16.2 m.

The length of the pier should be sufficient enough to have the required rotational stability. Furthermore, the length of the pier also has an influence on the sliding resistance and force distribution to the soil. The rotational stability will be calculated for LC1, in which the bending moment and horizontal force are the largest. The load combination factors, which are explained in Appendix E.7, are used in these calculations. The static loads are permanent loads and the hydrodynamic and wind loads as variable. The moments and horizontal forces resulting from the external loads are visible in Appendix E.13. The moments resulting from the internal forces (self-weight) are visible in Appendix E.11 The first estimation for the pier length is 14 m and will be verified in the following calculations.

To prevent rotation of the pier around its corners a calculation was done. The dimensions and forces for the design of the storm surge barrier are schematised in Appendix E.11. To calculate the moment resulting from the weight of the gate, the gate was chosen to be located close to the seaside between the piers. This is due to the fact that during conversations with SIWRP there was a possibility that a coastal highway would be included in the design of a potential storm surge barrier, as stated in Appendix F.1. It could also be more efficient to resist the large external forces from the sea. It is important to note that the barrier will be less efficient in resisting the forces coming from the riverside. Therefore, the maximum LC for the opposite direction should also be taken into account, which is LC2 as indicated in the table in Appendix E.13. The calculations show that the maximum bending moment from external hydrostatic and hydrodynamic loads at two points for the extreme LCs is sufficiently resisted by the self-weight of the structure. The values can be found in Appendix E.13. The distance from the middle of the lifting structure to the end of the pier (at the seaside) is chosen to be 4 m. Appendix E.14 shows the location of the lifting structure with a top view of the barrier.

To resist the sliding of the barrier, the horizontal stability is analysed. The maximum external horizontal force takes place during LC1 with a horizontal force of 55 MN as visible in Appendix E.13. By multiplying the friction coefficient with the sum of all vertical forces, the resistance against sliding is calculated, which should more than the acting external forces. The formula that is used is Equation (2):

$$\sum H < f \cdot \sum V \tag{2}$$

In which:

 $\sum H =$ Sum of horizontal forces

 $\mathbf{f} =$ Friction coefficient

 $\sum V =$  Sum of vertical forces

The total sum of the vertical forces from all elements, shown in Appendix E.11, including the gate and sill, is 33 MN. The friction coefficient for clean fine to medium sand is equal to 0.45 to 0.55. Therefore, a conservative value of 0.45 is chosen for the friction coefficient. The resistance against sliding is consequently 15 MN which is less than the acting external horizontal sum of forces, equal to 55 MN. This means that the foundation should be designed for the resultant force of 40 MN to make sure that the structure will not start sliding.

#### 7.7.5 Foundation

Both the sill beams and the piers will transmit the external forces to the pile foundation. The structure will have a pile foundation that rests on the strong sand layer located at MSL -25.6 m, see Section 7.6.4. The upper soil layers are not strong enough to support the structure. This makes the piles approximately 15.6 m long, but the length of individual piles vary depending on the location. The piles will be made out of concrete because steel foundation piles can corrode in wet conditions. Prefabricated piles are used because it is better for the ease of construction.

Horizontal and vertical loads will have to be transmitted resulting in vertical and horizontal loads on the piles. These loads are caused by the self-weight of the structure and the resulting horizontal load from hydrodynamic and -static loading. These loads have the effect that the structure might slide or rotate as mentioned in Section 7.7.4, which is why the foundation is extremely necessary. Piles that can take horizontal forces are driven under an angle to the vertical and are called 'raking piles' or 'batter piles' (Voorendt, 2022). The piles must be able to withstand horizontal forces in both directions the river can flow. An example of an embankment with batter piles is shown in Figure 7.11. If necessary, the batter piles could also be installed at multiple locations.



Figure 7.11: Piled embankment with batter piles, indicated as 'inclined piles' (Sakleshpur & Madhav, 2013).

#### 7.7.6 Bed protection

When building a structure in a river, scour can occur because of a change in water flow. This can lead to a dislocation of soil layers on the river bed near the structure, which can cause instability of the structure (Voorendt, 2022). When scour is severe, bed protection can be used to protect the river bed from scour.

There are two types of scour that can occur at the storm surge barrier; one is scour due to flow acceleration through the open gates, and the other is scour induced by waves on the closed structure. Calculations regarding both types of scour, and the related bed protection are described in Appendix E.15. A more detailed analysis on bed protection would be needed when including the occurring flow speeds during operation of the barrier. In the time span where the vertical lift gates are lowered towards the sill, significantly higher flow speeds are expected underneath the gate on the sill. This bed protection is expected to be decisive for the area in near proximity of the gates. However, due to lack of information on the exact behaviour during operation, this bed protection is not included in this analysis.

To prevent scour near the structure, a total length of 850 m of scour protection into both flow directions is needed. The upper layer of this scour protection should consist of the grading  $LM_A$  40-200, which has a  $D_{n50}$ of 34 cm. This grading prevents any movement of the bed protection. No calculations on the filter layers below the top layer are made, but simple filter stability calculations would lead to a sufficient design.

The expected scour depth as a result of wave action is very small, and one can assume that the designed bed protection to prevent clear-water scour is sufficient to prevent wave-induced scour.

## 7.8 Operation

The main function of the structure is that it must be closed in case of low discharges from the Ham Luong River and during storm surges, which can occur during both low and high discharges. The goal of the structure is to prevent saltwater intrusion and flooding from the sea while retaining fresh water in the estuary. The storm surge barriers' secondary functions are to open during high river discharges to let the river water flow into the ocean. This can be problematic in the case that there is high discharge from the river and a tropical storm at sea. The water from the sea can be stopped from entering the land but the river will still have a large discharge towards the sea. Because a storm usually occurs over the course of 1 to multiple days this will not cause a large problem. The river will probably not overflow the dykes in this time span. It is advised that when a barrier is built extra research into this situation is performed.

In the case of the opening or closing of the barrier, all of the gates will operate simultaneously. There is no need for very large ships to pass the barrier. The smaller ships can travel to other branches of the Mekong River to reach their destination. When the gates are put into action the sluices along the river will have to be closed as well to prevent the water travelling a different way. This must be coordinated precisely. If future research shows that a navigability option is required, a sluice could be constructed. Therefore the sluice and abutment are also depicted in Appendix E.14 with unknown dimensions.

It is difficult to say something about the effects of the barrier on water levels and salinity in the other river branches. Operation with structures that might be built in the future in other river branches is therefore not considered in this report.

# 8 Impact Evaluation

In this section the different alternatives will be evaluated on a set of criteria. Alternative 1 is a barrier at location 1, Alternative 2 is a barrier at location 2, Alternative 3 is a barrier at location 3 and Alternative 4 is no barrier but the dykes along the river are heightened to 7.8 m. For Alternatives 1, 2, and 3 the dykes between the barrier and the river mouth are heightened to 7.8 m and the other dykes behind the barrier are heightened to 4.0 m,as described in Section 7.1. The alternatives will be assessed on the criteria of freshwater supply, agricultural and aquaculture adaptation, biodiversity, stable riverbanks and navigability.

All alternatives will obtain a total score for all the criteria combined. Additionally, the alternatives will be assessed on costs, which is not one of the criteria in the criteria set but an independent factor. After these assessments, the preferred solution for the Ham Luong estuary will be selected.

## 8.1 Criteria assessment

This section describes how each criterion is evaluated. Criteria are assessed using a rating system ranging from 1 (the worst score, represented as "- - -") to 7 (the best score, indicated as "+++"). This score is relative to the situation now in the Ham Luong estuary. This means if a alternative has a ranking of "+++" it has the best possible scenario relative to the present. As mentioned in Section 4.1, "flood safety" is not considered as a criteria in this assessment but as a design requirement of the flood protection systems.

### 8.1.1 Freshwater supply

This criterion assesses the effectiveness of the different alternatives in retaining fresh water in the Ham Luong estuary. As mentioned in the introduction of Section 6.1, the scope of the project hinders the possibility to evaluate the salinity levels upstream of the closed barriers with the Delft3D model. For this reason, the salinity levels upstream of the barriers cannot be compared to the salinity levels downstream of the barrier, and no conclusion can be drawn on the effectiveness of blocking the saltwater intrusion with a river barrier based on the results of Delft3D. Hence, the evaluation of this criterion is based on a qualitative assessment. From theory and reference projects, it is generally understood that when a barrier is closed it retains fresh water upstream. The further a barrier is located to the sea, the more volume of fresh water can be retained upstream of the barrier. As this is more beneficial for the freshwater supply, Alternative 1 is awarded a +++, Alternative 2 ranks ++ and Alternative 3 receives a +. Alternative 4, as no intervention is placed in this system, is not able to retain any fresh water and therefore scores - -.

### 8.1.2 Agricultural and aquaculture adaptation

The type of agricultural or aquaculture product that is cultivated in an area in the Ham Luong estuary is highly dependent on the salinity of the water at that location. In assessing the impact of the different closure alternatives on salt-intrusion, and thus on the types of agri- or aquaculture, there are three aspects which should be taken into account. The first is the time a farmer has to adapt to the new circumstances created by the alternative, the so-called "adaptation time". In case the Ham Luong estuary is closed off with a barrier, the system experiences a significant and relatively abrupt change, which prompts farmers to adjust their businesses in a short period of time. Whereas no intervention in the system will give climate change the free hand, and the estuary (and thus the farmers) will undergo gradual changes. The second aspect to bear in mind, is the profit a farmer makes per agri- or aquaculture type. Section 3.4.1 mentions the incomes of farmers per agriand aquaculture type per hectare; shrimp: 45 million VND/ha, coconut: 70 million VND/ha, rice: 16 million VND/ha. Last but not least, the size of the agricultural or aquaculture area that is impacted by the changes in salinity also plays an important role. It is important to note that when a farmer would change to a different type of product, the investment cost, the complexity of the crop, the amount of new knowledge that is needed to cultivate a different product, and the risks involved in changing to a new business are usually considered negative. Therefore, the larger the area that needs to adapt to a new agricultural or aquaculture product, the more resistance can be expected on account of the aforementioned terms.

Besides these three important aspects, it should also be kept in mind that it is very difficult to predict exactly how the agricultural or aquaculture land use will change due to salt-intrusion. This is also highly dependent on the willingness of people to switch to a different product, which is something that is hard to predict. In order to be able to make an assessment, a number of assumptions have been made, which are stated in the list below. However, it should be kept in mind that changing these assumptions can possibly lead to a different outcome.

These assumptions are:

- A larger 'adaptation time' is considered more favourable compared to a shorter period of time in which farmers have to adapt.
- According to Appendix B.9, shrimp can be produced between salinity levels of 18 ppt and 24 ppt, coconut can be produced if the salinity level is  $\geq 4$  ppt, and rice can be produced if the salinity level is  $\leq 4$  ppt.
- If an area is not subject to changes in salinity levels, the type of agri- or aquaculture will not change.
- Whenever an area is subject to increased salinity values at which the current crops are no longer able to grow, the area will switch to coconut farms as this will lead to the highest profit for the farmer.
- The income per product remains constant over time.
- As mentioned in Section 5.5.3, the Delft3D model was not able to predict the exact salinity concentrations and salt-intrusion lengths. Instead, in this assessment the salt-intrusion length and concentration were used that was present in the Ham Luong estuary in 2020, as can be seen in Figure 3.1.
- The barrier will be closed for 2 months in 2100 and the salinity levels in the Ham Luong River can be observed in Figure 6.5. It is important to note that these salinity levels in the Ham Luong River are underestimated.

These assumptions were combined with the outcomes of the Delft3D model and have been translated to aerial maps which are depicted in Figure 8.1. According to Delft3D, the salinity levels at the seaside will be too high for shrimp farms if the Ham Luong estuary were to be closed off. Therefore, in case of closure, the shrimp farms near the coast will be replaced by coconut farms. In Alternative 4, fresh water still flows through the river in the dry season and the salt level will be less extreme at the seaside. For this reason it is expected that shrimp farms will remain at their original location. Estimating the other agricultural and aquaculture land use is difficult, but it can be expected that the salinity levels for rice will be too high and existing rice paddies will change to coconut farms. Agricultural land more inland of Ben Tre will probably be a mixture of fruit and coconut.



Figure 8.1: Rough visualisation of the distribution of agricultural and aquaculture land use in 2025 and the expected land use in 2100 for the alternatives.

In Table 8.1 an overview of each aspect is visible. It should be mentioned that this is a rough estimation. The area size and income are estimated with Figure 8.1. In each alternative some area should adapt to the changing salinity levels. Alternative 3 has the biggest area that should adapt. Relative to this area Alternative 4 has a slightly smaller adaptation area, while Alternative 1 and 2 have the smallest adaptation area. Each alternative has a higher estimated income than the current situation, due to a higher coconut area. Alternative 3 has the highest coconut area and thus the relative highest income. Alternative 4 has a somewhat lower income than Alternative 1 and 2 have the lowest relative income.

	Alternative 1	Alternative 2	Alternative 3	Alternative 4	
Adaptation	Short adaptation time	Short adaptation time	Short adaptation time	Long adaptation time	
$\mathbf{time}$					
Area size	Small area	Small area	Large area	Intermediate size area	
Income	Relatively low income	Relatively low income	Large income	Intermediate income	

 Table 8.1: Impact evaluation agricultural adaptation

Examining Table 8.1, it becomes challenging to distinguish between the alternatives based solely on adaptation area and income. Alternatives 1 and 2 offer a smaller adaptation area, which has its advantages, but this is offset by lower income. Conversely, alternatives 3 and 4 require a larger adaptation area and result in higher income. Nevertheless, a clear differentiation becomes evident when considering the adaptation time; Alternatives 1, 2, and 3 having notably shorter adaptation periods compared to Alternative 4. For this reason Alternative 1, 2 and 3 will get a score of + and Alternative 4 a score of ++. Every alternative has a higher net income than the current situation, and this fact contributes to assigning a positive score to each of the alternatives.

### 8.1.3 Biodiversity

As mentioned in Section 3.3.4, the Ham Luong estuary is home to a significant population of fish that migrate through the channel. The species rely on the mixture of fresh, salt, and brackish water. The installation of a barrier would make it more challenging for fish to migrate through the channel. Also, most fish have migration cues, triggered by variation in water level or discharge. This principle, which was explained in Section 3.3.4, is also threatened in the case of a barrier.

Furthermore, installing a barrier would split the salt and fresh water when it is closed, reducing the mixture of fresh, salt, and brackish water. As mentioned in Section 3.3.4, mangroves also require a combination of fresh and salt water to thrive. If the barrier is placed further inland, there is a greater length of river over which this mixing can occur. Therefore, an inland closure is preferable to one near the sea. Furthermore, blocking the river with a closure might trap more pollutants and decrease the river flow. This results in a more polluted river and thereby reducing biodiversity.

Another process that threatens biodiversity is sedimentation. As stated in Section 7.2.2, the sedimentation near the Ba Lai sluice had negative impact on the biodiversity. Therefore, the outcomes of the sedimentation tool, as described in Section 6.2, are used to assess the effect of biodiversity. When a barrier in the Ham Luong estuary is closed for multiple days, the mud concentration at the seaside of the barrier increases most for Alternative 1, which is shown in Figure 6.8. Increase in mud concentration can affect the ecology of the estuary in different ways:

- The light penetration reduces, which limits photosynthesis.
- Water quality reduces, because mud can contain heavy metals and pollutants.
- Available food for filter-feeding species increases.
- Species could adapt or new species that are adapted to handle this change in sediment concentration could settle in this area.

Assuming that the closure of a barrier would not last longer than a few consecutive months, the adaptation and change of species does not outweigh the reduction of light penetration and water quality. This, combined with the effect on fish migration and mangrove growth, means that Alternatives 1, 2, and 3 all exhibit adverse effects on biodiversity relative to the 2025 situation, resulting in scores of - - . Alternative 4 receives a score of 0 since it entails no interventions that either positively or negatively affect biodiversity in the Ham Luong estuary. There are possibilities to act to increase biodiversity and help nature recover from current pollution, such as implementing Building with Nature (BwN) solutions. These are not taken into account in this alternative, which is why it is given a rating of 0.

### 8.1.4 Stable riverbanks

In the province of Ben Tre, the different agricultural and aquaculture activities and several residential areas are located close to the riverbanks. Erosion of the riverbanks can lead to damage to structures or agricultural land, and can consequently lead to the forced resettlement of inhabitants. As stated in Section 4.1, this criterion does not include erosion near the barrier itself, as the stability near the barrier is considered as a technical requirement, which will be fulfilled by adequate bed protection. This is discussed in Section 7.7.6.

In assessing the stability of the river banks, two sources of information will be used: Delft3D results and literature theories. The sedimentation/erosion rates from the Delft3D model indicate short-term effects only, during the simulated 5 months. From Figure 6.6 and Figure 6.7 in Section 6.1.4 it follows that the only difference in erosion or sedimentation rates between the four alternatives is observed in the direct proximity of the barrier locations. These rates are the result of local acceleration and deceleration of flow. These direct impact of the erosion rates can be counteracted by bed protection. When assessing the criterion of stable riverbanks, impact is defined to be negative if the banks show erosive behaviour on any location. Therefore, based on the short-term results of the Delft3D model, it can be concluded that all alternatives show negative effects on the stability of the riverbanks and no distinction can be made between the four alternatives.

Theoretical knowledge and literature on the effect of river barriers on downstream erosion rates include longterm effects. In the analysis of the impact of storm surge barriers on sedimentation processes, it has been found that the presence of a storm surge barrier leads to erosion of the tidal flats in an estuary, which are also present in the Ham Luong estuary. An example of those tidal flats can be seen in Figure 8.2. This was found in a research on the Western and Eastern Scheldt in the Netherlands (de Vet, van Prooijen, & Wang, 2017), as well as a research on the Haringvliet in the Netherlands (De Ronde, 1990). In order to illustrate the relevance of these findings for the Ham Luong estuary, the expected changes of a barrier on the estuary will be explained. First of all, as stated in Section 3, the Ham Luong estuary can be described as a tide- and wave-dominated estuary. The morphology of tide-dominated estuaries are expected to depend strongly on the tidal behaviour within the estuary (Bosboom & Stive, 2023). As stated in Coastal Dynamics by Bosboom and Stive (2023), the long-term morphodynamic responses on changes in estuary equilibrium can be described by empirical relations. These empirical relations link the tidal prism to the volume of the tidal channels. It is generally known that the construction of a storm surge barrier in a tidal inlet, such as the Ham Luong estuary, will reduce the tidal energy, and therefore reduce the tidal prism of an estuary (de Vet et al., 2017). As the Ham Luong estuary is tide- and wave-dominated, this change in tidal energy is expected to lead to morphodynamic changes. A reduction in tidal prism leads to a reduction of the needed tidal channel volume. As a response, the tidal channels are filled by sediment to reach the new equilibrium channel volume. This new sediment need of the tidal channels leads to less sediment transport to the tidal flats, and therefore erosion of the tidal flats. In the context of the above described theory, one should bear in mind that the river discharge and the corresponding sediment supply by the river is not included in this theory. However, as this river sediment supply is identical for all alternatives, it is not expected to affect the mutual differences in erosion patterns of the tidal flats. The above described theory can not be used to describe any difference in erosion between the three different possible locations of the barrier. However, it can be concluded that the alternative of no barrier will lead to less erosion of tidal banks, as no change in tidal prism occurs.

Considering all the erosion causes mentioned previously, there is no clear distinction among the three proposed barrier locations. However, Alternative 4 is expected to have the least impact on erosion rates, mainly because there are no substantial interventions in the estuary that would alter erosion rates. Consequently, Alternative 4 is assigned a score of 0, while Alternatives 1, 2, and 3 receive a score of -.



Figure 8.2: Tidal flats along the Ham Luong estuary. The photo was taken at Ben do An Hoa Tay

### 8.1.5 Navigability

Navigability is a criterion that is important for people that make use of the Ham Luong River for shipping. As mentioned in Section 3.4.7, there are fishing boats and tourist boats at small ports and near Ben Tre, in addition to local boats in the side channels. A barrier would possibly obstruct the passing of these boats. If the barrier were built close to the river mouth, sea fishing boats would be forced to go through the barrier which might only be possible under certain weather and flow conditions. A sluice would have to be added to let boats pass through.

The city of Ben Tre is very far from any of the barrier locations. It is a 2-hour boat tour from Ben Tre City to location 3. Tourist boats do not travel that far. Therefore, the placement of the barriers will not have a significant effect on the tourist boats. The boats that are located in the side channels are used for these channels mainly. The fishing boats located in An Thuy Port are therefore the most important stakeholders. For these boats, no river barrier would mean no effect on the navigability compared to the situation of 2025 and therefore this alternative receives a score of 0. A barrier closer to the river mouth would influence the boats more in a negative way. Therefore location 1 will get a score of --, location 2 a score of --, and location 3 a score of --.

#### 8.1.6 Overview criteria assessment

Stable river banks

Navigability

In Table 8.2 all scores mentioned above are summarised.

Criterion	Alternative 1	Alternative 2	Alternative 3	Alternative 4
Freshwater supply	+++	++	+	
Agricultural adaptation	+	+	+	++
Biodiversity				0

 Table 8.2:
 Alternatives evaluation

0

0

### 8.2 Criteria performance of the alternatives

In this section, all criteria are given a weight corresponding to their importance. The importance of each criterion is evaluated based on the system description (Section 3), the field trip (Appendix A), meetings with experts in Ho Chi Minh City (HCMC) (Appendix F), and the survey (Appendix G). The weight determination is shown in Table 8.3. If a criterion is regarded more important than another one, the value of 1 is assigned to the important one and the value 0 to the less important one. If criteria are equally important they both receive a rating of 1. In this way all of the criteria will obtain scores which are eventually summed up. When a criterion receives a total value of 0 it eventually is assigned a value of 1 make sure that it is not let out of consideration in further assessment, and all other criteria are doubled. To obtain the weight for each criterion the doubled totals are summed together. The score of each criterion, divided by the total score and multiplied with 100, results in the weights.

Table 8.3:	Weight	determination	of	the	criteria
------------	--------	---------------	----	-----	----------

Criterion	a	b	c	d	e	Summed	Doubled	Weight
						total	total	
Freshwater supply (a)	x	1	1	1	1	4	8	35
Agricultural and aquaculture adaptation (b)	1	х	1	1	1	4	8	35
Biodiversity (c)	0	0	x	1	1	2	4	17
Stable river banks (d)	0	0	0	x	1	1	2	9
Navigability (e)	0	0	0	0	x	0	1	4

The criterion of freshwater supply is the most important one because the shortage of fresh water is regarded as the highest risk to the people in the area. It is a primary need for everyone. Agricultural and aquaculture adaptation is just as important as freshwater supply. This is because agriculture and aquaculture is by far the most important occupation in the area. Biodiversity is important for both the people and the environment but is less crucial than freshwater supply and agricultural adaptation. Therefore it is regarded third in importance. Stable river banks are less important because it does not influence the province of Ben Tre less compared to the earlier mentioned criteria. Lastly, the navigability is rewarded with the lowest importance for the Ham Luong River because there are no large ships that make use of the river and shipping is not a very important business for this specific part of the Mekong Delta.

The final results of the scores of the different alternatives can be seen in Table 8.4. The weights have been multiplied with the corresponding score of the alternatives on each of the criteria. This leads to the total score.

Criterion	Alternative 1		Alternative 2		Alternative 3		Alternative 4	
	S	W * S	S	W * S	S	W * S	S	W * S
Freshwater supply	7	245	6	210	5	175	1	35
Agricultural and aquaculture adaptation	5	175	5	175	5	175	6	210
Biodiversity	2	34	2	34	2	34	4	68
Stable river banks	3	27	3	27	3	27	4	36
Navigability	1	4	2	8	3	12	4	16
Total score		485		454		423		365

Table 8.4: Weighted scores for alternatives, where W = Weight and S = Score.

## 8.3 Cost analysis

The costs depend on multiple factors. The option to build a barrier or not, the amount of (river) dykes that have to be constructed or heightened, and the maintenance costs of a barrier and the dykes all have a large influence. The length of the dykes that have to be heightened or built varies depending on the location of the barrier. Furthermore, when no barrier is built all of the river dykes have to be heightened to at least the maximum water level in the river.

The amount of information on the costs of coastal defences in Vietnam is very limited. Most estimations in literature are therefore based on simple calculations combined with expert judgement. The costs of building a storm surge barrier are not easy to estimate and vary a lot for different projects. In 2009, the cost price per metre width for storm surge barriers worldwide ranged between  $\in$ 500.000 and  $\in$ 2.700.000 (Jonkman et al., 2013). In Vietnam, the costs will probably be a bit lower due to low labour costs. The Cai Lon - Cai Be sluice gates in Vietnam have only cost  $\in$ 136.8 million (Tuoi Tre News, 2022). More information about this barrier can be found in Section 7.2.1. This project was for a river that is 500 m in width, which is about 3 to 4 times as small as the Ham Luong River. However, it shows the order of magnitude of the costs for building such a structure in Vietnam. Therefore, for the cost calculations an estimate of  $\in$ 300 million per kilometre is used.

The management and maintenance costs of complex storm surge barriers are very high. These costs have been estimated at 5-10% of the total construction costs (Jonkman et al., 2013). For the maintenance costs of this barrier, 5% is taken for further calculations since labor costs in Vietnam are relatively low.

Dyke heightening costs depend on the amount of heightening. For a kilometre stretch, the dyke heightening costs are approximately €750.000 for every metre of heightening. In 2009, maintenance of the dykes in Vietnam cost roughly €20,000 per km dyke per year (Hillen et al., 2010). For the calculation of today's value, an inflation rate of 5.76% is used. This has been the mean inflation rate in Vietnam from 1996 until 2023 (Trading Economics, 2023). It is observed that the maintenance costs of the dykes are between 2-3% of the initial costs of the dykes annually. This is common for dykes in Vietnam (van Ledden et al., 2020). The costs of the barrier and dykes per kilometre can be seen in Table 8.5.

	2009	2023
Storm surge barrier cost $(\in)$	-	300,000,000
Maintenance costs/year for barrier ( $\in$ )	-	15,000,000
Dyke heightening costs per metre $(\in)$	750,000	$1,\!640,\!000$
Dyke maintenance costs/year ( $\in$ )	20,000	43,800

Table 8.5: Coastal defence costs for a km stretch

For all barrier locations, the crest of the river dykes will be heightened to 6.8 m between the river mouth and the location of the barrier. This is 3.8 m higher than in the current situation. Furthermore, a wall is placed on top of the dykes which makes the total height 7.8 m. However, this wall is left out of the calculation of the costs, as no good estimate for the costs of this wall could be found. The crest height of the dykes located more inland will be heightened to approximately 4.0 m, which is only 1 m higher than in the current situation. For the calculation of the dyke height see Section 7.1. In the case no barrier is built, the dykes along the entire length of the Ham Luong River will be heightened to 7.8 m. The location of the different dykes can be seen in Figure 7.2. Furthermore, the total length of the barrier is dependent on the location of the barrier, which influences the costs a lot. The total initial costs of the dykes and storm surge barrier for each alternative are shown in Table 8.6. Additionally, the total maintenance costs for every alternative can be seen in Table 8.7. It has to be noted that these costs only include measures within this project's spatial scope. Additional needed measures in upstream areas are not included.

	Alternative 1	Alternative 2	Alternative 3	Alternative 4
km storm surge barrier	2.25	2.65	1.60	0.00
km dyke heightened to $4.0 \text{ m}$	118.20	107.50	90.00	0.00
km dyke heightened to $7.8~\mathrm{m}$	20.20	30.90	48.40	138.40
Total initial costs (million $\in$ )	995	1,164	929	863

 Table 8.6:
 Initial costs of the different alternatives

 Table 8.7:
 Maintenance costs of the different alternatives per year

	Alternative 1	Alternative 2	Alternative 3	Alternative 4
Storm surge barrier maintenance (million $\in$ )	34	40	24	-
Dyke maintenance (million $\in$ )	6	6	6	6
Total maintenance costs (million $\in$ )	40	46	30	6

The initial costs and maintenance can be combined into a total cost for the 2100 forecast. The maintenance is assumed to remain constant in this calculation. There is a clear difference in costs for the different alternatives as is visible in Table 8.8. From most costly to least costly the order is: 2, 1, 3, 4. This is seen for both initial costs and maintenance costs.

**Table 8.8:** Total costs of the different alternatives for 2100 forecast

	Alternative 1	Alternative 2	Alternative 3	Alternative 4
Total initial costs (million $\in$ )	995	1,164	929	863
Total maintenance costs (million $\in$ )	3,066	3,528	2,315	467
Total costs (million $\in$ )	4,060	4,691	3,244	1,329

## 8.4 Preferred solution

From Section 8.2 it can be seen that the different alternatives have a total score that is relatively close to each other. The lowest score for Alternative 4 is 75% of the score of Alternative 1, which is the highest scoring alternative. At first glance, the difference of 75% between the best and worst alternative might look significant. Nevertheless, it is important to state that the outcome of the MCA is very sensitive to the weights of the criteria. This makes it difficult to identify one alternative as the best fitting alternative. However, the costs of the different alternative differ a lot. Alternative 4 is clearly the alternative that is the most cost-efficient. For this reason, when considering cost-efficiency as decisive aspect, Alternative 4 is the preferred solution for the Ham Luong estuary. This is the option where no barrier is built in the river and the dykes along the entire length of the Ham Luong estuary are heightened to 7.8 m.

# 9 Conclusion

The conclusions of this report are written down per research question. This will give a clear answer to the objective of this report; to find out what the impact of various closure alternatives will be on the hydraulic and social activities in the Ham Luong estuary, in the year 2100.

1. What are the key characteristics of the Ham Luong estuary considering economic and social activities and hydro-morphodynamic properties?

Ben Tre faces multiple challenges such as salt intrusion, land subsidence, erosion, and a rising sea level. The province has set up defences such as dykes and sluices. Land use and occupation of the inhabitants of Ben Tre are arranged based on the salinity levels. From east to west, the distribution of land use is: shrimps, paddy, coconut, and lastly fruits. The most important stakeholders are the agricultural sector, who make up 60% of the workforce, and the local authorities and Vietnamese government, who have the highest influence.

2. What is the impact of the partial closure of the Ham Luong estuary on the hydro- and morphodynamics of the system?

To determine the impact of the partial closure of the estuary, four alternatives are taken into account. Alternative 1, 2, and 3 have a barrier at different locations along the Ham Luong estuary and Alternative 4 has no barrier. In all alternatives the dykes are heightened. Delft3D is used to calculate and display the water level, sediment, and salinity levels in the Ham Luong estuary. The used Delft3D model is a result of a lot of assumptions and simplifications, which are further discussed in Section 10. Due to these assumptions and simplifications, it becomes challenging to precisely determine the effects of a partial closure on water levels, salinity levels, and sedimentation changes in the Ham Luong estuary. However, it is expected that salt intrusion will decrease, and both sedimentation and erosion rates will undergo alterations as a result of the partial closure. Due to RSLR, the maximum water levels in the Ham Luong estuary in 2100 will often be higher than the current river dykes, consequently these dykes should be heightened.

3. What would be the optimal design for a flood protection system?

For every alternative an optimal design has been made. For the alternives with a storm surge barrier the gates would be spanning 40 m and have a height of 14.7 m. These are connected to the piers, measuring 4 m in width and 14 m in length. The dimensions are based on all possible load combinations during both high and low discharge season. The gate needs to be integrated with a dyke system, with a crest height of 4 m on the river facing side. On sea facing side, a dyke of 6.8 m with a vertical wall of 1 m on top has to be constructed to counter the impact of the storm surge. For the alternative without a barrier all dykes have to be heightened to 6.8 m with a vertical wall of 1 m on top to counter the impact of the storm surge.

4. Considering the effect of the hydro-morphodynamic changes on the social and economic activities, what decision in terms of closure of the estuary can be regarded as the best fitting solution for the system?

For determining the best fitting solution an MCA in combination with a cost evaluation was carried out. The criteria used in the MCA are: freshwater supply, agri- and aquaculture adaptation, biodiversity, stable riverbanks, and navigability. Assessing each criterion is done qualitatively. This qualitative approach leads to large uncertainties as can be read in Section 10. The outcome of the MCA is that there is no clear distinction between a partial closure or no closure in the Ham Luong estuary. A flood protection system with a partial closure is priced at around  $\leq 1.3$  billion, taking into account both initial cost and maintenance costs until 2100. The most cost-effective choice is a flood protection system without a partial closure, making it the best fitting solution.

To answer the main research question: "What is the impact of various closure alternatives on the hydraulic characteristics and social activities in the Ham Luong estuary, considering a 75-year forecast?" It is expected that with or without closure of the Ham Luong estuary the system will change. The availability of fresh water will be improved by the presence of a closure, although more research is needed to specify this further. The increasing salt intrusion, as a result of RSLR will lead to agricultural and aquacultural adaptation in all alternatives either due to the construction of the barrier or due to the gradual RSLR. A closure also has an effect on the biodiversity, stability of the river banks, and navigability in the river. When implementing a closure these effects should be further investigated to assess the effect quantitatively.

# 10 Discussion

This section addresses the most important uncertainties, assumptions, simplifications, and shortcomings of this project. Each of these are grouped under their corresponding subjects. The uncertainties, assumptions, simplifications, and limitations identified in this section should be taken into careful consideration when interpreting and/or reproducing the results presented in this report.

## 10.1 Limitation of the scope

In hindsight, the spatial scope of the project is too restricted. The hydro- and morphodynamic processes and social activities taking place in the Ham Luong estuary are dependent on processes taking place outside the Ham Luong estuary as well. For instance, the discharge flowing through the Ham Luong estuary is not only dependent on what is happening in the Ham Luong estuary, but is also subject to changes in the other branches. Moreover, whether the local farmers want to cultivate a certain product depends on both the ability of the product to withstand salinity and the global market price of that product. Of course, the larger the scope, the more accurately the processes can be represented. However, this does increase the complexity, and for that reason the scope was chosen as it is. As a consequence, many assumptions were made to account for these processes outside the scope of this project. These assumptions should be taken into consideration when interpreting the results.

An important consequence of the limited scope of the project is how this affects the model. The upstream bifurcations of the Mekong Delta are not included in the model and, as a consequence, the interaction between the discharges in the branches is not captured. As this interaction influences the distribution of the discharge, this influences the processes in the Ham Luong estuary. The effect of this choice of the scope on the discharge especially affects the processes upstream of a closed barrier, as this is a situation where a redistribution of discharge between the branches is expected. Consequently, it was not possible to draw any conclusions regarding the water level, salinity, and sedimentation in the simulations with a closed barrier.

## 10.2 Uncertainties regarding collection of data

Uncertainties are inevitable when collecting data, especially in an area such as the Ham Luong estuary. In some cases there is limited data available, and in other cases there is no data available at all. Also the reliability of the datasets should be noted, as many of the datasets contain missing values or were given in Vietnamese.

The largest uncertainties in the data collection of this project arise from the prediction of future conditions taking into account the effects climate change. The climate change scenario of RCP8.5 was assumed throughout this project. This climate change scenario carries a lot of uncertainty in itself, as it is based on the extrapolation of data over large periods of time and is dependent on the way society decides to take action. This climate change scenario was used to extrapolate upstream river discharge, RSLR, storm surge, and the wave climate. As these datasets were used as input for the model and the design, a change in these values could influence the outcomes of this report.

Another important shortcoming in the data collection that should be discussed, is the lack of data on the future effects of the construction of hydropower dams and sand mining activities on the discharge and sedimentation rates downstream. As little is known, it was decided not to include these effects in the model. However, it can greatly impact the outcome of the results in the future scenarios.

Additionally, another aspect to bear in mind in the collection of data of this report is the bathymetry. It was found that the dataset of the depth profile of the Ham Luong estuary, which was provided by Thuyloi University, shows some curiosities. Upon superimposing the bathymetry data on the area in Google Earth using a GIS tool, it was found that the depth (of the dataset) at the river banks was equal to several metres. This is not realistic, as one expects the water depth at the river banks to be nearly zero (or negative).

## 10.3 Model limitations

In setting up a model and interpreting the results, many choices and assumptions are made which are accompanied with uncertainties. The most important choices are discussed here. Many concessions were made in view of computational time. The first concession which was made regards the dimension of the model. A 2DH model was used, instead of a 3D model. It is expected that one reason for the underestimation of salinity in the model can be attributed to this, as 3D models are able to capture salinity processes more accurately. Another concession concerns the grid size: a relatively coarse grid was used. This could be another reason for the fact that the salt intrusion reaches less far landwards than was expected. In view of computational time, only five months were modelled in the longest simulation, which is little in terms of erosion and sedimentation modelling. Therefore, the influence of the presence of the structure on the surrounding bathymetry overlooking a 75-year period was not considered based on model results. Also, the islands were modelled as so-called dry points, but in reality their shapes and sizes can change over time, which is a process that has been excluded from the model. Furthermore, another important concession in terms of computational time which should be discussed is the fact that it is impossible to model every weather scenario. For this reason, a selection has been made. In this selection, it was opted to only model the situations which lead to the most critical situations. No analysis has been done which assesses the probability of occurrence of these critical situations. This should be kept in mind when reproducing the results. The last thing to bear in mind on the limitations of the model is the representation of the sediment processes. Only one value in terms of friction and uniform grain size was assumed for the entire estuary, while in reality this is not the case. More on the effects of changing these parameters can be found in Section 5.6.

## 10.4 Flood protection system

There are many choices and assumptions in the design of a complex flood protection system for a large area. One of the assumptions in the design is that the hydrodynamic and hydrostatic loads are calculated at only a critical location. The maximum wave height at location 1 is chosen for all designs, which is a conservative approach. Additionally, equal dyke heights along the river and maximum and minimum water depths are simplified based on limited data. Another remark is that the hydrostatic and hydrodynamic forces acting on the steel frame of the gates are evenly distributed while in reality this is a non-linear distribution. These concessions on the calculation of the hydrostatic and hydrodynamic loads have a large influence on the dimensions of the elements for the final design.

Furthermore, for the calculation of the flow past the piers the Morrison equation was used. It is applicable to slender structures and assumes cylindrical bodies, which is not identical to the shape of the pier. This method is mostly used in offshore structure engineering in an undisturbed flow. The tables used for offshore wind turbines to determine drag and inertia coefficients might not be applicable for a river. When a different approach is used the results for these flow forces could change and the design could therefore change as well.

In the MCA for the gate type the weights for the different criteria are determined by making assumptions about the importance relative to each other. Changing these weights would change the outcome of the MCA and therefore the type of gate that would be constructed. Furthermore, the score of each of the alternative gate types on the different criteria also has a human factor. Changing the way that the different gate types score on the criteria would also change the choice of gate type significantly.

Concerning the operation of the structure, in the event of high flow discharge from the river and a tropical storm at sea, it is possible to prevent seawater from infiltrating the land. However, the river will continue to carry a large discharge towards the sea. The river will probably not overflow the dyke in a time span of 1-2 days but more research into this situation is needed. If this situation will lead to flooding of the area by the river, it will affect the operation and design.
## 10.5 Evaluation

In Section 8 a lot of assumptions are made which can drastically change the preferred solution. First, it should be mentioned that each criterion is qualitatively assessed. For example, the criterion agricultural and aquaculture adaption is more a thought experiment than a factual outcome. One could also use a different rating system, such as a rating system in which each alternative is ranked relative to the other alternatives. This would definitely give another outcome of the MCA. Each criterion is assigned a weight based on literature research and personal judgement after talking with experts and some local people. Another person could give other weights to the criteria leading to a different outcome of the preferred solution.

The calculation of the cost per alternative is a very rough estimation. It should be noted that in this cost estimation only the Ham Luong River is taken into account. In reality, if one only raises the dykes along the Ham Luong River and not those further inland and along the other Mekong branches, it is likely that Ben Tre province still will not be adequately protected against flooding. For a more accurate cost estimation, and coming back to the earlier statement about the limitation of the scope of this project, it is recommended to consider all other Mekong branches as well.

# 11 Recommendations

This study can be used to create knowledge about the potential impact of a storm surge barrier on the area of the Ham Luong estuary. However, the results and assumptions should be be treated with caution, considering the project's limited scope, which is elaborated upon in Section 10. In this section, recommendations are given on how to improve a similar study in future projects.

## 11.1 Scope recommendations

An expansion of the area of interest is recommended. This expansion will help analyse the impact of a barrier on upstream regions and other branches. Especially in the case of a long closure of the barrier, these upstream effects are important to understand. Lastly, additional weather scenarios could also be taken into account in addition to the assumed extreme climate scenarios. Building with Nature is something that could also be implemented when using a larger scope. Companies like Royal HaskoningDHV and WWF already have made plans relating to this and are useful to contact for extra information.

The alternative without any intervention, meaning no heightening of the dykes, should be investigated as well. This can show the necessity of intervening in this area as soon as possible.

To create a more accurate model, more thorough data concerning discharge, soil, sediment transport, salinity, and detailed bathymetry is needed. These datasets should be compared and used for calibration and verification. It is recommended to verify the reliability of the sources and compare datasets from different sources to find uncertainties. Future consequences of the upstream hydropower dams and sand mining should be taken into account as well, depending on the availability of reliable data. However, for every extension of the model, the additional computation time should be considered carefully.

For additional structural calculations, the foundation of the storm surge barrier has not been designed but is an important element of the construction. Obtaining more recent and precise borehole data is necessary to achieve this. For a more detailed design the hydrostatic and hydrodynamic load should be calculated at multiple locations. The dyke should be designed in more detail, rather than a quick estimation.

## 11.2 Evaluation recommendations

In Section 8, the criteria of agricultural adaptation, biodiversity, and navigability are being considered. For a more precise and substantiated assessment of these criteria, experts with backgrounds in economics, biology, and ecology are necessary. In general, the involvement of additional experts is essential for a thorough impact evaluation of a partial closure in the Ham Luong estuary. The given timeframe of 10 weeks with only 6 students is not enough for this goal.

Freshwater supply has turned out to be one of the most important criteria. For this reason research about the local water management and future possibilities for freshwater supply should be done. This could give solutions other than river barriers, which would be very interesting for the local citizens and the government.

## 11.3 Recommendations for other Multidisciplinary Project groups

To reach better conclusions about the social impacts of the structure, it is necessary to interview more people. For this project, thirteen useful surveys were conducted in approximately three hours with the help of three Vietnamese interviewers; more about the survey can be read in Appendix G. In Vietnam, the plans change more than in the Netherlands. Therefore, it is important to make arrangements about the survey schedule and plans, but remain flexible. This flexibility will help you in many ways during your stay in Vietnam.

Schedule wise, it is also advised to schedule the field trip to your area of interest in the first three weeks. During the field trip, a lot of new insights were gained and these could have helped earlier in the project. When taking a trip in the Mekong Delta, it is useful to know that it is not allowed to cross the maritime border near the coast.

## References

- Adamson, P. T., Rutherfurd, I. D., Peel, M. C., & Conlan, I. A. (2009). The Hydrology of the Mekong River. In *The mekong* (pp. 53–76). Elsevier. doi: 10.1016/B978-0-12-374026-7.00004-8
- ADB. (2020). Resilience of Rural Development and Climate Change in Viet Nam Asian Development Bank. Retrieved from https://www.adb.org/documents/tper-resilience-rural-development-and -climate-change-in-vietnam
- Albrecht, D., Hocquard, H., & Papin, P. (2010). Urban development in Vietnam: the rise of the local authorities. (Tech. Rep.). AFD.
- Andrew-Wells Dang, Le Kim Thai, & Nguyen Tran Lam. (2015, 10). Increasing citizen participation in governance in Vietnam. Retrieved from https://vietnamlawmagazine.vn/increasing-citizen -participation-in-governance-in-vietnam-5047.html
- Battjes, J. A., & Labeur, R. J. (2017). Unsteady Flow in Open Channels. Cambridge University Press. doi: 10.1017/9781316576878
- Bosboom, J., & Stive, M. (2023). Coastal Dynamics (1.2 ed.). Delft: TU Delft Open.
- Brian Eric. (2006). Fish migration triggers in the Lower Mekong Basin and other freshwater tropical systems. Technical paper # 14. Mekong River Commission. 2006. Retrieved from https://www.researchgate .net/publication/259529261\_Fish\_migration\_triggers\_in\_the\_Lower\_Mekong\_Basin\_and\_other \_freshwater\_tropical\_systems\_Technical\_paper\_14\_Mekong\_River\_Commission\_2006
- CEN. (2002). Eurocode Basis of structural design (Tech. Rep.). Brussels: CEN.
- Cong San, D., Duc Thang, T., & Manh Hung, L. (2017, 3). EXISTING SHORELINE, SEA DYKE, AND SHORE PROTECTION WORKS IN THE LOWER MEKONG DELTA, VIETNAM AND ORIENTED SOLUTIONS FOR STABILITY. Retrieved from https://www.researchgate.net/publication/ 315625566\_EXISTING\_SHORELINE\_SEA\_DYKE\_AND\_SHORE\_PROTECTION\_WORKS\_IN\_THE\_LOWER\_MEKONG\_DELTA \_VIETNAM\_AND\_ORIENTED\_SOLUTIONS\_FOR\_STABILITY
- CruiseMapper. (n.d.). Ben Tre City (Vietnam) cruise port schedule CruiseMapper. Retrieved from https://www.cruisemapper.com/ports/ben-tre-city-port-14128
- Daniel, R., & Paulus, T. (2019, 1). Structural Analysis. Lock Gates and Other Closures in Hydraulic Projects, 461-530. Retrieved from https://linkinghub.elsevier.com/retrieve/pii/B9780128092644000070 doi: 10.1016/B978-0-12-809264-4.00007-0
- Deltares. (2003). Flooding: Tools for a preliminary assessment of coastal problems and solutions. Delft: Deltares.
- Deltares. (2023). Delft3D-FLOW User Manual (Tech. Rep.). Retrieved from https://www.deltares.nl
- De Ronde, J. (1990). ENVIRONMENTAL IMPACTS OF ENCLOSURE DAMS IN THE NETHERLANDS (Tech. Rep.). The Hague: Rijkswaterstaat.
- de Vet, P., van Prooijen, B., & Wang, Z. (2017, 3). The differences in morphological development between the intertidal flats of the Eastern and Western Scheldt. *Geomorphology*, 281, 31–42. doi: 10.1016/ j.geomorph.2016.12.031
- Food and Agriculture Organization of the United Nations. (2018). *Rice market monitor* (Tech. Rep.). Retrieved from https://www.fao.org/markets-and-trade/commodities/rice/en/
- Galloway, W. (1975). Process Framework for Describing the Morphologic and Stratigraphic Evolution of Deltaic Depositional Systems.
- GIZ. (2019). FINAL REPORT: Climate risk analysis and assessment report for Cai Lon Cai Be sluice gate project based on the PIEVC protocol (Tech. Rep.). Retrieved from https://pievc.ca/wp-content/ uploads/2021/05/2019\_GIZ\_CSI\_VN\_PIEVC\_Final\_Report\_EN.pdf
- Groenewold, S. A., Albers, T., Sorgenfrei, R., Ngoc Anh Vu, D., Thi Viet Phuong, N., Huu To, H., ... Trong Thinh, P. (2018, 6). COASTAL PROTECTION FOR THE MEKONG DELTA (CPMD) (Tech. Rep.). Retrieved from chrome-extension://efaidnbmnnnibpcajpcglclefindmkaj/http:// coastal-protection-mekongdelta.com/download/library/Coastal%20Protection%20for%20the% 20Mekong%20Delta%20(CPMD)\_EN.pdf
- Gugliotta, M., Saito, Y., Nguyen, V. L., Ta, T. K. O., Nakashima, R., Tamura, T., ... Yamamoto, S. (2017, 9). Process regime, salinity, morphological, and sedimentary trends along the fluvial to marine transition zone of the mixed-energy Mekong River delta, Vietnam. *Continental Shelf Research*, 147, 7–26. doi: 10.1016/j.csr.2017.03.001
- Hebbar, K., Santhosh, A., Sukumar, A. P., P., N., S.V., R., & V., S. (2021, 4). Effect of sea water substitution on growth, physiological and biochemical processes of coconut (Cocos nucifera L.) seedlings—A hydroponic study. *Scientia Horticulturae*, 280, 109935. doi: 10.1016/j.scienta.2021.109935

Hillen, M., Jonkman, S. N., & Stive, M. (2010). Coastal defence cost estimates; a case study of the Netherlands, Vietnam and New Orleans. (Tech. Rep.). Retrieved from www.hydraulicengineering.tudelft.nl

- Hoan Van Huan, & Nguyen Huu Nhan. (2006). RESULTS ON STUDY ABOUT WAVE FIELD OF DONG NAI-SAIGON ESTUARIES AND SUGGESTION OF SEA BANK AND RIVER MOUTHS PROTECTION METHODS (Tech. Rep.). Hanoi: SIWRR.
- Holthuijsen, L. (2007). Waves in oceanic and coastal waters.

Hoving, J., & van der Male, P. (2022). TU DELFT - CIEM4120-OE44097 2023 L04 - Hydrodynamic Loading.

- Institute of Coastal and Offshore Engineering. (2020). Current Status of Coastal Water Resources in Ben Tre Province (Tech. Rep.). Ho Chi Minh City: Institute of Coastal and Offshore Engineering.
- iStock. (2013). Foto's, afbeeldingen en stockfoto's van De stuw bij Driel in de Rijn. Retrieved from https:// www.istockphoto.com/vi/b%E1%BB%A9c-%E1%BA%A3nh/the-weir-at-driel-in-river-rhine
- Jonkman, S. N., Hillen, M. M., Nicholls, R. J., Kanning, W., & Van Ledden, M. (2013, 9). Costs of adapting coastal defences to sea-level rise - New estimates and their implications. *Journal of Coastal Research*, 29(5), 1212–1226. doi: 10.2112/JCOASTRES-D-12-00230.1
- Journée, J. M. J., & Massie, W. W. (2001). OFFSHORE HYDROMECHANICS First Edition (Tech. Rep.).
- Käkönen, M. (2008, 6). Mekong Delta at the Crossroads: More Control or Adaptation? *AMBIO*, 37(3), 205–212.
- Kaveney, B., Barrett-Lennard, E., Chau Minh, K., Dang Duy, M., Nguyen Thi, K. P., Kristiansen, P., ... Condon, J. (2023, 4). Inland dry season saline intrusion in the Vietnamese Mekong River Delta is driving the identification and implementation of alternative crops to rice. *Agricultural Systems*, 207, 103632. doi: 10.1016/j.agsy.2023.103632
- Korbee, D., Hong Quan, N., Hermans, L., & Ho Long, P. (2019, 7). Navigating the bureaucracy: an analysis of implementation feasibility for the Mekong Delta Plan, Vietnam. Journal of Environmental Planning and Management, 62(9), 1545–1561. doi: 10.1080/09640568.2019.1623014
- Kuenzer, C., Guo, H., Huth, J., Leinenkugel, P., Li, X., & Dech, S. (2013, 2). Flood Mapping and Flood Dynamics of the Mekong Delta: ENVISAT-ASAR-WSM Based Time Series Analyses. *Remote Sensing*, 5(2), 687–715. doi: 10.3390/rs5020687
- Kuijper, H., Bezuyen, K., Glerum, A., Kuiper, W., Voortman, H., & Vrijling, J. (1998, 1). Waterbouwkundige kunstwerken deel A Algemeen (Tech. Rep.).
- Kuijper H.K.T. (1998). CTwa3330 Waterbouwkundige kunstwerken (Tech. Rep.). TU Delft.
- Kummu, M., Lu, X. X., Rasphone, A., Sarkkula, J., & Koponen, J. (2008, 8). Riverbank changes along the Mekong River: Remote sensing detection in the Vientiane–Nong Khai area. *Quaternary International*, 186(1), 100–112. doi: 10.1016/J.QUAINT.2007.10.015
- Loc, H. H., Van Binh, D., Park, E., Shrestha, S., Dung, T. D., Son, V. H., ... Seijger, C. (2021, 2). Intensifying saline water intrusion and drought in the Mekong Delta: From physical evidence to policy outlooks. *Science of The Total Environment*, 757, 143919. doi: 10.1016/J.SCITOTENV.2020.143919
- McElwee, P., & M Horowitz, M. (1999, 1). Environment and Society in the Lower Mekong Basin: a Landscaping Review. Retrieved from https://www.academia.edu/1652011/Environment\_and\_Society\_in\_the\_Lower \_Mekong\_Basin\_a\_Landscaping\_Review
- Mekong River Commission. (2017, 6). Summary of the basin-wide assessments of climate change impacts on water and waterrelated resources in the Lower Mekong Basin (Tech. Rep.). Mekong River Commission.
- Mekong River Commission. (2023a, 10). Daily Discharge Kratie. Retrieved from https://portal.mrcmekong .org/time-series/chart?ts=17c0ffe74e224b50a9e3142fc404262b
- Mekong River Commission. (2023b). Mekong River Commission (MRC). Retrieved from https://www .mrcmekong.org/
- Meteoblue. (2023). Simulated historical climate & weather data for An Binh.
- Minderhoud, P., Erkens, G., Pham, V. H., Bui, V. T., Erban, L., Kooi, H., & Stouthamer, E. (2017, 6). Impacts of 25 years of groundwater extraction on subsidence in the Mekong delta, Vietnam. *Environmental Research Letters*, 12(6), 064006. doi: 10.1088/1748-9326/aa7146
- Ministry of Agriculture and Rural Development. (2016, 10). The preparatory survey for Ben Tre water management project (Tech. Rep.). Retrieved from https://www.jica.go.jp/Resource/ english/our\_work/social\_environmental/id/asia/southeast/vietnam/c8h0vm00009b2g33-att/ c8h0vm0000agn6p0.pdf
- Ministry of Agriculture and Rural Development, & Southern Institute for Water Resource Planning. (2021). QUY HOACH PHONG, CHONG THIEN TAI VE THUY LOI THOI Y 2021-2030, TAM NHIN DEN NAM 2050 (Tech. Rep.). Hanoi.

Molenaar, W., & Voorendt, M. (2022). *Hydraulic Structures General lecture notes Edition 2022* (Tech. Rep.). Molenaar W.F. (2011, 3). *CIE3330 Hydraulic Structures* (Tech. Rep.). TU Delft.

- Mooyaart, L., & Jonkman, S. (2017, 7). Overview and Design Considerations of Storm Surge Barriers. Journal of Waterway, Port, Coastal, and Ocean Engineering, 143(4). doi: 10.1061/(asce)ww.1943-5460.0000383
- Mooyaart, L., Jonkman, S., De Vries, P., Van der Toorn, A., & Van Ledden, M. (2014, 10). STORM SURGE BARRIER: OVERVIEW AND DESIGN CONSIDERATIONS. *Coastal Engineering Proceedings*, 1(34), 45. doi: 10.9753/ICCE.V34.STRUCTURES.45
- MRC. (n.d.). Hydropower. Retrieved from https://www.mrcmekong.org/our-work/topics/hydropower/
- Netherlands Advisory Team of the Mekong Delta Plan. (2013). Mekong Delta Plan Long-term vision and strategy for a safe, prosperous and sustainable delta (Tech. Rep.). The Hague: Dutch Government.
- Nguyen, A. D. (2008). Salt Intrusion, Tides and Mixing in Multi-channel Estuaries (Tech. Rep.). Delft: UNSESCO IHE.
- Nguyen Hoang Phuong. (2019). Advantages And Challenges For Tourism In The Mekong Delta, Vietnam. *INTERNATIONAL JOURNAL OF SCIENTIFIC & TECHNOLOGY RESEARCH*, 8(9). Retrieved from https://www.ijstr.org/final-print/sep2019/Advantages-And-Challenges-For-Tourism-In -The-Mekong-Delta-Vietnam-.pdf
- Nguyen Tho Sao. (2008). Storm surge predictions for Vietnam coast by Delft3D model using results from RAMS model (Tech. Rep.). Hanoi: Hanoi University of Science.
- Nguyen Trug Viet, Hitoshi Tanaka, Nguyen Phuong Mau, & Nguyen Chien. (2006, 8). INTERACTION BE-TWEEN MORPHOLOGY CHANGE AND SALINITY DISTRIBUTION IN THE DINH AN ESTUARY, VIETNAM (Tech. Rep.). Hanoi, Vietnam: Vietnam-Japan Estuary Workshop.
- NPR. (2009). Billion-Dollar Floodgates Might Not Save Venice. Retrieved from https://www.npr.org/ templates/story/story.php?storyId=112995748
- Pawlowicz, R., Beardsley, B., & Lentz, S. (2002). Classical Tidal Harmonic Analysis with Error Analysis in MATLAB using T\_TIDE (Tech. Rep.). Vancouver: University of British Colombia.
- Piman, T., & Shrestha, M. (2017). Stockholm Environment Institute, Project Report 2017-03 Case study on sediment in the Mekong River Basin: Current state and future trends (Tech. Rep.). Retrieved from https://mediamanager.sei.org/documents/Publications/Bangkok/SEI\_2017\_Report \_Mekong\_sediment\_LoRes.pdf
- Sakleshpur, V. A., & Madhav, M. (2013, 6). Embankments on Soft Ground: An Overview. Binh Duong. Retrieved from https://www.researchgate.net/publication/306107953
- Sánchez-Juny, M., Triadú, A., Andreu, A., & Bladé, E. (2019, 12). Hydrodynamic Determination of the Kinematic Viscosity of Waste Brines. ACS Omega, 4(25), 20987. Retrieved from /pmc/articles/PMC6921273//pmc/articles/PMC6921273/?report=abstracthttps://www.ncbi .nlm.nih.gov/pmc/articles/PMC6921273/ doi: 10.1021/ACSOMEGA.9B02164
- Savenije, H. (2012). SALINITY AND TIDES IN ALLUVIAL ESTUARIES (2nd ed.). Delft: TU Delft.
- Schiereck, G., & Verhagen, H. (2012). Introduction to Bed, bank and shore protection (2nd ed.). Delft: VSSD Delft.
- Shibasaki, R., Shimada, T., & Suzuki, M. (2021, 1). Greater Mekong Subregion: Is the Mekong River shipping competitive with other modes? Global Logistics Network Modelling and Policy: Quantification and Analysis for International Freight, 201-221. Retrieved from https://www.sciencedirect.com/science/ article/pii/B9780128140604000113 doi: 10.1016/B978-0-12-814060-4.00011-3
- Ta Quynh Hoa, Nguyen Quang Minh, Nguyen Cao Lanh, & Nguyen Hai Ninh. (2019). Community Participation in Urban Planning in Vietnam towards Sustainable Development - Prospects and Challenges (Tech. Rep.). Hanoi: National University of Civil Engineering. Retrieved from https://repository.vnu.edu.vn/ bitstream/VNU\_123/136826/1/KY\_20211109235526.pdf
- The Asia-Pacific Journal. (2010, 1). The Mekong River Under Threat The Asia-Pacific Journal: Japan Focus. Retrieved from https://apjjf.org/-Milton-Osborne/3286/article.html
- Thuyloi University. (2020). Typhoon: Date and winds.
- Toan, T. Q. (2014, 1). Climate Change and Sea Level Rise in the Mekong Delta: Flood, Tidal Inundation, Salinity Intrusion, and Irrigation Adaptation Methods. Coastal Disasters and Climate Change in Vietnam: Engineering and Planning Perspectives, 199–218. doi: 10.1016/B978-0-12-800007-6.00009-5
- Trading Economics. (2023). Vietnam Inflation Rate. Retrieved from https://tradingeconomics.com/ vietnam/inflation-cpi#calendar-table
- Tran Thuc, Nguyen Van Thang, Huynh Thi Lan Huong, Mai Van Khiem, Nguyen Xuan Hien, & Doan Ha Phong. (2016). CLIMATE CHANGE AND SEA LEVEL RISE SCENARIOS FOR VIET NAM (Tech. Rep.). Hanoi: Ministry of Natural Resources and Environment.
- Tromp, R. (2013). Study on the design of barriers in the Tien River estuary and the upstream effects due to closure (Unpublished doctoral dissertation). TU Delft, Delft.

- Truong Hong Son. (2012). INNOVATIVE STRUCTURE SOLUTION FOR DISCHARGE SLUICE AT VUNG TAU GO CONG VIET NAM (Unpublished doctoral dissertation). TU Delft, Delft.
- Truong Hong, S. (2012, 6). INNOVATIVE STRUCTURE SOLUTION FOR DISCHARGE SLUICE AT VUNG TAU GO CONG VIET NAM (Tech. Rep.). Delft: Delft University of Technology.
- Tuoi Tre News. (2022). \$144mn irrigation project inaugurated in Vietnam's Mekong Delta. Retrieved from https://tuoitrenews.vn/news/society/20220306/144mn-irrigation-project-inaugurated-in -vietnams-mekong-delta/66042.html
- University of Florida. (2023). *Habitat Requirements South Florida Aquatic Environments*. Retrieved from https://www.floridamuseum.ufl.edu/southflorida/habitats/mangroves/requirements/
- USACE. (n.d.). Harlan County Dam Tainter Gate Repair. Retrieved from https://www.nwk.usace.army.mil/ Media/Images/igphoto/2001533096/
- USAID. (n.d.). Fact Sheets Vietnam U.S. Agency for International Development. Retrieved from https://www.usaid.gov/vietnam/fact-sheets
- Van der Meer, J., Allsop, N., Bruce, T., De Rouck, J., Kortenhaus, A., Schüttrumpf, H., ... Zanuttigh, B. (2018). Manual on wave overtopping of sea defences and related structures. An overtopping manual largely based on European research, but for worldwide application (Tech. Rep.). EurOtop. Retrieved from http://www.overtopping-manual.com/assets/downloads/EurOtop\_II\_2018\_Final\_version.pdf
- Van Der Vegt, H., Storms, J. E., & Walstra, D. J. (2014). Establishing 3D numerical reservoir analoguesmodelling the formation of sand bodies in deltaic environments. 76th European Association of Geoscientists and Engineers Conference and Exhibition 2014: Experience the Energy - Incorporating SPE EUROPEC 2014, 812–816. doi: 10.3997/2214-4609.20141501
- van Hamel, A. (2018). PARTICIPATORY MONITORING MYANMAR A research on the motivation of people in Myanmar to participate in 'Participatory Monitoring' by making use of Q methodology (Tech. Rep.). Delft: Delft University of Technology. Retrieved from https://repository.tudelft.nl/islandora/ object/uuid%3A04f3202e-1212-4fdd-9616-1b754aa6fac6
- van Ledden, M., Tran Thanh, T., Huy Nguyen, D., & Thanh Nguyen, L. (2020, 8). risk: An assessment of the coastal protection system in Vietnam (Tech. Rep.). Hanoi: World bank group. Retrieved from http://www.worldbank.org/prwp.
- Van Rijn, L. (n.d.). Sedimentation basin Van Rijn.
- VESDEC. (2020). The natural environment of The Mekong Delta INSTITUTE FOR ENVIRONMEN-TAL SCIENCE AND DEVELOPMENT (VESDEC). Retrieved from http://vesdec.com.vn/ index.php?language=en&nv=news&op=VESDEC-activities/The-natural-environment-of-The -Mekong-Delta-120
- Vietnam Briefing. (2020). Investing in Ben Tre Province: What Investors Need to Know Vietnam Guide — Doing Business in Vietnam. Retrieved from https://www.vietnam-briefing.com/ doing-business-guide/vietnam/where-to-invest/investing-in-ben-tre-province-what -investors-need-to-know
- Vietnam Energy. (2019). Soc Trang attractive to wind power developers Tap chi Nang luong Viet Nam. Retrieved from https://vietnamenergy.vn/soc-trang-attractive-to-wind-power-developers-23037 .html
- Vietnam Inland Waterways. (n.d.). PROJECT ON UPGRADING HAM LUONG RIVER BEN TRE (Tech. Rep.). Retrieved from https://m.viwa.gov.vn/documents/22307/29606/Project+on+upgrading+Ham+ Luong+river.pdf/0f802610-7fcb-4cd8-a749-597f4716c723
- VietnamPlus. (n.d.). Ben Tre promoting tourism. Retrieved from https://en.vietnamplus.vn/ben-tre
  -promoting-tourism-advantages/245267.vnp
- VNCOLD. (2014). Cong dap Ba Lai Hoi dap lon va phat trien nguon nuoc Viet Nam. Retrieved from http://www.vncold.vn/Web/Content.aspx?distid=237
- Voorendt, M. (2014). File:Hartelkering 2.JPG. Retrieved from https://commons.wikimedia.org/wiki/File: Hartelkering\_2.JPG
- Voorendt, M. (2022). Manual Hydraulic Structures edition 2022.
- Watersnoodmuseum. (n.d.-a). *Hartelkering*. Retrieved from https://watersnoodmuseum.nl/kennisbank/ hartelkering-2/
- Watersnoodmuseum. (n.d.-b). *Maeslantkering.* Retrieved from https://watersnoodmuseum.nl/kennisbank/ maeslantkering-3/
- Woillez, M. N., & Espagne, E. (2022). The Mekong Delta Emergency, Climate and Environmental Adaptation Strategies to 2050. Final Report GEMMES Viet Nam project. (Tech. Rep.). Paris: Agence Francaise de Développement.

- World Bank Group, & Asian Development Bank. (2021). Vietnam Climate Risk Country Profile (Tech. Rep.). Retrieved from www.worldbank.org
- WorldData.info. (n.d.). Climate and temperature development in Vietnam. Retrieved from https://www .worlddata.info/asia/vietnam/climate.php
- WWF. (n.d.-a). *DFCD Mekong Delta WWF*. Retrieved from https://vietnam.panda.org/en/our\_work/ sustainable\_finance/dfcd/dfcd\_mekong\_delta/
- WWF. (n.d.-b). *Mekong Delta WWF Asiapacific*. Retrieved from https://asiapacific.panda.org/ priority\_places/mekong\_delta/
- Xuan Hien Nguyen, Van Uu Dinh, Van Khiem Mai, Van Tra Tran, & Van Tien Pham. (2017). Impacts of climate change on wave regimes in the east sea. Vietnam Journal of Sience, Technology and Engineering, 59(1).
- Xuan Quang, N., Vanreusel, A., Xuan Dong, N., & Thi Thu Trang, N. (2017, 1). INITIAL STUDY ON IMPACT OF BA LAI DAM CONSTRUCTION TO HUMANITY ECOSYSTEM IN BINH DAI DISTRICT, BEN TRE PROVINCE.. Retrieved from https://www.researchgate.net/publication/ 319001153\_INITIAL\_STUDY\_ON\_IMPACT\_OF\_BA\_LAI\_DAM\_CONSTRUCTION\_TO\_HUMANITY\_ECOSYSTEM\_IN\_BINH \_DAI\_DISTRICT\_BEN\_TRE\_PROVINCE
- ZJA. (2023). Ramspol Inflatable Surge Barrier, Kampen. Retrieved from https://www.zja.nl/en/Ramspol -inflatable-dam-Kampen

# Appendices

# Appendix A Field trip

The area of Ben Tre was visited by the MDP group to verify the system description and to see if some parts were missing. Also, a survey was conducted among several inhabitants in the area. This is described in Appendix G. During the field trip, the city of Ben Tre was visited and the agricultural areas in the province along the river and the coast. This was done by riding motorbikes. Furthermore, a river expedition with a boat was set up to investigate the river banks and river use of the Ham Luong estuary. An overview in photos can be seen in Figure A.1.

Ben Tre City is the biggest city in the area and has several markets and numerous small businesses. During the visit, the city seemed to be relatively calm and not busy. From Ben Tre City, tourist boat tours were also organised but these did not go very far along the river.

In the rural areas of the province, the main production seemed to be coconut, see Figure A.1d. The coconut industry was all around the area. Besides the coconut farms, a lot of coconut storage and trucks loaded with coconuts were observed. Shrimp production was also observed and can be seen in Figure A.1f, but less than coconut production. Furthermore, there were some paddy fields but those observed were not very large in numbers and size. Adaptation of the area was also observed; in Figure A.1e newly planted coconut trees can be seen. The hypothesis is that these coconut trees were planted on former rice fields. However, this was not confirmed.

A couple of smaller sluice gates along the river banks were visited. These can be seen in Figure A.1g and Figure A.1h. These gates seemed to be operational and were seen closed during the visit.

During the boat tour, the mangroves along the river embankments were very clearly visible. They can be seen in Figure A.1a with next to them a couple of fishing boats. The mangroves appeared to be everywhere along the embankments and new mangroves were growing as well, see Figure A.1b.

Fishing boats were not only observed along the river embankments but also in small harbours and in side channels. The sizes of the boats were not larger than the ones depicted in Figure A.1a and Figure A.1c, but could be heavily loaded with for example tree trunks.



(a) Boats and mangroves



(b) Mangrove growth



(c) Fishing boats



(d) Coconut field



(e) Coconut field adaptation



(f) Shrimp farm



(g) Sluice along the river bank  $\left( \begin{array}{c} \mathbf{g} \end{array} \right)$ 



(h) Small sluice along the river bank

Figure A.1: Field trip photos

# Appendix B Data collection

This Appendix focuses on the collection of the relevant data, which is needed as input for the Delft3D model and the design of the flood protection system.

## B.1 Bathymetry

Thuyloi University (Hanoi, Vietnam) provided a dataset containing depth measurements at specific locations in the Ham Luong estuary. These depths were measured in 2022. The bathymetry, which can be seen in Figure B.1, was created by triangular interpolation of the provided depths using Delft3D. Particularly large depths can be observed at multiple locations in the estuary. These depths can be attributed to sand mining processes in addition to natural river and estuary processes.

This bathymetry is used as input for the Delft3D model. Note that the heights of the islands are not wellrepresented in this bathymetry due to the fact that there were no depths measured at the islands, for this reason the interpolation results in a too low elevation at the islands. In order to account for these low elevations in Delft3D, the grid cells corresponding to the locations of the islands were marked as "dry points".



Figure B.1: Bathymetry Ham Luong estuary, recorded in 2022

#### B.2 Upstream river discharge

In Figure B.2, the daily average discharge entering the Ham Luong River during 2019 and 2020 is plotted in blue and orange, respectively. This data was retrieved from Thuyloi University. The original data consists of discharge measurements at My Thuan, located upstream of the Ham Luong estuary. Downstream of My Thuan, the Tien River splits into five branches. SIWRP uses a fraction of 15% of the total discharge of the Mekong, measured in Kratie (Cambodia) that flows into the Ham Luong estuary. This assumption is also used in this report. As the discharge in My Thuan is still affected by the tide, the daily average discharge is plotted.



Figure B.2: Daily average discharge in 1982, 1998, 2019 and 2020 in the Ham Luong River

Although a lot of data about the discharge of the Mekong is available, the amount of future predictions is limited and shows a big scatter in prediction. In a summary of the Mekong River Commission (MRC) about the climate change impacts on the Mekong River basin, the projected discharge magnitudes and relative changes in Kratie are used to make an estimation on the predicted future discharge regime of the Mekong. These are predictions for the year 2060 and are based on the fifth assessment of IPCC (Assessment Report (AR)5). It includes different scenarios that are based on the emission scenarios of IPCC (RCP2.6, RCP4.5 and RCP8.5) (Mekong River Commission, 2023b). As stated in Section 4.4, the Vietnamese government advises to use the RCP8.5 scenario, which corresponds to the most extreme climate change scenario of the summary by MRC (Mekong River Commission, 2017). Therefore, the most extreme values for discharge were chosen. This results in an extreme low discharge during dry season and an extreme high discharge in wet season. The relevant numbers are stated in Table B.1.

Table B.1: Climate change predictions on the discharge of the Mekong River (Mekong River Commission, 2017)

	Minimum prediction	Maximum prediction	Extreme condition 2060
Peak flow wet season	-26%	+44%	+44%
Peak flow dry season	-43%	+20%	-43%

It is assumed that the discharge distribution into the different branches of the delta remains constant in the future. This makes it possible to use existing discharge data of the Ham Luong estuary and extrapolate this, using the predicted conditions of the future discharge regime in Table B.1. In this case, the most extreme values of the years 2019 and 2020 in Figure B.2 are taken, considering high and low discharge season. Combining these values with the climate change predictions results in the expected daily average discharge shown as the red dashed line below in Figure B.3.



Figure B.3: Discharge time series in the Ham Luong estuary. Blue: discharge in 2019 based on My Thuan discharge. Orange: discharge in 2020 based on My Thuan discharge. Red: Extreme scenario in 2100 based on 2019 and 2020 data. Green: Input discharge data for the model.

In Figure B.3, different time series are plotted. In order to find representative discharge values to use as input data in the model, different data was analysed. First of all, daily averaged discharge data in My Thuan is used to estimate extreme discharge (low following dry season, high following wet season). These are the blue and orange line in Figure B.2 and Figure B.3. Secondly, more years of discharge data was consulted using the dataportal of the Mekong River Commission (Mekong River Commission, 2023a). In this case, daily discharge in Kratie was used to obtain the discharge in the Ham Luong estuary, using the percentage of 15%. This was done, in order to have data without the tidal influence because it is outside the tidal influence due to its proximity to the coast. Comparing the daily discharge in Kratie, including future extreme conditions, showed that the magnitudes of the extreme scenario based on 2019 and 2020 (red line in Figure B.3) is in the right order. However, when looking at earlier years of discharge data, it becomes evident that the high discharge peaks in 2019 and 2020 do start significantly later in the year. Therefore, the discharge peak of the chosen simplified discharge time series for the model starts earlier compared to 2019 and 2020 (green line in Figure B.3). As these discharge peaks differ from year to year, the highest overall discharge was used as input for the simulations with the storm in the high discharge season.

#### B.3 Tide

In order to identify the tidal regime at the Ham Luong estuary, a one-year water level time series of 2021 at An Thuan was used, which is located in the mouth of the Ham Luong estuary. This data was provided by the ICOE. This time series is analysed using the T-Tide analysis in Python. The goal of T-Tide is to identify the different tidal constituents in a time series, including the corresponding amplitude and phase (Pawlowicz et al., 2002). These values can be used as input for the Delft3D model. In Table B.2, the most relevant tidal constituents are stated. In Figure B.4, two days and one month of tidal signals are shown. It can be seen that the tide shows two high waters in one day, which verifies that the tidal regime is indeed "mixed, more semidiurnal", as stated in Section 3.2.3. In the monthly tidal signal, the spring-neap tidal cycle can be seen. Analysing the data gives a maximum high tide of +138 cm and a minimum low tide of -301 cm, which makes the maximal tidal range 439 cm.

Constituent	Amplitude (cm)	Phase (°)
M2	77.7	170.7
K1	52.1	315.8
01	35.2	37.1
S2	27.6	102.9
SA	27.0	28.89
P1	17.4	321.8
N2	15.4	2.49
K2	8.6	90.5
SSA	6.5	97.6
Q1	5.9	237.0
L2	4.3	335.8

Table B.2: Most relevant tidal constituents



Figure B.4: Tidal signals

### B.4 Relative sea-level rise

As stated in Section 4.4, the MONRE recommends to design for the IPCC scenario RCP8.5. In the corresponding report on climate change impacts in Vietnam, the predicted absolute SLR is also presented. For the region of of the Mekong Delta, an absolute SLR of 73 cm between 2005 and 2100 is predicted (Tran Thuc et al., 2016).

Besides absolute SLR, the significant amount of land subsidence makes it necessary to look into RSLR. In the report of MONRE, no values for land subsidence are given. In other research, a significant difference between urban and rural areas can be observed. For rural areas, the land subsidence between 1991 and 2015 has been 1.5 cm/year (Minderhoud et al., 2017). The main driver of this observed subsidence is groundwater extraction in the period of 1991 to 2015. As it is difficult to predict the groundwater extraction in the next centuries, in this report no significant change in groundwater extraction is assumed, thus a rate of 2 cm/year is used as subsidence rate. Combining absolute SLR and land subsidence into RSLR, results in a RSLR of 156 cm between 2005 and 2100.

#### B.5 Storm surge

In order to estimate the expected storm surge water elevation at the coastline, the tool Flooding, developed by Deltares, was used (Deltares, 2003). This tool takes wind speed, depth, fetch and properties of the water as input. In order to find a good estimation of the storm surge level at the mouth of the Ham Luong estuary, a set of assumptions were made:

- 1. The tool is used for a water depth of 30 m, which is the depth of the South Chinese Sea at approximately 100 km distance from the coastline.
- 2. The occurrence of storm surges is varying strongly along the coastline of Vietnam. The Mekong Delta is known as a coastline with a low risk of storm surges. In order to find the design wind speeds in the Mekong Delta, a document provided by the ICOE was used (Institute of Coastal and Offshore Engineering, 2020). This document states that all coastal dykes in the region of Ben Tre are built according to a new criteria to ensure resistance against wind speeds up to level 9 Beaufort. Considering the importance of a potential storm surge barrier, and increasing value of the hinterland, it is decided to design for wind speed up to level 10 Beaufort. This corresponds to a wind speed of 28.4 m/s.
- 3. Since typhoons are circular formed storms that travel along a certain path, the exact fetch is difficult to define. A simplification, used by Truong Hong Son (2012), defines the fetch length as 1.4 times the radius of typhoon. This simplification takes into account that the wind is not acting on a straight line due to the circular form of the typhoon and will be used for the estimation of the surge level.

Estimating the radius of a typhoon, and the corresponding fetch length, is a difficult task. Based on historical

data of typhoons in Vietnam a storm radius of 80 nautical mile will be used (Thuyloi University, 2020). This corresponds with a fetch length of 112 nautical mile (208 km). The combination of this radius with wind speeds of up to 10 Beaufort have been observed in recent years, which makes these assumptions a valid reference storm. Using the mentioned tool by Deltares, leads to a storm surge level of +2.18 m.

To validate this number, some historical observations and future predictions of storm surge elevations along the coastline were used. An expected maximum surge level of 2.4 m is stated by Nguyen Tho Sao (2008). In the report of the MONRE, a possible maximum storm surge of 2.7 m at the coastline of the eastern Mekong Delta is mentioned (Tran Thuc et al., 2016). These values show a slight higher storm surge level than the calculated level. As the exact background of the two literature values is unknown, it is decided to use the estimated value of +2.18 m.

#### **B.6** Sediment characteristics

The main soils in the Mekong Delta are the following with corresponding grain size distributions (Piman & Shrestha, 2017):

- Gravel > 2.0 mm
- $\bullet~$  Coarse s and 0.5-2.0 mm
- $\bullet\,$  Medium s and 0.25-0.5 mm
- $\bullet~{\rm Fine~sand}~0.063\text{-}0.25~{\rm mm}$
- Silt and clay < 0.063 mm

The research from Piman and Shrestha (2017) shows that more than half of the sediment, about 60%, in the Lower Mekong Delta (LMD) consists of fine sand, 25% consists of silt and the rest consists of medium sand. There is almost no coarse sand or gravel in the area. For the sake of simplicity, a uniform grain size is chosen. This choice is based on the research of Tromp (2013). In this research, an average  $D_{50}$  of 200 µm has proven to be a good estimate, thus this assumption has been adopted in this project.

Suspended Sediment Concentration Data was retrieved by ICOE through measurements performed at the Hung Phong station, in the middle of the Ham Luong estuary. These hourly measurements were done in 2020 during a day in October, which is during high flow. The average of these measurements is 138.6 mg/L suspended sediment. This suspended sediment consists of cohesive mud and non-cohesive sand in approximately 85% and 15%, respectively. These percentages and the differences between high and low discharge originate from data of the Mississippi River and have been used by other researchers in the Mekong Delta, because these rivers seem to be similar concerning sediment transport (Tromp, 2013). This results in estimated suspended sediment concentrations during high discharge and low discharge which are shown in Table B.3.

Table B.3: Suspended sediment concentrations

Sediment concentration in $kg/m^3$	Mud	Sand
High discharge	0.1178	0.0208
Low discharge	0.0094	0.0000

For the sediment transport of these concentrations, default formulas are used. For sand this is the transport formula defined by Van Rijn (2007), including bedload and suspended load. For mud the transport formula of Partheniades and Krone is used, only including suspended load (Deltares, 2023).

Lack of data and measurements in the Ham Luong estuary is the reason for the simplification of the values and the choice for default sediment transport formulas. This should not be done if detailed results for sedimentation and erosion are desired outcomes of a model. In the case of this report, the differences between alternatives can be shown by using the same simplified input and default formulas and the details are not of very high importance.

#### B.7 Wind conditions

The wind conditions in the coastal zone show clear monsoon characteristics. During the wet season, the SW monsoon occurs, and during dry season the NE monsoon occurs, with their corresponding dominant wind directions. In Figure B.5a the two dominant wind directions can be seen. The wind speed distribution throughout

the year in Figure B.5b shows that the NE-winds during dry season have higher wind speeds with velocities up to approximately 30 km/h. The average wind speed at the Vietnamese coast is 6 m/s (Vietnam Energy, 2019).



Figure B.5: Wind conditions near Ham Luong River mouth (Meteoblue, 2023)

In exceptional situations, the highest wind speeds occur during tropical storms like typhoons. These winds can reach gust speeds up to 40 m/s, according to historical data of typhoons along the coastline of Vietnam (Thuyloi University, 2020). The effect of these wind speeds on the water level will be discussed in Appendix B.5.

## B.8 Wave climate

The paper of Hoan Van Huan and Nguyen Huu Nhan (2006) refers to wave data measurements that were conducted between 1986 and 2006 at Bach Ho, an offshore wave station close to the coast of the Ham Luong estuary. The data can be found in Table B.4. These measurements were combined with data from nearby offshore oil platforms of Vietsovpetro and the significant wave heights and corresponding wave periods were determined for different return periods. Only the wave data originating from the NE and SW direction are taken into account in this report, as these are the result of the most frequent and most strong winds.

Table B.4:	Wave climate data	measured at	Bach 1	Ho station	from	1986-2006	(Hoan	Van	Huan	& Nguyen	Huu	Nhan,
2006)												

Roturn	NE		SW			
period	Significant wave height (m)	Wave period (s)	Significant wave height (m)	Wave period (s)		
1 year	3.5	8.1	3.0	7.9		
10 years	4.5	8.7	4.1	8.1		
25 years	5.5	9.2	4.4	8.5		
50 years	6.4	9.5	4.8	8.6		
100 years	7.2	9.7	5.5	8.7		

The current averaged offshore significant wave height in January is 1.97 metre with a period of 7.72 seconds and in July 0.72 metres with a period 4.93 seconds. The expected averaged off-shore significant wave height for 2100 for the RCP 4.5 scenario in January is 1.84 metre with a period of 7.69 seconds and in July 0.77 metres with a period 4.94 seconds. As stated before in this report the used climate scenario RCP 8.5, but due to the lack of data of this RCP scenario above values are used (Xuan Hien Nguyen, Van Uu Dinh, Van Khiem Mai, Van Tra Tran, & Van Tien Pham, 2017).

#### B.9 Salinity levels for agriculture

Salt intrusion is likely to increase in the future in Ben Tre Province, as stated in Section 1.1. In order to incorporate the effect of salt intrusion in this analysis, the acceptable salinity for different kind of crops are used. Although no exact value for a critical salinity for rice crops is known, some research has shown acceptable values that can be used as an estimation. Salt tolerant rice shows significant yield losses at a salinity of 4 ppt (Kaveney et al., 2023). Coconut trees can withstand salinity levels up to 50 ppt, which is higher than that of seawater (Hebbar et al., 2021) and shrimp have a best fit range of salinity level of 18-25 ppt (Ministry of Agriculture and Rural Development, 2016).

## Appendix C Simplified cross-section

The upstream boundary of the model is located around 80 riverkilometres (RKM) upstream of the mouth. At this location, the tidal influence is still affecting the discharge regime in the river. In order to create a grid of which the upstream boundary, including the inflow of discharge, is not affected by the tidal regime, the grid needs to be extended. In order to estimate the river bathymetry of this extended upstream reach of the river, the cross-section is described by an exponential function, as was proposed by Savenije (2012). This methods assumes a rectangular cross-section and that the cross-sectional area, river width and river depth can be described as:

$$A = A_0 \cdot \exp\left(-\frac{x}{a}\right) \tag{3}$$

$$B = B_0 \cdot \exp\left(-\frac{x}{b}\right) \tag{4}$$

$$h = h_0 \cdot \exp\left(\frac{x(a-b)}{ab}\right) \tag{5}$$

In which:

- A =Cross-sectional area  $[m^2]$
- $A_0 =$ Cross-sectional area of the mouth  $[m^2]$
- a =Cross-sectional convergence length parameter [m]
- B = River width [m]
- $B_0 =$ River width of the mouth [m]
- b = Width convergence length parameter [m]
- h = Depth[m]
- $h_0 = \text{Depth at river mouth } [m]$

In order to find the fitted values for the coefficients a and b, the bathymetry, received from Thuyloi University, was used. In total 11 cross-sections where taken between the upstream boundary and the mouth. The most downstream cross-section is defined as the river mouth, with a cross-section of 20,565 m<sup>2</sup> and a width of 2.64 km. The remaining 10 cross-sections where taken at sections where no islands are present, since the calculations can not include the islands.



Figure C.1: Exponential functions, fitted on the bathymetry data from the model

When looking at Figure C.1a, the graphs look like a good fit on a logarithmic scale. In Figure C.1b, Figure C.1c and Figure C.1d, a bigger scatter is observed. This is not unexpected, since the function is fitted based on only 11 cross-sections. However, in general it can be stated that the exponential function as defined by Savenije (2012), forms an acceptable fit to the received dataset.

For the extended upstream reach, it was decided to determine the cross-sectional area of the river on 10 locations. In total, a distance of additional 130 km is added to the model. By using the exponential function in Figure C.1b, the cross-sectional area of the upstream locations is determined.

The exponential increase in water depth in Figure C.1d is the result of the assumption that the cross-sectional area and width can be estimated for upstream reaches, and an assumed rectangular cross-section. The result of these assumptions is that numerically, an exponential increase in water depth is expected. However, this is not expected in reality. For the extended upstream reach, the average water depth of the most upward bathymetry section is taken as water depth. The determined cross-section and this constant depth leads to a upward decrease of river width.

The results that were obtained in this analysis were verified using previous research. In the thesis of Nguyen (2008), the the same method is used on the Ham Luong estuary. However, since the used bathymetry data is unknown, a new analysis has been done. The resulting values from Nguyen (2008) can be used as verification values. In Table C.1, the results of both analysis are stated. It can be seen that the assumptions related to the river mouth are slightly different. Especially the cross-sectional area that results from the bathymetry data is significant bigger than the area that was assumed by Nguyen. This difference leads to a different value for the *a*-parameter. The *b*-parameter, which is related to the river width, shows a good match between the two results. These values show that the result of the analysis, based on the bathymetry data, is close to literature values. In order to be consistent within the model, the results that are obtained with the received bathymetry data will be used to for the upstream reach.

	Analysis results	Verification values (Nguyen, 2008)
a	71	55
b	54	55
$A_0 (\mathrm{m}^2)$	20,565	17,000
$B_0$ (m)	2,640	2,800
$h_0$ (m)	7.7	6.1

Table C.1: Results and verification cross-section analysis

# Appendix D Sediment Tool

In Table D.1 below, the input values of the sedimentation tool are stated. This tool was used to estimate the sedimentation layer in front of a closed barrier and the mud concentration in this basin in front of the barrier. The bold numbers are location specific values: tidal range and area of basin. The italic values are default values that are not changed and are expected to match this case. The remaining number were defined for the Ham Luong estuary, but do not vary for the different barrier locations.

Table D.1:	Input for	$\operatorname{sedimentation}$	$\operatorname{tool},$	calculating	the m	ud	$\operatorname{concentration}$	and	$\operatorname{sedimentation}$	$\inf$ ront	of	$_{\rm the}$	${\rm closed}$
barriers.													

Parameter		L1	L2	L3
dt (s)	Timestep	300	300	300
$h_0$ (m)	Water depth below MSL	10	10	10
b (m)	Entrance width	2500	2500	2500
$A (km^2)$	Area of basin	18	30	50
$u_0 (m/s)$	Peak velocity outside basin	0.18	0.18	0.18
$n_0 ({ m m})$	Tidal amplitude	1.31	1.33	1.40
$c_0 \; (\mathrm{kg/m^3})$	Initial concentration in harbour basin	0	0	0
$c_{01} \; ({\rm kg/m^3})$	Constant concentration ouside	0.1	0.1	0.1
$c_{02} \; ({\rm kg/m^3})$	Variational concentration outside	0.02	0.02	0.02
$\Delta \rho_M ~(\mathrm{kg/m^3})$	Maximum density difference	3	3	3
$ ho~({ m kg/m^3})$	Mean fluid density outside basin	1025	1025	1025
$w_s (m/s)$	Effective settling velocity	0.0003	0.0003	0.0003
$\Delta_{dry} \; (\mathrm{kg/m^3})$	Dry bulk density of deposited mat.	400	400	400
T(s)	Tidal period	43200	43200	43200
$F_{1,flood}$ (-)	Horizontal exchange coefficient	0.025	0.025	0.025
$F_{1,ebb}$ (-)	Horizontal exchange coefficient	0	0	0

## Appendix E Calculations flood protection system

#### E.1 Dyke calculations

In this appendix, the calculation for the height of the river dykes for the year 2100 is made. Equation (6) and Equation (7) are used to determine the crest height of the dyke without settlements, compaction and safety margin. The maximum of the two is used for the final design of the crest height (Van der Meer et al., 2018).

$$\frac{q}{\sqrt{g \cdot H_{m0}^3}} = \frac{0.026}{\sqrt{\tan \alpha}} \cdot \gamma_b \cdot \xi_{m-1,0} \cdot e^{-\left(\frac{2.5 \cdot Rc}{\xi_{m-1,0} \cdot H_{m0} \cdot \gamma_b \cdot \gamma_f \cdot \gamma_\beta \cdot \gamma_v}\right)^{1.3}} \tag{6}$$

$$\frac{q}{\sqrt{g \cdot H_{m0}^3}} = 0.1035 \cdot \exp\left[-\left(1.35 \frac{R_c}{H_{m0} \cdot \gamma_f \cdot \gamma_\beta \cdot \gamma^*}\right)^{1.3}\right]$$
(7)

In which:

- $\gamma_b$  = Influence factor for a berm [-]
- $\gamma_f$  = Influence factor for roughness elements on a slope[-]

 $\gamma_{\beta}$  = Influence factor for oblique wave attack [-]

- $\gamma_v$  = Influence factor for a wall at the end of a slope [-]
- $\gamma * =$  Influence factor for non-breaking waves [-]
- $H_{m0}$  = Significant wave height estimate [m]
- $\alpha$  = Slope dyke angle [°]
- $\xi_{m-1,0}$  = Breaker parameter [-]
- $R_c$  = Crest free board [m]
- $q = \text{Overtopping discharge } [\text{m}^3/\text{s/m}]$
- $g = \text{Gravity acceleration } [\text{m/s}^2]$

At the existing dykes there is no berm present. For ease of constructability therefore no dyke berm will be constructed in the future. Existing dykes then do not have to be adjusted too much to match the design. The influence factor for a berm,  $\gamma_b$ , is therefore equal to 1.0. The influence factor for roughness elements on the slope,  $\gamma_f$ , is also equal to 1.0 because there is grass present (Van der Meer et al., 2018). Influence factor  $\gamma_\beta$  is equal to 0.8 for a wave direction of 60°. Furthermore, there will be a wall on top of the dyke to reduce the dyke height. This affects the influence factor  $\gamma_v$ , see also Equation (8). The height of the wall will be 1.0 m. In the case of a smooth dyke with a storm wall  $\gamma_v = \gamma *$  (Van der Meer et al., 2018). This is the case for this dyke design.

$$\gamma_v = e^{\left(-0.56 \cdot \frac{h_{\text{wall}}}{h_c}\right)} \tag{8}$$

In which:

 $h_{wall}$  = Wall height [m]  $R_c$  = Crest height of the dyke [m]

The difference between significant wave height  $H_s$  and the estimate of the significant wave height  $H_{m0}$  is often very small (Van der Meer et al., 2018). Therefore these are assumed to be the same, namely 2.2 m as determined with Simulating WAves Nearshore (SWAN) within Delft3D. The value for the outer slope of the dykes  $\alpha$  is equal to  $\frac{1}{3}$ , this is taken from Section 3.3.2.

For the maximum accepted overtopping discharge guidelines are available. It depends on the design characteristics of the dyke. For a grass-covered crest and landward slope, maintained and closed grass cover and a value for  $H_{m0} = 1 - 3$  m the overtopping discharge is equal to 5 l/s/m or  $0.005 \text{ m}^3/\text{s/m}$ . For the determination of the breaker parameter Equation (9) is used (Van der Meer et al., 2018).

$$\xi_{m-1,0} = \frac{\tan \alpha}{\sqrt{\frac{H_{m0}}{L_0}}} \tag{9}$$

In which:

 $L_0$  = Wavelength [m]  $H_{m0}$  = Significant wave height estimate [m]  $\alpha$  = Slope dyke angle [°]

The calculation results in a crest height of 4.9 m without settlements, compaction and a safety margin. A compaction rate of around 5 % is assumed and added to the crest height, as well as RSLR. This results in a final crest-free board of 6.8 m, excluding the vertical wall with a height of 1 m.

#### E.2 Hydrostatic load calculations

To calculate the hydrostatic loads, the pressure caused by a water column has to be calculated first with Equation (10):

$$p = \rho \cdot g \cdot h \tag{10}$$

In which:

 $\rho$  = Density of the water [kg/m<sup>3</sup>]

 $g = \text{Gravity acceleration } [\text{m/s}^2]$ 

h = Height of the sea [m]

The density of the water differs per side. The density of the fresh river water is  $1000 \text{ kg/m}^3$  and the density of seawater is  $1020 \text{ kg/m}^3$ . To calculate the hydrostatic force, Equation (11) is used:

$$p = \frac{1}{2} \cdot \rho \cdot g \cdot h^2 \tag{11}$$

The critical water heights in 2100 are given in Table E.1. The maximum sea water level during the high discharge season is a combination of high tide, RSLR, and storm surge. The minimum sea water level during that period is caused by the low tide. The maximum water level is limited by the height of the river dykes. This is equal to 3.0 m along the Ham Luong River close to An Dien and My An (Institute of Coastal and Offshore Engineering, 2020). As said before in Section 3.3.2 it is assumed that this is with respect to MSL. Furthermore, it is assumed that the height of the river dykes is 3.0 m along the entire Ham Luong River. The minimum water height of the rivers is related to the minimum discharge during that period.

During the low discharge season there are no large storms so the maximum sea water level is not influenced by storm surge but by high tide and RSLR. The minimum sea level is caused by the spring tide or Lowest Astronomical Tide (LAT). The maximum river height follows from the maximum river discharge during that season and the minimum river height follows from the minimum river discharge. To determine these heights, water level data has been used from the measurement station My Thuan received from Thuyloi University. The resulting extreme water levels are visible in Table E.1.

Table E.1: Maximum and minimum water levels in 2100

Season	Sea (max)	River (max)	Sea (min)	River (min)	Unit
High discharge season	5.12	3.00	-3.01	-1.42	m+MSL
Low discharge season	2.94	1.48	-3.01	-1.60	m+MSL

In Figure E.1a, Figure E.1b and Figure E.1c the different extreme situations for static loads are visualised. The stress distribution over the height of the wall can be seen, as well as the resulting forces. Note that in the figures, the hydrostatic distribution is given from the riverbed to the water level. The river bed is approximately -10 m + MSL in the Ham Luong estuary. This of course varies for each location but for preliminary design this is a good starting point.



Figure E.1: Hydrostatic loads for different load combinations



(c) Hydrostatic loads for LC2  $\,$ 

#### E.3 Hydrodynamic load calculations

In this appendix two different calculations for hydrodynamic loads on the structure are explained. The first calculation is for wave impact of incoming waves on a vertical structure. The second calculation is for drag and inertia forces of the water against the pier.

Linear wave theory is assumed to determine the wave length, which is needed in this calculation. The water depth is 10 m and the period is 8.7 s (SW wind and return period 100 years) on the sea and is not affected by the water depth when entering the river. By using Equation (12), the wave length could now be determined by iteration.

$$\lambda = \frac{g}{2\pi} T^2 \tanh\left(\frac{2\pi d}{\lambda}\right) \tag{12}$$

In which:

d = Water depth [m]  $\lambda$  = Wave length [m] T = Period [s]

This results in a wavelength of 78 m. By dividing the water depth by the wavelength, it is clear that the water is intermediate deep by using the intermediate water criteria visible in Equation (13):  $\frac{10}{78} = 0.13$ .

$$\frac{1}{20} \le \frac{d}{\lambda} \le \frac{1}{2} \tag{13}$$

By taking the water depth as unknown, the boundaries of the intermediate water zone could be determined. The boundaries are between 3.9 m and 39 m. Therefore, the Ham Luong River is characterised as intermediate water.

#### Wave impact on closed gate

For linear wave theory the pressure distribution by incoming waves is described with the following formula (Voorendt, 2022):

$$p = \rho \cdot g \cdot H_i \cdot \frac{\cosh(k \cdot (d+z))}{\cosh(k \cdot d)} \quad \text{for } -d < z < 0$$
(14)

$$p = \left(1 - \frac{z}{H_i}\right) \cdot \rho \cdot g \cdot H_i \quad \text{for } 0 < z < H_i \tag{15}$$

In which:

 $H_i$  = Half of the wave height of an incoming wave [m]

- k = Wave number of the incoming wave  $\left(k = \frac{2\pi}{L}\right)$  [m<sup>-1</sup>]
- $\rho$  = Density of salt water (typically  $\rho = 1020 \text{kg/m}^3$ ) [kg/m<sup>3</sup>]
- $g = \text{Gravity acceleration } [\text{m/s}^2]$
- d = Total depth [m]
- z = Depth level [m]

The wave height in the estuary is determined with using the wave module from Delft3D. This wave module uses SWAN. SWAN is a third-generation wave model that can model waves in coastal regions and inland waters (Holthuijsen, 2007). The wave height at sea and period are used as input for a Joint North Sea Wave Project (JONSWAP) spectrum used in the model. From Figure B.5a it can be seen that the direction of the

waves coming from the east will enter the Ham Luong estuary. The maximum wave height is 7.2 m and the period is 9.7 s at sea estimated for 2100, see Table B.4. This results in a wave height of 2.2 m at location 1. This is at the location of the barrier closest to the river mouth, which is the critical location regarding wave action.

The resulting pressure distribution for a wave height of 2.2 m can be seen in Figure E.4. The pressure distribution from the river bed to the still water level is non-linear. The extra water pressure from the wave results in a linear pressure distribution from the still water level to the wave height. A mean depth of 10 m is used for the calculations. The total force (per metre stretch) that is acting on the structure is shown with the blue arrow pointing to the structure. The force acts over the complete length of the structure as a distributed load. It is found by integrating the entire pressure distribution over the height.



(a) Wave force, high discharge season LC1



(b) Wave force, low discharge season LC5

Figure E.2: Wave forces on a closed structure for different load combinations

#### Water flow past the piers

The drag and inertia forces of the water against the piers will be calculated with the Morison equation. To calculate the maximum force from the Morison equation, Equation (16) will be used. The drag and inertia components are out of phase.

$$f_{\max}(z) = \sqrt{f_{D_{\max}}(z)^2 + f_{I,\max}(z)^2}$$
(16)

For a cylindrical structure, the drag and inertia components of the Morison force are written as follows in Equation (17) and Equation (18):

$$f_{D_{\max}} = \frac{1}{2} \cdot \rho \cdot C_d \cdot D \cdot u_{\max} \cdot |u_{\max}|$$
(17)

$$f_{I_{\max}} = \frac{\pi}{4} \cdot \rho \cdot C_m \cdot D^2 \cdot \dot{u}_{\max} \tag{18}$$

In which:

D = Diameter [m]  $u_{max} = \text{Maximum flow velocity [m/s]}$   $\dot{u}_{max} = \text{Maximum flow acceleration [m/s<sup>2</sup>]}$   $C_d = \text{Drag coefficient}$  $C_m = \text{Inertia coefficient}$ 

For the calculation of the maximum flow velocity and acceleration Equation (19), Equation (20) are used.

$$u_{max}(z) = \gamma_{ws} \cdot u_{wave}^{\max}(z) + \gamma_{cb} \cdot u_{current}(z)$$
<sup>(19)</sup>

$$\dot{u}_{max}(z) = \gamma_{ws} \cdot \dot{u}_{wave}^{\max} \tag{20}$$

In which:

 $\gamma_{ws}$  = Wave spreading factor [-]  $\gamma_{cb}$  = Current blockage [-]  $u_{current}(z)$  = speed of the current at z [m]  $u_{wave}^{max}$  = Maximum wave velocity [m/s]  $\dot{u}_{wave}^{max}$  = Maximum wave acceleration [m/s<sup>2</sup>]

The wave spreading factor is used to take into account that the environmental conditions are multi-directional and random. A wave spreading factor of 0.906 is chosen which is a standard value (Hoving & van der Male, 2022). During extreme weather events, this value could potentially change to a lower value. Furthermore, a current blockage factor is used because Morison equation only considers a single body which does not influence the current. But the presence of the structure and having multiple bodies with a significant length in the water interferes the current. Therefore it is estimated that the current blockage factor is 0.8. The speed of the current is the flow speed in the river described in Table E.2. These values are obtained from Delft3D. For the maximum flow velocity and acceleration regarding the waves, the following calculations are made:

$$u_{\text{wave}}^{\max} = \max\left(\hat{\zeta} \cdot \omega \cdot e^{k \cdot z} \cdot \cos(k \cdot x - \omega \cdot t)\right) = \hat{\zeta} \cdot \omega \cdot \delta_u \tag{21}$$

$$\dot{u}_{\text{wave}}^{\max} = \hat{\zeta} \cdot \omega \cdot \delta_u \tag{22}$$

In which:

 $\hat{\zeta}$  = Half of the wave height [m]

- $\omega = 2 \cdot \frac{\pi}{T} = \text{Frequency [rad/s]}$
- $k~=2~\cdot \frac{\pi}{L}$  = Wave number [-]
- d = Total depth [m]
- z = Depth level [m]
- t = Time [s]

T = Time period [s]

$$L = Wave length [m]$$

 $\delta_u$  = Depth decay function [-]

The depth decay function is described by:

$$\delta_u = \frac{\cosh(k \cdot (z+d))}{\sinh(k \cdot d)} \tag{23}$$

The drag and inertia coefficients that are needed in the Morison equation are calculated with the Keulegan–Carpenter (KC). This is a formula that describes the relative importance of the drag forces over inertia forces for a slender object (Journée & Massie, 2001). With Equation (24) the KC number can be calculated.

$$KC = \frac{u_a \cdot T}{D} \tag{24}$$

In which:

- $u_a$  = Amplitude of the flow velocity [m/s]
- T = Oscillating flow period [s]
- D = Cylinder diameter [m]

In this formula, the piers are assumed to be cylindrical. A diameter of 4.0 m allows the piers to still be slender structures, which is an assumption for the Morison equation. Furthermore, it is approximately the same width as the piers of the Cai Lon - Cai Be sluice. It is assumed that the amplitude of the flow velocity is similar to the flow velocity and the oscillating flow period is the same as the wave period. The flow speeds of the water are taken from the Delft3D model for the different LC's. They can be seen in Table E.2. For each LC the three barrier locations are checked upon the maximum value for the flow speed. This value is taken as the critical value for that situation.

The KC number in combination with the surface roughness can be used to determine the inertia and drag coefficients. The formula for the surface roughness can be seen in Equation (25). For the roughness height the value of  $\epsilon = 0.005$  m is used for course concrete. This is because the concrete will be placed in an environment where organic matter can easily grow on it, increasing the surface roughness.

Surface roughness = 
$$\frac{\epsilon}{D}$$
 (25)

In which:

 $\epsilon = \text{Roughness height [m]}$ 

D = Cylinder diameter [m]

From Figure E.3a and Figure E.3b the right values for the drag and inertia coefficients can be determined for the different KC numbers for each situation. Note that the graphs are designed for vertical cylinders which is not the case. Because of the very low KC values for each situation all the values for drag an inertia are the same. The drag coefficient has a value of  $C_d = 1.2$  and the inertia coefficient has a value of  $C_m = 2.0$ . The drag and inertia forces are then calculated with Equation (16).

In Table E.2 the resulting drag and inertia coefficients for each situation are visualised. The flow speeds from Delft3D and the corresponding KC numbers for every LC are also depicted. It is observed that the difference in flow speeds does not result in big differences in the drag an inertia coefficients. The values of  $C_d = 1.205$  and  $C_m = 2.00$  are therefore used for the calculation of the Morison forces.

 Table E.2:
 Drag and inertia coefficients

	LC3	LC4	LC7
u [m/s]	1.00	2.00	1.50
KC [-]	2.18	4.35	3.26
$C_d$ [-]	1.205	1.205	1.205
$C_{m}$ [-]	2	1.975	2



(a) Suggested inertia coefficient values (Journée & Massie, 2001) (b) Suggested drag coefficient values (Journée & Massie, 2001)

Figure E.3: Inertia and drag coefficients, based on Keulegan Carpenter Number

Now, all previous formulas will be filled in. The incoming drag and inertia forces for flow coming from the sea in both the high and low discharge seasons are calculated and visualised in Figure E.4a and Figure E.4b. These situations correspond to LC3 and LC7. Storm surge is not regarded for the water height of the sea during the low discharge season as there are no storms then. These situations normally only occur when the gates fail to close as discussed earlier.

During the high discharge season, the water level in the river is very high. In the case the gates are opened this leads to large drag and inertia forces on the piers. A steady flow is assumed without large waves. Furthermore,

the maximum water height of the river is the height of the dykes above MSL, which is 3.0 m as mentioned in Section 3.3.2. This situation corresponds to LC4 and is visualised in Figure E.4c. This is the visualisation of the total distributed Morison force acting on a single pier. This is in kN because the diameter of the structure is already included in the calculation.



(a) Morison force sea, high discharge season, LC3



(b) Morison force sea, low discharge season, LC7



(c) Morison force river, high discharge season, LC4

Figure E.4: Morison forces on the piers for different load combinations

## E.4 Boreholes

In this appendix different boreholes in the sea between the Go Cong Dong district near the river mouth of the Mekong and Vung Tau are displayed. The locations can be seen in Figure E.6a. The actual boreholes itself can be seen in Figure E.5, the one used for the determination of soil stresses is the LK01 borehole on the left. In Figure E.6b the geological index of the different boreholes is visualised.



Figure E.5: Soil layer distribution near Vung Tau (Truong Hong Son, 2012)



(a) Boreholes locations near Vung Tau (Truong Hong Son, 2012) 20

(b) Geological index of the soil layers (Truong Hong Son, 2012)

Figure E.6: Location and soil layers of boreholes

## E.5 Relationship between opening size and peak tidal flow during mean tide

The graph in Figure E.7 shows a relation between opening sizes of storm surge barriers and peak tidal flow during mean tide. The figure shows different ranges for peak tidal velocities. For the determination of the total area of the gates for the storm surge barrier in the Ham Luong River a peak tidal velocity of 1.5 m/s was chosen. This value is the mean value of the maximum flow velocities given in Table E.2. Therefore this is a good approximation for all alternatives.



Figure E.7: Relationship between opening size and peak tidal flow during mean tide (Mooyaart & Jonkman, 2017)

#### E.6 Combined hydrodynamic and -static load on the gate

In Figure E.1a the hydrostatic and -dynamic loads for LC1 are visualised. This is a combination of relatively large hydrodynamic load from wave impact and a high hydrostatic load on the gates. The components of the the total load are visualised with red and green colour. The total load on the gate is visualised with blue colour. The maximum load is depicted with the red dot marker. For LC2 and LC5 the same calculation is made. The hydrodynamic and -static loads are different resulting in different total loads on the gates. LC2 has no hydrodynamic loads acting on the gate.



Figure E.8: Hydrostatic and -dynamic Load LC1



Figure E.9: Hydrostatic and -dynamic Load LC2



Figure E.10: Hydrostatic and -dynamic Load LC5

#### E.7 Load combination factors

For the load combination factors, the Eurocode is used. This is done because there is not a very good guideline for structural calculations in Vietnam and often Eurocode is used as well. The design value of the load effect  $E_d$ , the fundamental combination, is as in Equation (26). This is according to Eurocode 0 for structural design (CEN, 2002).

$$E_{d} = E\left\{\sum_{j\geq 1}^{n} \gamma_{G,j} \cdot G_{k,j} + \gamma_{P} \cdot P + \gamma_{Q,1} \cdot Q_{k,1} + \sum_{i>1}^{n} \gamma_{Q,i} \cdot \psi_{0,i} \cdot Q_{k,i}\right\}$$
(26)

In which:

 $G_{k,j}$  = Characteristic value of permanent load j

 $\gamma_{G,j}$  = Partial factor for permanent load j

P = Representative value for the pre-stressing load

 $\gamma_P$  = Partial factor for the pre-stressing load

 $Q_{k,1}$  = Characteristic value of the main variable load

 $\gamma_{Q,1}$  = Partial factor for the main variable load

 $Q_{k,i}$  = Characteristic value of variable load i

 $\gamma_{Q,i}$  = Partial factor for variable load i

 $\psi_{0,i}$  = Combination reduction factor for variable load i

For the self-weight of the structure and permanent loads, the load combination factor  $\gamma_{G,j} = 1.35$  is used. For the live loads; water pressures, wind loads, and other variable loads, load combination factor  $\gamma_{Q,1} = 1.5$  is used.

## E.8 Load schematisation on the gate

In Figure E.11 and Figure E.12 the loads on the gate structure are visualised. These are the loads without load combination factors. For calculating the results the load combination factors for these loads have been applied.



Figure E.11: Hydrostatic load on the gate  $\$ 



Figure E.12: Hydrodynamic load on the gate



Figure E.13: Steel frame stresses



Figure E.14: Steel frame deflections

## E.9 Euler buckling frame beams

The beams that connect the two arcs that form a single lens frame are compressed when loaded. This may lead to buckling. The beams are regarded as hinged on both sides for this particular structure, resulting in a buckling length of  $L_b = L$ . Euler buckling; Equation (27) is used for the UC on buckling.

$$P_{\rm cr} = \frac{\pi^2 \cdot E \cdot I}{L_b^2} \tag{27}$$

In which:

E = Elastic modulus [N/mm<sup>2</sup>]I = Moment of inertia [mm<sup>4</sup>]

 $L_b = \text{Buckling length [mm]}$ 

## E.10 Load schematisation and moments girders

In Figure E.15a and Figure E.15b the hydrostatic and hydrodynamic loads on the girder are schematised. In Figure E.15c the resulting moments can be seen.



Figure E.15: Loads and moment vertical girder




Figure E.16: Front view of the storm surge barrier



Figure E.17: Cross section with loads of the storm surge barrier for LC1  $\,$ 



Figure E.18: Cross section with loads of the storm surge barrier for LC2

G1 until G4 are calculated by multiplying the volume of the element by the density of reinforced concrete (2500 kg/m<sup>3</sup>) and subtracting the buoyancy force. The buoyancy force is dependent on the current water level on both sides of the gate. This is a too detailed calculation for a primarily design. It is conservative to assume that the water level is equal to the maximum water level of the sea, so this assumption is made for the pier. Therefore, the buoyancy force will also work halfway the length of the pier. For the gate, the buoyancy is also dependent water levels and whether the frames are submerged or not. Furthermore, it is also dependent on what side of the gate is submerged, because only 1 side is shielded by a skin plate. This skin plate prevents the frames from becoming submerged. For this simplicity of this calculation it is therefore assumed in which the maximum buoyancy takes place. This is when the river water level is at its maximum, which is the case for LC4 with a river water level of 13 m. Around 90% of the gates volume is then submerged. For the calculation of the buoyancy fore, the density of salty seawater (1020 kg/m<sup>3</sup>) is multiplied by the "submerged" volume.

Table E.3:	Internal	forces	acting	on	the storm	surge	barrier
------------	----------	--------	--------	----	-----------	-------	---------

Element	Self weight [MN]	dA [m]	M A [MNm]	dB [m]	M B [MNm]
G1 (Lifting structure)	5.1	10.0	-51	4.0	20
G2 (Pier)	13.9	7.0	-97	7.0	97
G3 (Gate)	8.1	10.0	-81	4.0	32
G4 (Sill)	6.1	10.0	-61	4.0	24
Total	33.1		-290		174

#### E.12 The elevation of the water and barrier during operation

The height of the gate has been determined in a way, such that the maximum significant wave height at location 1 will be stopped while the maximum sea water level of 15.1 m is present (=LC1). Therefore, the gate height is equal to 14.7 m, as visible in Figure E.19. For the lifting structure, the height has been designed to be able to let the maximum water level of 13 m flow underneath the gate (=LC2). This resulted in that the top of the lifting structure is 27.7 m. In case of a gate opening failure (=LC3), the gate is open when there is a storm coming from the sea. In that case, not all the water will be able to flow underneath the gate, because its designed to allow the maximum river water level to flow underneath the gate. The sea water level from the sea is higher, so 2.1 m of the water will cause a hydrostatic load and 1.1 m will be exposed to wave impact against the gate. For this primarily design, this calculation will not be executed. For this calculation, it will be assumed that the gate will be raised 3.2 metre further out of the water such that the sea water could flow underneath. The wave impact is in the same order of magnitude as the wind load which supports this assumption. For LC4 and LC7, the gates are also opened but water levels are low enough to let the water flow underneath.



Figure E.19: The heights of the water and barrier during operation and rest. For this figure, the sea water level for LC1 is drawn on the left and the river water level for LC2 is drawn on the right, both in red. This is not an existing LC, but is used to explain complications during operation.

#### E.13 Moments and forces resulting from external loads

With the help of Figure 7.7, the moments and horizontal force can be calculated. The static hydrostatic load is multiplied by the permanent safety factor and the wind load, Morison load or wave load are multiplied by the variable safety factor. These factors are 1.35 and 1.5 relatively. Furthermore, the static loads, wind loads and wave loads are multiplied by the width of the gate and pier. So now, the moments and loads are not per metre depth anymore. The results are visible in Table E.4

Table E.4: Bending moments and horizontal forces per LCs, resulting from the load schematisation

	LC1	LC2	LC3	LC4	LC5	LC7
$\sum M$ [MNm]	271	-135	18	-16	161	16
$\sum F$ [MN]	55	-35	1.0	-0.9	37	1.0

For LC1, the maximum bending moment resulting from the external hydrostatic and hydrodynamic loads around point A, is equal to 271 MNm (clockwise). The bending moment resulting from the self-weight of the structure

is -290 MNm (anticlockwise) so the pier will not start to rotate around point A. The UC is 0.94 for LC1. Note that the gate and sill will also resist the rotation of the pier with its self-weight.

For LC2, the maximum bending moment resulting from the external hydrostatic and hydrodynamic loads around point B, is equal to -135 MN (anticlockwise). The self-weight of the structure creates a bending moment of 174 (clockwise) which is enough to resist the rotation around point B. The UC is 0.77 for LC2.

#### E.14 Top view storm surge barrier



Figure E.20: Top view of the storm surge barrier with a potential sluice and abutment for future plans with unknown dimensions

#### E.15 Bed protection calculations

For the design of the bed protection to prevent clear-water scour behind the sill, the following formula was used (Schiereck & Verhagen, 2012):

$$\frac{h_{se}}{h_0} = \frac{0.5\alpha\bar{u} - \bar{u}_c}{\bar{u}_c} \tag{28}$$

In which:

 $h_{se} = \text{scour depth [m]}$   $h_0 = \text{initial water depth [m]}$   $\alpha = 2$  [-]  $\bar{u} = \text{flow velocity [m/s]}$  $\bar{u_c} = \text{critical flow velocity [m/s]}$ 

For the calculation of the needed length of the scour protection, a situation with very low water levels (LAT = 7 m) and the highest occurring flow velocities (2 m/s) is assumed. The critical flow velocity is based on the assumed grain size of 200  $\mu$  m (Appendix B.6) and is defined as 0.35 m/s. A value of 2 for  $\alpha$  is taken, as the relation between the sill height (1.5 m) and water level gives a value of 0.125 (see Figure E.21).



Figure E.21:  $\alpha$  value for scour protection (Schiereck & Verhagen, 2012)

This calculation results in an expected scour depth of 56 m. Assuming a slope of slide of 1:15 gives a needed protection length of 850 m.

In order to make a first estimation on the needed grain size on the scour protection, the Shields formula is used (Schiereck & Verhagen, 2012):

$$D_{n50} = \frac{K_v^2 u^2}{\psi_c \Delta C^2} \tag{29}$$

In which:

 $D_{n50} = \text{grain size of top layer [m]}$   $K_v = \text{separation factor [-]}$  u = flow velocity [m/s]  $\psi_c = \text{Shields parameter [-]}$  $\Delta = \text{relative density [kg/m^3]}$ 

 $C = Chézy \text{ coefficient } [m^{1/2}/s]$ 

The most critical location regarding the needed grain size of the protection layer, is just behind the reattachment zone downstream of the barrier. This is the location where the flow velocity is at its highest. The factor  $K_v$  is accounting for the induced turbulence due to the separation of flow behind the sill. By using the initial water depth of 12 m and the sill height of 1.5 m, a value of 1.27 for the  $K_v$ -coefficient can be determined in Figure E.22.



Figure E.22:  $K_v$  coefficient behind sill (Schiereck & Verhagen, 2012)

Besides the turbulence related to the reattachment zone, in general a turbulence related to turbulent flow should be included when defining the flow velocity:

$$\iota = u(1+3r) \tag{30}$$

This results in an effective flow velocity of 3.52 m/s. The value of  $\psi_c$  (Shields parameter) is chosen as 0.03, as this value is considered as the safest value that corresponds to no transport and highest stability (Schiereck & Verhagen, 2012). The relative density is related to the density of the used material and the density of the water. Using material with a density of 2650 kg/m<sup>3</sup> and a density of 1000 kg/m<sup>3</sup> for fresh water leads to a value of  $\Delta$ =1.65. The value of the Chézy coefficient is iteratively calculated with:

$$C = 18 * \log\left(12\frac{h}{k_r}\right) \tag{31}$$

In which:

h = initial water depth [m] $k_r = \text{roughness } (\approx 2D_{n50})$ 

In an iterative way, these equations are used to determine the grain size of the upper protection layer. The result is a  $D_{50}$  of 0.34 m ( $D_{n50}$ = 0.28 m). This corresponds to the grading LM<sub>A</sub> 40-200.

For calculating the needed bed protection to prevent scour induces by wave action on the closed barriers, the following equation is use (Voorendt, 2022):

$$h_{\max} = 0.4 \cdot H\left(\sinh\left(\frac{2 \cdot \pi \cdot h_0}{\lambda}\right)\right)^{-1.35}$$
(32)

In which:

H = Wave height [m]

 $h_0 = \text{Water depth} [m]$ 

 $\lambda = \text{Wave length } [m]$ 

The most critical combination of water depth and wave height is formed by LAT (7 m) in combination with the maximum expected wave height (2.2 m). This situation leads to scour depth of 0.002 m. Is is expected that the designed scour protection to prevent clear-water scour will be sufficient to prevent scour due to wave action at a closed gate.

## Appendix F Meetings

During the time in Vietnam the research mostly took place in Hanoi, the capital of the country. During a two-week trip to the south of Vietnam to study the area of interest, HCMC was also visited. In this city, a lot of government institutions and companies are situated that relate their research to the Mekong Delta. During the stay, research was done at the SIWRP which provided a lot of helpful information. There were meetings with experts from this institute about the Ham Luong River. Furthermore, Royal HaskoningDHV, a Dutch company that is an expert in consulting engineering and has done a lot of research on the Mekong Delta, was visited. The company has contributed to the Mekong Delta Regional Masterplan and is involved in various projects in the area. Also, the ICOE was visited, which is another institution from the Vietnamese government. They also provided information and insights about the subject. During all of the visits, a presentation was given by the MDP group about the progress of the project and the preliminary findings. The institutions and company also prepared their own presentations about their work related to the Mekong Delta. Afterwards, a fruitful discussion was held about the topics. In this appendix, the different meetings are reported in more details, but the most important take-aways are:

- The Delft3D model is quite limited in scope and processes, which should be taken into account in evaluating the results.
- Closing one estuary has a big effect on other estuaries and will only protect a limited area. The dykes need to be raised in the case of a barrier placement, as well as the case of no barrier.
- The increase of salinity is not necessarily a bad thing, and could also be considered as an opportunity. The affected people can adapt to it
- In agri- and aquaculture the tidal fluctuation is used for watering and draining, while saving energy.
- The Vietnamese people already adapt to the changing circumstances and they can continue to do so, since climate change is relatively slow.
- Adaptation can be combined with Building with Nature.
- The Cai Lon barrier, in the west of the Mekong Delta, was deemed necessary but the operation schedule still allows for salt intrusion and there is a lot of controversy over that barrier.

#### F.1 Meeting SIWRP

Date: 25-09-2023

Location: SIWRP office HCMC

Present:

D. Duc Do	Head of the institute
L. Thanh Dang	Deputy Director of the Institute
D. Thanh Nguyen	Vice Director of the Center for Disaster and Climate Change
B. Hong Tran	Employee
L. Nguyen	Employee
S. Bao Ngoc Hoang	Employee
H. Thu Thi Nguyen	Employee
N. Trung Nguyen	Employee

The SIWRP is an institute that is connected to the MARD. It focuses on research about irrigational planning, and water flows in the Mekong Delta and the southern regions. A short summary about the meeting is given below. The discussion has been subdivided into several main topics but they are all interconnected and were discussed interchangeably during the meeting.



Figure F.1: Presentation at SIWRP

#### Modelling

At the SIWRP the modellers work with Mike11 in 1D which can be used to obtain flow speeds. This program uses Qgis and Gis, and they use Sobek as well. A 1D-model of the entire Mekong Delta exists within the SIWRP. The used discharge values are based on the discharge upstream in Kratie, Cambodia. About 10% of this discharge goes into the Ham Luong estuary. During the dry season, 3000  $\text{m}^3$ /s comes in from the Upper Mekong Basin. Then 70% is divided over the 9 river mouths.

Our Delft3D model that is used for the Ham Luong River, is a 2D model for only the Ham Luong estuary. The 2D is considered to be sufficient, according to SWIRP. There is a huge discharge from the river into the sea, so the water distribution will change. The effect of the closure of the Ham Luong estuary on other rivers is important, but this effect is not included in our model. Our model is more useful to choose the location of the sluice gate than the impact on connected rivers and the upstream. The shortcoming of our model must be reflected on in our report. The brackish area in the Ham Luong estuary makes it easy to see the effect on salt water. However, water level changes take longer to see in the model. Biodiversity changes take very long and are impossible to see now at this moment. Literature raises this issue. A comparison with the big barrier in the Netherlands could be useful. It rarely closes and is only operated for big storms. It is good to see the big picture.

Furthermore, the RCP8.5 is correct for long term. For mid term, the RCP4.5 is used. For the return period, 100 years must be used for a structure. So 0.5% or 1%. There are some standards for that, but in general, 100 years is considered.

#### Barrier

When building a barrier the most important questions are: Is the Ham Luong estuary important for the Mekong Delta or not? Should we build the gate and where? To answer these questions the model and project need to cover more of the Mekong Delta and Delft3D might not be suitable for that.

The SIWRP has used a number of criteria for the choice of location:

- 1. Transportation with a connection between the gate and the road
- 2. The road along the sea dyke; connect with the sluice gate or not
- 3. Smallest width of the estuary for the gate placement
- 4. Effects on fresh water and salt intrusion
- 5. Hydrodynamic loads (tidal range, storm surge, waves)

The goal is to get control of fresh water in the Mekong Delta. The effect of the barrier on the shortage of fresh water, with the SLR, salt intrusion and inundation.

Considering storm surge from sea. The SIWRP did not study heavily about the environment. The tidal change in the West Sea and East Sea are different but very high. The environmental impact has to be considered when a big structure like the Ham Luong barrier would be constructed. The geology is not very different over the length of the Ham Luong estuary. For a big structure, it can be regarded the same, because the foundation is 40 m below ground. In the early Mekong Delta Plan, hydraulic impacts from the sea were the main focus. The newer plan looks more upstream and more to the freshwater storage upstream. A barrier on location 3 has less hydrodynamic impact and less storm surge impact. Before, the SIWRP looked into other locations (more downstream), but now we prefer location 3. Small side channels will be (or are) closed with sluices and freshwater intake gates.

#### Multi Criteria analysis

Regarding the MCA it was not yet clear what was meant with agricultural adaptation. It must be clear what is meant with this in the report. Disaster prevention and water supply are clear. It is also good to look at more specific criteria. Change and social effects are interesting. Furthermore, river morphology was not mentioned, this could be a good criteria to investigate as well.



Figure F.2: Group photo with Bao in front of the SIWRP building

### F.2 Meeting Royal HaskoningDHV

Date: 26-09-2023

Location: Royal HaskoningDHV Office HCMC Present: A. Le Ngoc Senior Coastal Modelling Engineer H. Hoang Modelling Specialist

Royal HaskoningDHV has been contributing to the Mekong Delta Master Plan. There were 2 studies made, one in 2013 (not approved by the government) and one in 2021 (funded by the Vietnam government, the first regional planning approved in Vietnam). They have been involved in projects in the Mekong Delta for years, making them experts in the area. Below is a short summary given about the meeting. The discussion has been subdivided into several main topics but they are all interconnected and were discussed interchangeably during the meeting.

#### Adaptation

In the coastal area, the land use could change from agricultural to forestry. This is more sustainable and water quality will increase. Taking into account cost-benefit analysis, the rice is very cheap, shrimp is more valuable. Upstream, along the border of Vietnam the proposed solution is flood-based agriculture. Irrigation in the area is really developed; there are 3.5 rice crops a year. Soil quality becomes reduced due to extensive harvesting. With flood-based agriculture: Fish in the flood season, and rice in the dry season is a solution for this problem. More sediment from the floods will be on the land. The bed will become better and no pesticides have to be used. Regarding the brackish area; do not make it fresh, but keep it brackish. That is the new approach from Royal HaskoningDHV. An intermediate/intermittent zone must be kept, not completely fresh like before. Change rice for shrimp for the agricultural sector. Coconut can also survive in brackish area.

In the Go Cong district shrimp farms are right next to the sea. The people have to accept brackish-fresh zones. People are already adapting to salt water. There have not been any direct interviews done by Royal HaskoningDHV, however this was observed from satellite imagery. People have learned from their own experiences. For example they know that, when they need fresh water, they collect and store it in April and May, so they retain the water, because later in the year there will not be enough fresh water.

If everyone in the coastal area changes to shrimp and coconut, prices will not decrease. Shrimp and Coconut have a price on the global market, so it will not be affected that much. Considering food security in Vietnam there is enough rice for the people. If some areas change to shrimp and coconut this is not an issue. Rice fields have 1.8 million hectares, only 300.000 hectares can be changed into intermittent zone. Pineapple has a domestic and global market as well.

About fresh water: Fresh water for domestic use is small part. They can use fresh water from upstream for instance from My Tho. Furthermore, is agricultural fresh water necessary? During low tide you can also take the fresh water in. Side channels are also used. Use pumps for a freshwater supply more upstream. Groundwater extraction results in land subsidence, salt in the soil and polluted soil. Currently they use a deep layer to get groundwater. Problems could be caused by upstream factors, such as outtake of fresh water in upstream areas.

#### Modelling

For the upstream boundary of the model My thuan is not far enough for the Delft3D model. If the branch were to be closed, water would flow to another branch. For a thesis, include the Mekong River to the border of Vietnam and Cambodia. Furthermore, there is a 1D model of the Mekong from Cambodia to the sea in Mike11.

Storm surge is only one case (in 20 years, there has been 1 storm), there are no storm surges in Southern Vietnam. Storm surge is not significant. Flood is not related to the wet season but to tidal amplifications.

#### Local people

For irrigation the inhabitants of the Mekong Delta use tidal fluctuation, when to drain it wait for the low water, if they need water they wait for high water. The tidal difference of 3 m/day, while the land is 1 m below sea level, is useful to save energy. They use this to make their living. Shrimp can use brackish water. When a barrier is closed during dry season, farmers have to use fresh water for all activities, since there will not be any brackish water available due to the closure. Shrimp farms will also need salt.

People are also willing to change. A lot of locations on maps were shown where adaptation already took place. People changed to shrimp farms. Besides, the experts at Royal HaskoningDHV think the brackish border only moves up about 10-12 km inland, for SLR of 1 metre. A 100 year time period is a lot of time to prepare for adaptation and climate change.

In Tra Vinh area there is a brackish area and a shrimp area. There is a change from fresh to brackish water. People will adapt to their income. If we close it off, fresh water will only be there. Water will be used from inland channels and rivers.

The income from shrimp is high, while investment is also high. Some of the farmers like to work the easy way and keep the fresh water. They don not want high investment and keep their rice crops. If you accept that some parts will become saltier, salt water is not a problem.

#### Barrier

For the barrier it must be taken into account that an island (location) may not be a stable location. An existing barrier is located in the Ba Lai River. Due to the barrier the water quality is very low and there is an sedimentation issue. For Ba Lai it was also the idea to store fresh water but this is a failure.

The Cai Lon barrier in the west of the Mekong Delta, is 500 metres wide. They want to close it for salt water and get fresh water from the Hau River. But people already use salt for their agriculture. The cost of water is very high because it is pumped from the Hau River. Rice is not that beneficial, so you have to think why you want to keep the fresh water area. Shrimp is more beneficial. 15 years ago the Cai lon was planned, then they thought it was a good plan. Now people have questions about it. The river gates still has to be open at high tide so the salt will still intrude in the channel.

Mekong Delta plan was presented. The MARD said that they want to build a barrier. An alternative approach is to make a dyke along the rivers. It is cheaper and easier to build. One gate has impact for 500 hectares only. The Mekong Delta has more mouths: closing one has effect on others. If you build one gate, you should make more gates. More adaptation is the better way.



Figure F.3: Interaction and discussion during the meeting at Royal HaskoningDHV

### F.3 Meeting ICOE

Date: 29-09-2023

Location: ICOE Office HCMC

Present:

H. Manh Phan Head of the section mathematic modelling and GIS Several other employees

The ICOE is an organisation that conducts research on hydraulic engineering, sediment, climate and coastal areas in Vietnam. The institute is connected to MARD. During the meeting several experts from different disciplines were present. There were experts on oceanography, geoscience and remote sensing, forests, hydraulic structures, environment management, hydrology, natural resources and environment, coast protection and infrastructure. The focus of the ICOE in Ben Tre is on hydrodynamics, rainfall, climate change. Below a short summary is given about the meeting. The discussion has been subdivided into several main topics but they are all interconnected and were discussed interchangeably during the meeting.



Figure F.4: Presentation from ICOE concerning the Mekong Delta

#### Climate

It was confirmed that during the dry season storms do not occur. Every two years there is one storm. Wind velocities are about 16 m/s. A lot of dams are built which has caused the flow from upstream to decrease.

#### Adaptation

Upstream flow decreases because water is transferred to other areas (for example for irrigation purposes). Sluice gates are considered by the government because of the shortage of freshwater supply; to keep fresh water in the Mekong Delta. Vietnamese are good at adapting. Land use changes because of SLR, fortunately, the profits are also higher for saline agriculture.

Climate change/SLR change, but slowly. The structure is very sudden, so the change is significant. The effects are clearly visible. Before, the shrimp farm area was very small. Due to SLR more salt intrusion occurs. Therefore the land use changed from rice crop to shrimp farm gradually.

Upstream flow decreases because water is transferred to other areas (for example irrigation purposes). Sluice gates are considered by the government because of the shortage of freshwater supply; to keep fresh water in the Mekong Delta. Vietnamese are good at adapting. Land use changes because of SLR, fortunately, the profits are also higher for saline agriculture.

The ICOE has made a tool that predicts the water levels at multiple locations in the Mekong Delta. It alerts local people when to protect their property. There are three levels of warnings. This information is passed to

the local administration and they will inform the local people. The tool is based on measured water level data from 20 years before.

#### Modelling

The tidal range is increasing. The tide is higher every year with about 10 mm/year. SLR is about 3 mm/year, subsidence is around 7-8 mm/year. Subsidence rates are about three times larger than SLR. Ben Tre area is average in elevation with respect to the rest of the Mekong Delta. SLR in Binh Dai in Ben Tre was roughly 41 cm in 30 years, also including land subsidence. Some areas have more problems than other areas, which is why they receive more money.

Furthermore, rainfall decreases, but extreme rainfall increases. The flow percentage of the Ham Luong River from the upstream discharge at Kratie is the following:

Year 20009%Dry season12%Flood season7%

#### Local people

A barrier affects the Southern people. They heard a lot about building with nature. This type of structure is not "building with nature". The structure is built for salt intrusion and storm surges. It will affect biodiversity, it will change the habitat/lifestyle of the local people. When a structure is built they have to change agriculture. They just use rice crops. But they have to adapt if this type of structure is built.

#### Barrier

Just two years ago, the Cai Lon barrier was built. A lot of people did not want the barrier. When looking at cost-benefit it might be a good idea. The occupations of the people in the west is also agriculture. Similar to people living in Ben Tre Province. The ICOE did not decide on the project, but they were involved in the research. In their research, they mainly focus on the effect of salinity on agriculture. The conclusion is: we have to built. There was more benefit than cost so it was built. Vietnam government has not yet decided yet about the Ham Luong River.

There is a lot of sedimentation in the Bai Lai branch. This was shown during the discussion on a map.

#### Provided data

The ICOE has no data on water quality in Ben Tre, but mainly on the provinces in the west of the delta. The institute has discharge and current data in Ben Tre, water level, and sediment. The measurement stations are from the institute. Salinity data from January to May in An Thuan. The institute provided data about the following:

- Water level and discharge for 20 years in Ham Luong River
- Salinity data from January to May (dry season) in An Thuan
- Suspended sediment data in the Ham Luong River
- Dyke heights and failure of the dykes



Figure F.5: Group photo with the employees of ICOE

## Appendix G Survey Ben Tre

As stated in the Section 2, a survey was conducted to gather local stakeholders' current opinions and knowledge in Ben Tre Province. The survey questions were designed to find out what functions or aspects of the Ham Luong estuary are most important in their daily life.

The survey was short and consisted of three parts. First, personal information about age, location, and occupation was asked. After that, the participants were asked if they have noticed changes related to climate change and if they could give examples of it. Then the participants were asked to put the statements below in the boxes that they associated it the most with. Each box could only contain one number. To the right the issues with which the participant most strongly disagreed were listed and to the left the issues with which the participant the most strongly agreed with. This is an interview method based on Q methodology, also known as Q-sort. This method is used to study the viewpoint of people on a certain topic using psychology. Q-sort is based on the fact that participants have to rank a group of questions or statements, in this case the issues around the Ham Luong River that are listed below. This is done in such a way that it reflects their viewpoint the best. The participants have no choice but to rank the options, as they can not be all equally important (van Hamel, 2018). The last questions were open questions that allowed the participants to comment on the choices that they had made. Below the list of statements is displayed:

- 1. Availability of fresh water
- 2. Availability of both fresh and salt water
- 3. Enough fish to catch
- 4. Ability to use boats on the river
- 5. Avoiding floods
- 6. Preventing erosion
- 7. Stable employment
- 8. Environment, ecology, and landscape
- 9. Management and intervention in the river and estuary area

This structure was chosen to make the survey fast and clear. The questions were answered on paper. The participants were randomly selected, with the aim of including a diverse range of people with different occupations. During the survey crowded places were visited such as a ferry station and a local coffee shop.

The results of the survey turned out not to be very useful. Although we had three Vietnamese people with us to make clear to the participants what the survey was about, it turned out to be difficult to gather good results. In the end only thirteen filled-in surveys were collected. There were various reasons why the survey was not a great success:

- The participants sometimes did not understand the survey correctly. For instance, multiple numbers were placed in the same box and sometimes not everything was filled in.
- Participants seemed to influence or copy each other's answers. Some participants sat together to fill out the survey. One of the participants did not fill in the survey because he said he had the exact same opinion as another participant. They mentioned this on their survey sheet.
- The explanation and filling out of the survey took very long per person. Therefore not a lot of answers could be collected in the short time that we were in the area.
- Participants seemed not to care so much about the survey. They only filled in what they found interesting and left the other parts blank.
- Some of the results were also influenced by a Vietnamese translator who helped us to explain the survey to the participants. When the results were incomplete, they were filled in by the translator instead of the participant.
- Some of the participants did not trust the survey and did not give honest answers, being afraid that he/she might give away information that could affect him/her negatively.

• Some experts mentioned that people in Ben Tre do not often think about the effect of climate change but are more busy with their day to day activities. If they are asked to give their opinion about our subject, they have not thought about their opinion yet.

During collecting results people also gave information about their experiences. A woman working at a café mentioned that occasionally the water pumped for domestic use has a salty taste. One fisherman said that some of his fish die. He believes that the water is more polluted than it used to be, which he thinks is the reason for his fish dying. What also emerges from the thirteen survey results is that most people considered statement 1, the availability of fresh water, important. Additionally, certain individuals, such as fishermen, emphasized the significance of statements 2 and 3. Furthermore, in our conversations with the local people, we noticed that they primarily value statements that have a large impact on their daily lifes. For instance, statements 6, 8 and 9 were frequently seemed unimportant by the people in Ben Tre.

For surveys in the future, it is recommended to visit public places as was done for this survey. People from different kinds of occupations visit these places so it is easier to get fast and diverse responses. A cafe, restaurant or other place where people are not busy with their job is also recommended. For the layout of the structure itself it is recommended to make it simpler. Probably, most participants had not seen a survey like this before which made it difficult to explain and eventually for the participants to fill out. Therefore, yes and no questions are recommended. Furthermore, it is almost impossible to explain a survey when Vietnamese is not your spoken language. Bring translators to explain the survey, and even then it can be very difficult as was experienced. The Vietnamese people that are used as translators must be instructed carefully because otherwise they can influence the results. With only 4 hours for conducting our survey, it is recommended to have at least a full day with Vietnamese translators to conduct the survey effectively. Officially for conducting a survey in Vietnam it is necessary to get permission from the local authorities. With this permission people trust the survey and they are more likely to fill in the survey. The survey used in this project can be seen on the next page.



Khảo sát từ sinh viên Hà Lan về sự phát triển ở khu vực Cửa sông Hàm Luông



Ngày ThángNăm
Tên:
Tuổi:
Giới tính
Quê quán:
Công việc:
Công việc của bạn có liên quan đến:
Trồng lúa: Có/Không
Cây ăn trái: Có/Không
Đánh bắt cá: Có/Không
Nuôi Tôm: Có/Không

[1] Bạn có cảm nhận được sự thay đổi về biến đổi khí hậu trong cuộc sống hàng ngày không ? Có/Không, làm ơn đưa ra 1 hoặc 2 ví dụ cho sự cảm nhận của bạn

.....

# [2] Đối với chất lượng cuộc sống của tôi gần sông Hàm Luông, chúng tôi đã đề xuất 9 vấn đề liên quan như sau:

- 1. Có đủ nước ngọt
- 2. Có cả nước ngọt và nước mặn
- 3. Có đủ cá để đánh bắt
- 4. Có thể sử dụng thuyền di chuyển trong sông
- 5. Không bị ngập lụt
- 6. Không bị xói lở
- 7. Có công việc ổn định
- 8. Môi trường, sinh thái & cảnh quan
- 9. Việc can thiệp, chỉnh trị khu vực sông sông, cửa sông

Hãy đặt 9 vấn đề này vào các ô trong bảng sau, mỗi ô chỉ có thể có một vấn đề, xin cám ơn



[3] Ý kiến, mong muốn góp ý khác của bạn (nếu có)
Vấn đề bạn rất đồng ý vì:
Vấn đề bạn rất không đồng ý vì: